ED 429 331	EA 029 644
TITLE	Minnesota Education Yearbook, 1998: The Status of Pre-K-12 Education in Minnesota.
INSTITUTION	Minnesota Univ., Minneapolis. Coll. of Education and Human Development.
PUB DATE	1998-00-00
NOTE	95p.
AVAILABLE FROM	Web site: http://edpsy.coled.umn.edu/oea/
PUB TYPE	Reports - Descriptive (141)
EDRS PRICE	MF01/PC04 Plus Postage.
DESCRIPTORS	Comparative Analysis; *Educational Assessment; Elementary
	Secondary Education; *Formative Evaluation; Outcomes of
	Education; *Public Education; State Norms; Yearbooks
IDENTIFIERS	*Minnesota

ABSTRACT

This yearbook provides a review of research concerning the differences in providing quality education to all children. A number of data sources at both the national and state levels were reviewed in compiling the indicators of educational inputs featured within this report. Four focal points are as follows: (1) the financial resources brought to bear on Minnesota's public-education system; (2) several characteristics of the state's current K-12 teaching force; (3) the changing demographics of the state's student population; and (4) the ratio of teachers and staff to students in Minnesota schools. The yearbook is divided into five chapters. Chapters 1 and 2 describe how tests have changed and provide an overview of learning. Educational inputs and processes, such as school finance and teacher characteristics, are covered in chapter 3, whereas chapter 4 focuses on participation and coursework, attendance, and graduation. The last chapter examines achievement and discusses comparative assessments. Findings include the fact that Minnesota's per-pupil funding has increased over time, but at a slower rate than for the nation as a whole. The average age of teachers has risen dramatically over the past 10 years, with 38 percent of the teachers falling into the 45-54-year-old age bracket. Minority students are characterized by higher rates of poverty, limited English proficiency, and disability. (RJM)

******	* * * * * * * * * * * * * * * *	* * * * * * * * * * * *	******	* * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * *
*	Reproductions	supplied by	EDRS are	the best that can	n be made *
*	-	from the	original	document.	*
******	* * * * * * * * * * * * * * *	* * * * * * * * * * * * *	* * * * * * * * * *	* * * * * * * * * * * * * * * * * *	****



EA 02964

ERĬC



U.S. DEPARTMENT OF EDUCATION Office of Educational Research and Improvement EDUCATIONAL RESOURCES INFORMATION

CENTER (ERIC) This document has been reproduced as received from the person or organization originating it.

Minor changes have been made to improve reproduction quality.

Points of view or opinions stated in this document do not necessarily represent official OERI position or policy. ٠

2

PERMISSION TO REPRODUCE AND DISSEMINATE THIS MATERIAL HAS BEEN GRANTED BY

avids

TO THE EDUCATIONAL RESOURCES **INFORMATION CENTER (ERIC)**

1

EA

OFFICE OF EDUCATIONAL ACCOUNTABILITY University of Minnesota College of Education and Human Development University Technology Center 1313 Fifth Street SE, Suite B-1 Minneapolis, MN 55414-4504

The 1998 Minnesota Education Yearbook can be found on the Internet at the following address:

http://edpsy.coled.umn.edu/oea/

The 1998 Minnesota Education Yearbook has been produced by:

THE OFFICE OF EDUCATIONAL ACCOUNTABILITY College of Education and Human Development University of Minnesota

OEA Staff

Mark Davison, Ph.D., Director Ron N. Erickson, Ph.D., Associate Director Ernest C. Davenport, Ph.D., Associate Professor Nohoon Kwak, Ph.D., Statistician Margaret L. Irish, Editor/Graphic Designer

> John Bielinski, Graduate Assistant Heidi Danielson, Graduate Assistant Se-Kang Kim, Graduate Assistant Young Seok Seo, Graduate Assistant Mary Jo Smith, Graduate Assistant Stacia L. Wick, Graduate Assistant

Editorial Review Board

Jim Angermeyer, Bloomington Public Schools

Rolf Blank, Council of Chief State School Officers

Dick Guevermont, Financial Management Division, Minnesota Department of Children, Families & Learning

Carol Hokenson, Data Management Division, Minnesota Department of Children, Families & Learning

Van Mueller, University of Minnesota

Ed Roeber, Advanced Systems

3



1998 MINNESOTA EDUCATION YEARBOOK: TABLE OF CONTENTS

	List of Figures		i
	LIST OF TABLES		iii
	Executive Summary	vii	i– xi
Chapter 1	Introduction		1
	Making Change Work How Tests Have Changed	1 2–3	
Chapter 2	Schools in an Era of Reform		5
	Overview of the Profile of Learning	7–8	
Chapter 3	Educational Inputs and Processes		11
	School Finance	11	
	Teacher Characteristics	15	
	Student Characteristics	16	
	School Characteristics	18	
Chapter 4	PARTICIPATION AND COURSEWORK, ATTENDANCE,		
	and Graduation		21
	Secondary School Coursework	21	
	Profile of Learning Performance Package: Arts and Literature	21–24	
	State Course Requirements and National Recommendations	21	
	College Level Coursework for High School Students	23	
	Attendance	26	
	Profile of Learning Performance Package: Decision Making	27	
	Dropout Rates	28	
Chapter 5	Achievement		31
	Comparing the Achievement of Minnesota Students	31	
	10 Outer Mations Minnecota Comprehensive Assessments	31	
	Comparing the Achievement of Minnesota Students		
	to the Nation	35	
	Student Performance in the Minnesota Achievement		
	Testing Programs	39	
	Profile of Learning Performance Package: Scientific Application	39-40	
	Equity and Excellence Across Gender and Ethnicity	44	



Chapter 6	ER 6 FINDINGS AND CONCLUSIONS		51
	Educational Choices	51	
	Profile of Learning Performance Package: Math Applications	51- 54	
	Educational Equity	52	
	Excellence	53	
Appendix A	Minnesota Comprehensive Assessments and		
	Basic Standards Tests with Results Broken		
	Out by Category		57
Appendix B	Content and Performance Standards		71
Appendix C	Glossary of Terms		75
	List of References		87





1998 MINNESOTA EDUCATION YEARBOOK: LIST OF FIGURES

		~
Figure 2.1	Statewide Assessments by Content Area	5
Figure 2.2	Statewide Assessments by Grade Level	6
Figure 2.3	Educational Indicators Covered in Chapters 3 – 5	9
Figure 3.1	Expenditures Per Pupil in Average Daily Attendance: Minnesota and the National Average	12
Figure 3.2	Proportion of School Funding Received Through Federal, State, Local, and Private Funding Sources, for Minnesota and the Nation	12
Figure 3.3	Per Pupil Operating Expenditure by Minnesota Region	13
Figure 3.4	Percentage of Students with F/R Lunch, and Students with LEP, by Minnesota Region	13
Figure 3.5	Percentage of Students in Each Ethnic Group, by Minnesota Region	17
Figure 4.1	Percentage of Students Having Completed the ACT Recommended Core Academic Preparation for the Years 1987–98	22
Figure 4.2	Percentage of Minnesota ACT Test-takers Having the Recommended Core Academic Preparation, by Ethnic Group (1997–98)	23
Figure 4.3	Number of Students Participating in Post Secondary Enrollment Options Programs on Post-secondary Campuses for the Years 1992-97	24
Figure 4.4	Percentage of Eighth Grade Students Taking Algebra and High School Students Taking Advanced Mathematics and Science	25
Figure 5.1	1992-94 NAEP Grade 4 Reading Results for Minnesota and the Nation	34
Figure 5.2	1994 NAEP Grade 4 Reading: Percent Proficient by Subgroup	34
Figure 5.3	1992-96 NAEP Grade 4 Mathematics Results for Minnesota and the Nation	34
Figure 5.4	1996 NAEP Grade 4 Mathematics: Percent Proficient by Subgroup	35
Figure 5.5	1990–92–96 Grade 8 Mathematics Results for Minnesota and the Nation	35
Figure 5.6	1996 NAEP Grade 8 Mathematics: Percent Proficient by Subgroup	36
Figure 5.7	1996 NAEP Grade 8 Science: Percent Proficient by Subgroup	36
Figure 5.8	Grade 3 Percentage of Students at or above Level II and Level III for Reading and Mathematics, by Gender	42
Figure 5.9	Grade 5: Percentage of Students at or above Level II and Level III for Reading, Mathematics, and Writing, by Gender	43
Figure 5.10	Grade 8: Percentage of Students Meeting High School Minimum Standard for Reading and Mathematics, by Gender	43
Figure 5.11	Grade 3: Percentage of Students at or above Level II and Level III for Reading and Mathematics, by Ethnicity	44

6



Figure 5.12	Grade 5: Percentage of Students at or above Level II and Level III for Reading, Mathematics, and Writing, by Ethnicity	44
Figure 5.13	Grade 8: Percentage of Students Meeting Basic Standard for Reading and Mathematics, by Ethnicity	44
Figure 5.14	Grade 3: Percentage of Students (with and without LEP) at or above Level II for Reading and Mathematics, by Ethnicity	45
Figure 5.15	Grade 5: Percentage of Students (with and without LEP) at or above Level II for Reading, Mathematics, and Writing, by Ethnicity	45
Figure 5.16	Grade 8: Percentage of Asian and Hispanic Students (with and without LEP) Meeting Basic Standard for Reading and Mathematics	46
Figure 5.17	Minnesota and National ACT Composite Scores, 1988–98	48
Figure 5.18	1997–98 Minnesota ACT Composite Scores, by Gender and Ethnicity	48



,

x



1998 MINNESOTA EDUCATION YEARBOOK: LIST OF TABLES

Table 3.1	1996–97 Per Pupil Operating Expenditures in Dollars and Percent of Total	14
Table 3.2	1996–97 Minnesota Teachers Profile: Full-time Teachers (100% FTE)	15
Table 3.3	Number of Students for Each Ethnic and Gender Group, by Grade, for the 1997–98 School Year	17
Table 3.4	1995–96 National Pupil-Teacher Ratios	18
Table 3.5	1996–97 Minnesota Pupil-Teacher Ratios	19
Table 3.6	1996–97 Minnesota Pupil-Staff Ratios	20
Table 4.1	Recommended Course Credit Requirements	22
Table 4.2	Growth in the Advanced Placement Program	24
Table 4.3	Percentage of Students in Attendance for Third, Fifth, Eighth, and Twelfth Grades	25
Table 4.4	Four-year Graduation and Dropout Rates	26
Table 5.1	IEA Fourth Grade Reading Achievement: Average Score	30
Table 5.2	IEA Ninth Grade Reading Achievement: Average Score	30
Table 5.3	Minnesota Fourth Grade Mathematics Performance in TIMSS, Compared to the U.S. and Other Nations	31
Table 5.4	Minnesota Eighth Grade Mathematics Performance in TIMSS, Compared to the U.S. and Other Nations	31
Table 5.5	Minnesota Fourth Grade Science Performance in TIMSS, Compared to the U.S. and Other Nations	32
Table 5.6	Minnesota Eighth Grade Mathematics Performance in TIMSS, Compared to the U.S. and Other Nations	33
Table 5.7	1998 Grade 3: <i>Minnesota Comprehensive Assessment</i> Results in Reading, for All Public School Students Tested	39
Table 5.8	1998 Grade 3: <i>Minnesota Comprehensive Assessment</i> Results in Mathematics, for All Public School Students Tested	39
Table 5.9	1998 Grade 5: <i>Minnesota Comprehensive Assessment</i> Results in Reading, for All Public School Students Tested	40
Table 5.10	1998 Grade 5: <i>Minnesota Comprehensive Assessment</i> Results in Mathematics, for All Public School Students Tested	40
Table 5.11	1998 Grade 5: <i>Minnesota Comprehensive Assessment</i> Results in Writing, for All Public School Students Tested	41
Table 5.12	1998 Grade 8: <i>Basic Standards Test</i> Results in Reading, for All Public School Students Tested	41



Table 5.13	1998 Grade 8: <i>Minnesota Basic Standards Test</i> Results in Mathematics, for All Public School Students Tested	
Table A.1	1998 Grade 3: <i>Minnesota Comprehensive Assessment</i> Results in Reading, for All Public School Students Tested, Except those with Limited English Proficiency	56
Table A.2	1998 Grade 3: <i>Minnesota Comprehensive Assessment</i> Results in Reading, for All Public School Students Tested, Except those New to Their District Since January 1, 1997	56
Table A.3	1998 Grade 3: <i>Minnesota Comprehensive Assessment</i> Results in Reading, for All Public School Students Tested, Except those in Special Education	57
Table A.4	1998 Grade 3: <i>Minnesota Comprehensive Assessment</i> Results in Mathematics, for All Public School Students Tested, Except those with Limited English Profi- ciency	57
Table A.5	1998 Grade 3: <i>Minnesota Comprehensive Assessment</i> Results in Mathematics, for All Public School Students Tested, Except those New to Their District Since January 1, 1997	58
Table A.6	1998 Grade 3: <i>Minnesota Comprehensive Assessment</i> Results in Mathematics, for All Public School Students Tested, Except those in Special Education	58
Table A.7	1998 Grade 5: <i>Minnesota Comprehensive Assessment</i> Results in Reading, for All Public School Students Tested, Except those with Limited English Profi- ciency	59
Table A.8	1998 Grade 5: <i>Minnesota Comprehensive Assessment</i> Results in Reading, for All Public School Students Tested, Except those New to Their District Since January 1, 1997	59
Table A.9	1998 Grade 5: <i>Minnesota Comprehensive Assessment</i> Results in Reading, for All Public School Students Tested, Except those in Special Education	60
Table A.10	1998 Grade 5: <i>Minnesota Comprehensive Assessment</i> Results in Mathematics, for All Public School Students Tested, Except those with Limited English Profi- ciency	60
Table A.11	1998 Grade 5: <i>Minnesota Comprehensive Assessment</i> Results in Mathematics, for All Public School Students Tested, Except those New to Their District Since January 1, 1997	61
Table A.12	1998 Grade 5: <i>Minnesota Comprehensive Assessment</i> Results in Mathematics, for All Public School Students Tested, Except those in Special Education	61
Table A.13	1998 Grade 5: <i>Minnesota Comprehensive Assessment</i> Results in Writing, for All Public School Students Tested, Except those with Limited English Profi- ciency	62
Table A.14	1998 Grade 5: <i>Minnesota Comprehensive Assessment</i> Results in Writing, for All Public School Students Tested, Except those New to Their District Since January 1, 1997	62
Table A.15	1998 Grade 5: <i>Minnesota Comprehensive Assessment</i> Results in Writing, for All Public School Students Tested, Except those in Special Education	63
Table A.16	1998 Grade 8: <i>Basic Standards Test</i> Results in Reading, for All Public School Students Tested, Except those with Limited English Proficiency	63



9

Table A.17	1998 Grade 8: <i>Basic Standards Test</i> Results in Reading, for All Public School Students Tested, Except those New to Their District Since January 1, 1997	64
Table A.18	1998 Grade 8: <i>Basic Standards Test</i> Results in Reading, for All Public School Students Tested, Except those in Special Education	64
Table A.19	1998 Grade 8: <i>Basic Standards Test</i> Results in Mathematics, for All Public School Students Tested, Except those with Limited English Proficiency	65
Table A.20	1998 Grade 8: <i>Basic Standards Test</i> Results in Mathematics, for All Public School Students Tested, Except those New to Their District Since January 1, 1997	65
Table A.21	1998 Grade 8: <i>Basic Standards Test</i> Results in Mathematics, for All Public School Students Tested, Except those in Special Education	66



.

· .



	The 1997 Omnibus Education Bill authorized the formation of an independent Office of Educational Accountability (OEA) to advise the education committees of the Minnesota Legislature and the Commissioner of the Department of Children, Families and Learning, at least on a biennial basis, on the degree to which the statewide educational account- ability and reporting system includes a comprehensive assessment framework that measures school accountability for students achieving the goals described in the state's results-oriented Graduation Rule. The University of Minnesota's College of Education and Human Development was contracted to establish the OEA in early 1998.
	The mission of the OEA is to analyze and periodically report on the needs of students and the condition of K-12 education in Minnesota, as reflected in a comprehensive set of indicators. Our reporting will be designed to inform and facilitate the improvement of elementary and secondary education on a statewide basis. It is our hope that the OEA will play an important role in enhancing the capacity of local schools and school districts to use performance information in their own continuous improvement efforts.
	The educational system can be likened to a forest in which the districts and schools are the trees. This initial report attempts to examine the forest, rather than each particular tree. It is designed to be the first in a series of annual reports charting the progress of Minnesota's educational system. Data on individual schools and districts appears on the website of the Minnesota Department of Children, Families and Learning at: <u>http://cfl.state.mn.us</u> .
Minnesota: A State of Reform	Our state's current educational reform initiative, commonly referred to as the <i>Graduation Rule</i> , has its roots in at least two other major policy initiatives over the past twenty-five years (Mazzoni, 1993). The first was the school finance reform initiative of 1971, known as the "Minnesota Miracle." At that time, Minnesota began financing elementary and secondary education through the use of a minimum foundation program, which was intended to provide adequate funding for the state's public education system while compensating for variations in property wealth among the state's hundreds of local school districts.
	The second major reform period began in the mid-1980s, with several pieces of successful legislation focused on school choice. The Postsecondary Enroll- ment Options, the "open enrollment" options program, and the formation of charter schools are all examples of Minnesota's dedication toward providing

Our current drive toward standards-based reform began in 1992, when the Legislature approved the *Graduation Rule*, comprised of: (1) the *Basic Standards*, representing the minimum skills required of all students for high school graduation; and (2) a sequence of *Preparatory Standards* for grades K-8 and *High Standards* for grades 9-12 that students are expected to achieve before leaving school. New assessments followed the adoption of the Rule to evaluate



choice.

student progress toward these new standards. The *Minnesota Comprehensive Assessments* at Grades 3 and 5 assess students' mastery of the Preparatory Standards at those grade levels. The *Basic Standards Tests* of reading, mathematics and writing are now offered to students beginning in eighth grade, and must be passed before graduating from high school. And finally, performance-based assessments, or "packages" linked to the Preparatory and High standards are completed by students throughout the grades. These classroombased assessments provide information for a *Profile of Learning* to be developed for each individual student.

THE ORGANIZATION OF The 1998 Yearbook is supported by a review of the research into what makes a difference in our attempts to provide quality education to all our children and THE REPORT youth, and builds upon earlier efforts in our state to identify the most important educational inputs, processes and outcomes on which to focus our collective work at helping students reach higher levels of achievement (e.g., Bruininks, Bielinski, Danielson, Davison, Erickson, Lock, Lydell, Norline-Weaver, Seppanen, Thurlow, & Ysseldyke, 1996). Educational inputs are commonly thought of as the human, financial and material resources brought together in teaching and learning. Indicators of educational inputs featured within this report include school finance information, teacher characteristics, student characteristics, and staffing ratios. Educational processes define the interactions of these inputs-shifting from the question of "what is present in the school?" to "what is happening in the school?" The process indicators included in this report include student participation in challenging high school courses, attendance patterns, and graduation rates. While we recognize graduation is commonly considered an educational goal, it has been placed within this report as a process indicator to reflect its critical importance to later achievement and student success after high school. Educational outcomes in this report are framed in terms of student achievement, with the performance of Minnesota students benchmarked at the international, national and statewide levels. Also included within this outcome area are indicators of the academic preparedness of college-bound students.

Our Findings on Educational Inputs A number of data sources at both the national and state level were reviewed in compiling the indicators of educational inputs featured within this report. The report focuses primary attention on (1) the financial resources brought to bear on our state's public education system, (2) several characteristics of our current K-12 teaching force; (3) the changing demographics of our student population; and (4) the ratio of teachers and staff to students in Minnesota schools. Among our findings:

- Minnesota's per pupil funding (adjusted for inflation) has increased over time, but at a slower rate than for the nation as a whole. During the 1959-60 school year, Minnesota spent \$2,177, or 13% over the national average of \$1,920. In that year, Minnesota ranked tenth nationally in per pupil spending. By 1989-90, Minnesota's per pupil funding had fallen below the national average of \$5,899, and did not again exceed the national average until the 1994-95 school year.
- The age distribution of Minnesota teachers has shown a dramatic change in the last ten years. In 1987-88, a majority of teachers fell in the 35–44 age bracket (40%). Today, the largest group (38%) of teachers is between 45 and 54 years old (Minnesota Department of Children, Families & Learning, 1997).



• Minority students are characterized by higher rates of poverty, limited English proficiency, and disability. If increasingly greater numbers of Minnesota's minority students are faced with these same challenges, higher rates of achievement and graduation will require new strategies for instruction and curriculum.

Student achievement can never be realized without the active participation of students in the learning process. This report investigated several dimensions of student participation in school, including attendance rates, graduation rates, and measures of how well our students are taking advantage of course opportunities. Among our findings:

- From 1987-88 to 1997-98, the proportion of college-bound ACT testtakers completing a recommended basic academic core for postsecondary success has risen from 55% to 73%. This rise in preparation is extremely encouraging, but the uneven distribution of that preparation over Minnesota's ethnic groups is not.
- Increasingly, high school students in Minnesota can elect to take college level work in Advanced Placement, International Baccalaureate, and Postsecondary Enrollment Options programs, and they have been choosing to do so in increasing numbers.
- There is a decline in school attendance from fifth to twelfth grade, which may reflect the declining attendance of some students who eventually drop out of school. Stemming the decline in attendance during the secondary grades may be essential to cutting the dropout rate.
- For the state as a whole, 78% of the 1994 ninth graders in the study graduated from a Minnesota high school four years later. Eleven percent left high school, and another 11% were still enrolled in high school but had not yet completed work for their diploma.

Our review of the achievement performance of Minnesota students in national and international achievement testing was limited by the subject areas assessed and the grade levels included in such studies. State level achievement data was readily available for students in Grades 3, 5, and 8, due to the expansion of Minnesota's statewide testing programs. Also included in our examination was the performance of college-bound students in the ACT Assessment. Our review of the performance of Minnesota students in these various assessment programs leads us to the following conclusions about the performance of Minnesota students:

REGARDING NATIONAL AND INTERNATIONAL COMPARISONS OF READING ACHIEVEMENT:

• Despite our state's decline in reading scores in the National Assessment of Educational Progress from 1992 to 1994, only Maine clearly outscored Minnesota students, and the reading achievement of our fourth graders remains near the top nationally. In a recent international study of reading, the United States average was significantly exceeded by that of only one country. Therefore, we can be relatively sure that Minnesota students' performance in reading would be internationally very competitive, at least at the fourth grade level.



Our Findings on Educational Outcomes: The Achievement of Our Students in International, National and Statewide Testing



13

REGARDING NATIONAL AND INTERNATIONAL COMPARISONS OF MATHEMATICS ACHIEVEMENT:

• At both fourth and eighth grade, the achievement of Minnesota students compared very favorably to that of other states in the U.S., but in mathematics comparing favorably to other states does not make the performance internationally competitive. Based on our performance in an international mathematics assessment and an analysis of our curriculum, a re-analysis of our mathematics curriculum seems warranted.

REGARDING NATIONAL AND INTERNATIONAL COMPARISONS OF SCIENCE ACHIEVEMENT:

• Minnesota's public school eighth graders achieved an average NAEP science scale score of 159, well above the national average scale score of 148, and significantly exceeded by only two other participating states (Maine and North Dakota). In a recent international science assessment, only eighth grade students in Singapore significantly outscored Minnesota students.

REGARDING THE MINNESOTA STATEWIDE TESTING PROGRAM:

- The large number of eighth grade students failing to meet the high school minimum on the *Basic Standards Tests* remains one of the state's most pressing educational problems. By high school, students should move beyond basic skills to more challenging outcomes in core subjects, in vocational training, and in the arts. The uneven achievement of student ethnic groups in the *Basic Standards Tests* needs to be systematically addressed.
- The largest gender difference is in fifth grade writing, where 87% of girls and 74% of boys scored at or above Level II; 52% of girls and 32% of boys displayed the solid mastery of challenging material characteristic of Level III.
- The large differences among various ethnic groups in reading, writing, and math across all grades are incompatible with society's diversity goals and with our state's drive toward higher standards for all students. Any approach toward addressing the inequity of achievement among Minnesota's minority students must take into account various factors: lower attendance, higher rates of poverty, greater mobility, and higher proportions of students with disabilities and limited English proficiency.
- The achievement levels in charter schools pose a challenge to parents and to the agencies which charter those schools.

REGARDING COLLEGE ADMISSIONS TESTING:

- While ACT college admissions scores for Minnesota declined through the 1980s, they have risen through the 1990s. Of the states with 50% or more of its students taking the ACT, only one state had an overall average scale score higher than Minnesota in the 1998 ACT administration.
- With the emphasis on minimum competency engendered by the *Basic Standards Tests*, there has been concern that the education of high ability students may be neglected. Our state's rising ACT scores and the increasingly higher course preparation of college bound students should help allay this concern.



References:

Bruininks, R., Bielinski, J., Danielson, H., Davison, M., Erickson, R., Lock, M., Lydell, L., Norline-Weaver, J., Seppanen, P., Thurlow, M., & Ysseldyke, J. (1996). *Minnesota educational accountability reporting system: Feasibility and design study.* (Vol. 1). Minneapolis: University of Minnesota: College of Education and Human Development.

Mazzoni, T. (1993). The changing politics of state education policy making: A 20-year Minnesota perspective. *Educational Evaluation and Policy Analysis*, 15(4), 357-379.

Minnesota Department of Children, Families & Learning. (1997). School District Profiles 1996-97. St. Paul, MN: Division of Management Assistance.





CHAPTER 1 INTRODUCTION

The state of Minnesota is involved in a continuing effort to evaluate and improve the quality of education for all children.

To understand the complexity of such an effort, one must remember that Minnesota's public system enrolls 840,000 students attending 2,000 school sites distributed among 350 districts. In addition, schools offer a variety of programs designed to address the needs of students, such as special education programs, orEnglish instruction for children with limited English proficiency; and any of these programs may be administered separately from the "regular" educational curriculum.

The educational system can be likened to a forest in which the districts and schools are the trees. This report looks at the forest rather than at each particular tree: it examines the entire educational system in Minnesota, not individual district or school performance. It is designed to be the first in a series of annual reports charting the progress of Minnesota's educational system. For data on particular schools and districts, the reader should consult the Minnesota Department of Children, Families, and Learning, at:

http://www.cfl.state.mn.us

This report covers the outcomes currently in Minnesota's educational reporting system, the reforms designed to enhance those outcomes, and the investments we are making to achieve them.

Chapter 2 describes Minnesota's reform efforts and briefly explains how they are different from reform efforts in other states. The reforms adopted by other states can suggest modifications to our own current efforts.

Chapter 3 describes educational inputs: school finances, the teaching faculty, student characteristics, and school characteristics. These can form the basis for a discussion of whether

there are adequate resources to reach the desired outcomes.

Chapter 4 addresses issues of student participation, including courses studied, attendance, high school graduation rates, and dropping out. These behaviors are of interest for their own sake, and because they contribute to achievement.

Chapter 5 describes achievement, beginning with national and international data benchmarking the achievement of Minnesota students against that of other students and countries. It then moves to a more detailed analysis of achievement within Minnesota as reflected in the statewide tests given to all students at third, fifth, and eighth grades. Then it describes the performance of college-bound Minnesota students on the ACT Assessment (ACT).

We summarize our major findings and conclusions in Chapter 6.

The vision of Minnesota's account-

MAKING CHANGE WORK

The theme of this report is change: how and what is changing in Minnesota's education system. The Minnesota Education Yearbook is designed to report on the status of education in Minnesota, and to provide information for policymakers and other stakeholders that will facilitate the process of decision making.

Much of this information is very quantitative: it measures how much, how often, how many. It describes the exact limits of the research that has been done. It takes note of where exact information has not been collected, and it enables the reader to compare pieces of information by making sure that, figuratively speaking, apples are compared to apples rather than to oranges.

Quantitative information is extremely useful. It can help us to identify exactly which students do well or poorly, help us to see whether inputs such as dollars are having the desired effect on student performance, and tell where our perceptions of the public education system do not match the reality.

However, it may not be the best way of conveying a sense of the totality of the circumstances surrounding students' or teachers' experiences with education.

The shaded boxes that appear throughout this report will help to supply some of this type of information. We will include comparisons of tests from the 1930s and today, explaining some of the major differences in what is tested and the reasons for the change. We will also include examples of real-life student and teacher experiences with the Profile of Learning's High Standards. These sections illustrate how the Profile of Learning functions in an actual school situation with real students: what a project assignment looks like, how a student might approach each kind of project, and how projects might typically be evaluated.



HOW TESTS HAVE CHANGED

In the late 1930s, Minnesota had statewide examinations in both elementary and secondary grades, administered by the Minnesota State Board of Education. Compared to our present focus on reading, mathematics, and writing, these earlier tests covered several different content areas, including English, Mathematics, Science, Geography, and Social Studies.

How does the content of yesterday's eighth grade General Mathematics Test compare with today's Basic Standards Test?

The 1938 *General Mathematics II* test was administered on Tuesday, May 31 starting at 10:15 and ran for two hours. As compared to today's *Basic Skills Test* in Mathematics, which students first take in eighth grade, the old test had more computation problems. For example:

Subtract:	8973	Divide:	3.2)41.6
	- 4316		

The 1938 test also contained definitional questions. For instance, the student had to select one term (from a list of 15 terms) that means "Profits divided among stockholders."

There were also applied problems similar to the ones which dominate today's test. For example:

Mr. Jones receives a salary check of \$75 on the 1st and 15th of each month. Items spent were:

Food \$20	Clothes \$5.50
Rent \$39	Savings \$15
Gas \$1.24	Books and shows \$3.25
Telephone \$2.75	Meat \$3.00
Electricity \$1.75	Milk \$4.00

What is Mr. Jones' monthly salary?

What per cent of his salary was spent on rent? _

Find the area and the perimeter of a square 40 rods on a side. How many acres are there in the square plot?

(cont'd on p. 3)

ability and reporting system covers a broad array of indicators spanning education from pre-kindergarten to post-secondary levels. Of necessity, this reporting system will continue to develop over time, since some indicators are not available for all schools before certain dates, and other information has never been collected at all.

Future reports will cover an expanded set of indicators. For instance, early childhood education and school readiness will receive more attention in upcoming reports. Postsecondary outcomes, vocational education, arts education, school safety, and adult education are potential topics for future reports. Graduation rates and educational attainment data in Chapter 5 suggest more attention to adult education in future reports.

Besides reporting on these additional indicators, future reports will also provide trend information on key indicators to provide a better picture of our continuing progress. In compliance with statutory requirements (Minnesota Statute, \$120B.30, Subdivision 1b [1998]) data on student performance featured in this initial report will serve as a baseline against which future performance can be evaluated.

Educational accountability in Minnesota is part of the state's larger, system-wide reform effort. As in many states, our reform effort is standards-driven. That is, the state and many districts began by establishing educational standards—



How TESTS HAVE CHANGED (cont'd from p. 2)

Today, because of the widespread availability of calculators, there is less need to do mathematical calculations by hand. Students can use calculators for all but the estimation portion of the *Basic Standards Test* in Mathematics. There is more emphasis on statistics. Students must still know definitions, but they must be able to apply them practically, as in this example:

Angelo saw the figures below on his way to school. In which figure did he see the shape of a pyramid?



Most questions require students to apply their mathematical skills. These simple statistics problems are typical:

The circle graph below shows how every dollar could be spent from the family budget. How much money is spent for clothing out of every dollar of income?



Hand calculation problems have been replaced by ones in which students must estimate the correct answer without the aid of a calculator:

The fuel cost of driving a truck is about 12 cents a mile. Approximately how much does it cost to drive a truck 96 miles?

A.	\$ 0.90
B.	\$ 6.00
C.	\$ 8.00
D.	\$12.00

In the spinner game below, you win if you spin an odd number, and you lose if you spin an even number. What is the probability of winning?



statements of what students should know and be able to do at various points in the educational process.

These standards have served as guides for curriculum reform, instruction, and assessment. Most importantly, for our purposes, these standards have guided our selection of outcome indicators — measures of what students have learned and attained.

All education occurs within the

13

context of a variety of circumstances. In order to understand what is happening now in our educational system, it is helpful to take a backward look, toward the history of educational reform in Minnesota, and in the United States.





THE CALL FOR EDUCATIONAL REFORM

ike most other states, Minnesota has embarked on a major deffort to transform its system of public education so that tomorrow's students are prepared to meet the challenges of the 21st century. Numerous federal education initiatives, coupled with growing public dissatisfaction over our country's present system of schooling, have fueled a serious reevaluation of our traditional notions of educational policies and practices. Communities of stakeholders at all levels are being challenged to think differently about virtually every aspect of elementary and secondary education. On a national level, several factors appear to have inspired our current interest in reform:

- National reports of declines in educational achievement
- Reports that American students' mathematics and science achievement test scores are lower than those of students from other industrialized countries
- Taxpayer concerns about whether increased spending is improving the quality and the outcomes of education
- A strongly-held perception that educational reform is the pivot of economic development
- Increasing demands that education respond to our increasing orientation toward information and technology
- Strong concerns about the poor performance of students in our core cities

• A focus on educational equity and the achievement of female, minority, and disadvantaged students.

Identifying the exact origin of these various concerns is difficult. However, in the opinion of many, the major flashpoint of public concern was the declaration in *A Nation at Risk* (National Commission on Excellence in Education, 1983) that our country's educational problems were undermining the nation's future. The resulting national discussion challenged states to improve the quality of their educational systems and, at the same time, their students' ability to compete successfully in a global market.

STANDARDS: THE COMMON THREAD

If there is a common theme among the various state-level efforts to meet these challenges, it is undoubtedly in the articulation of *standards*, the knowledge and skills we expect our students to have. Many would argue that no other aspect of educational reform can advance without first specifically defining our expectations for learners. Among educational reformers, two different types of standards are most commonly discussed: *content* and *performance* standards.

Content standards articulate and define the knowledge and skills that are expected of students in different content areas, such as English, mathematics, and science. Performance standards are the established benchmarks of student performance that are linked to assessments of the content standards. While content standards answer the question, "What is it we want our students to know and to do?" performance standards address the issue of "How well do we expect them to know it or do it?" Many states have identified performance standards not only for students, but for schools and local school districts as well.

In 1997, the American Federation of Teachers' report, *Making Standards Matter* (1997a), stated that 49 states



Figure 2.1 Statewide Assessments by Content Area







are either developing or implementing a system of academic standards for students. Thirty-nine of these have developed new standards or revised their current ones within the past year. In almost all cases, states have focused on developing content standards in the four curricular content areas of language arts, mathematics, science, and social studies (Council of Chief State School Officers, 1998). These standards are the product of extensive input from various constituencies representing students, parents, teachers, school administrators, postsecondary institutions, businesses, and other community agencies.

Assessment: The Measure of Our Success

The setting of higher academic standards has guided many other components of state educational reform efforts, including school improvement programs, increased requirements for high school course work, staff development and training, and the addition or revision of school accreditation processes.

But nowhere is the influence of standard setting more readily seen than in the rapid expansion of statewide testing programs. In its latest national survey of statewide assessment programs, the Council of Chief State School Officers (Bond, Roeber, & Connealy, 1998a) found that the traditional subject areas of language arts, mathematics, writing, science, and social studies continue to be the most commonly assessed content areas in statewide assessment programs (see Figure 2.1, p. 5).

The Council's 1997 survey also revealed that states are using assessment at all grade levels, with higher rates of testing found at Grades 4, 8, and 11 (see Figure 2.2, above). Tests at these grade levels are used to measure overall student performance at the elementary, middle school, and high school levels.

Consequences in Accountability Systems

Along with higher standards, and assessment programs realigned to these standards, many would argue that real educational reform cannot be realized unless there are clear consequences attached to student performance. These consequences are often described as "stakes" for individual students, schools, or districts. The terms "low stakes" and "high stakes" express the varying levels of risk

being placed on those responsible for the expected results.

In regard to *student* accountability, twenty-two states currently make, or plan to make, a high school diploma dependent upon a student's performance in particular tests commonly referred to as *graduation*, *proficiency*, or *exit* exams. Along with Minnesota, these states include Alabama, Alaska, Arizona, Florida, Georgia, Hawaii, Indiana, Louisiana, Maryland, Massachusetts, Mississippi, Nevada, New Jersey, New Mexico, New York, North Carolina, Ohio, South Carolina, Tennessee, Texas, and Virginia.

A relatively small number of states have called for school or district accountability by mandating consequences for schools or districts based on their students' performance (Bond, Roeber, & Connealy, 1998b). Some of the most widely adopted measures include:

- Probations or watch lists (14 states). Schools or districts with low performance and little progress are given cautionary notice that improvements must be made within a certain time frame to avoid harsher consequences. Watch lists are usually made available to the public.
- Accreditation loss (11 states). A lack of adequate performance or progress in meeting goals leads to losing status granted through state accreditation agencies.
- Takeover by state agencies (11 states). A very high-stakes consequence often involving removal of local school board and top district administrators, with temporary governance provided by state educational officials.
- Funding gains (9 states). In this case, monies are distributed to local districts, principals, or school



staff as the result of meeting or exceeding expected student performance goals.

- Regulatory waivers (9 states). State educational agencies relieve local districts or schools from certain regulatory requirements in response to positive student performance.
- Funding loss (3 states). A high stakes consequence for lowperforming schools and districts, involving the loss of state financial aid.
- Dissolution (4 states). This extremely high-stakes consequence results in the entire school or district being shut down under the supervision of the state's educational agency.

Researchers who have studied such reward and sanction programs have suggested that performance-based rewards do not always provide sufficient motivation for school improvements. In their study of four states using rewards for school improvement, King and Mathers (1996) found that intrinsic rewards, nonmonetary recognition of improvement, and public awareness of poor performance were more motivating than money.

THE BOTTOM LINE: ACCOUNTABILITY

Legislators and other policymakers at state and federal levels increasingly question whether greater regulation of local school processes-the day-today activities in which teachers and stu-dents, schools and districts engage- ensure excellence. Years of process monitoring by state and

federal education agencies have not vielded consistent and high levels of student achievement. Thus, attention has begun to focus on the results of our educational endeavors, rather than on the specific methods used to meet our goals.

With the advent of new governing structures like site-based management, choice options, and charter schools, schools have been given increasingly more autonomy in developing specific curricula, programs, and instructional approaches. But this autonomy has a corresponding expectation-namely, that school systems, schools, and students will now be held accountable for meeting certain articulated results. It is a political tradeoff that grants an increase in local control of operations, but requires schools to meet

OVERVIEW OF THE PROFILE OF LEARNING

The Profile of Learning is a framework of Standards ensure that students have standards-based education and applied learning. It is designed to assure accountability for individual student results and to ensure that students actually experience the learning that is necessary in order to function successfully in postsecondary education and in the work world.

In the Profile of Learning, experiences are organized into ten Learning Areas (see Learning Areas table, p. 8). The Learning Areas represent complex skills and processes that build sequentially through the primary, intermediate, middle, and high school levels. While the categorization of requirements in the Profile of Learning is somewhat different from the traditional subject categories employed in districts in improving achievement over most schools, there is a clear relationship. For example, Learning Areas 4 and 6 focus on math and science respectively. Learning Areas 1 and 2 elaborate the essential components of communication: reading, writing, and speaking.

Within each Learning Area, there are specific standards that must be met. The term, "standard," refers to what students should know and be able to do. The standards required in K-8 are called Preparatory Standards. These Preparatory

sufficient content background and skills to pursue the High Standards in high school. For example, in Learning Area 2, high school students may choose to emphasize academic writing or technical writing, after having completed preparatory standards in both kinds of writing. (See CFL web site http://cfl.state.mn.us/ grad/highstandards.htm .)

To evaluate student progress on the Preparatory Standards, Minnesota teachers, national testing experts, and the Department of Children, Families, and Learning (CFL) developed the Minnesota Comprehensive Assessments (MCA) for third and fifth graders. These tests also measure the success of schools and time. They are used to generate information for school improvement and accountability, and they allow for comparison of schools and districts.

To graduate from high school, students must meet 24 of 48 possible High Standards. Students demonstrate what they know and are able to do by completing several assignments. These assignments or "performance packages" (developed locally based on Department of Children, Families, and Learning

models) are designed to assess a student's ability to meet the standard. In the course of this report, we will look at several performance packages.

The score a student receives on a content standard is determined by the teacher or school district designee, after taking into account the level of accomplishment at which a student performs on a series of tasks (the performance package) representing an entire content standard. Within the performance packages, checklists provide feedback to the students about their work relative to the content standard. The progress guidelines consist of a Y (yes), meaning that the student has met the performance task; or an N (no) if the student has not met the performance task. Once the student has attained all Y's on the performance tasks (satisfying the content standard requirements), their work will be evaluated. The scoring criteria for the completed content standard is based on a four point scale:

4 - Exemplary: Indicates evidence of student learning in all parts of the standard at a level that exceeds expectations by using and applying knowledge consistently in new and insightful ways.

(cont'd on p. 8)



more externally imposed expectations.

Minnesota's Legacy of Educational Innovation

Educational reform calls for innovation, a term that is often used to

describe Minnesota's educational policies. Our state's current reform initiative, commonly referred to as the *Graduation Rule*, has been preceded by at least two other major policy initiatives over the past twenty-five years (Mazzoni, 1993). The first was the school finance reform initiative of 1971, known as the "Minnesota Miracle." At that time, Minnesota began financing elementary and secondary education through the use of a minimum foundation program, which was intended

(cont'd from p. 7)

3 - *Proficient:* Indicates evidence of student learning in all parts of the standard at a consistently proficient level.

2 - *Novice:* Indicates evidence of student learning in all parts of the standard at an adequate level some or all of the time.

1 - *Emerging*: Indicates evidence of student learning in all parts of the standard at a superficial level some or all of the time.

The Minnesota Graduation Rule recog-

nizes that, while all students need a comprehensive educational experience to prepare them for lifelong learning, people are different, having different skills, interests, and areas of strength and weakness. Therefore, the *Graduation Rule* does not demand that all students achieve outstanding levels of performance in all areas. Rather, individual achievement on content standards produces a student *profile*, indicating those areas and standards in which the student excels and those in which the student has not achieved at a high level.

The *Graduation Rule* also recognizes individual learning styles and preferences by allowing the achievement of High Standards in varied contexts, programs, courses, and learning environments. The *Graduation Rule* is also working to establish a consistent means of recording and reporting student results as scored against high quality examples of excellent achievement. This information – this *profile* – will help the student, and those who teach and employ the graduate later, to recognize both strengths and needs for further experiences and learning.

1. Read, View and Listen Read, view and listen to complex information in the English language	2. Write and Speak Write and speak effectively in the English language	3. Arts and Literature Apply and interpret artistic expression	4. Math Applications Solve problems by applying mathematics	5. Inquiry Conduct research and communicate findings
High School	High School	High School	High School	High School
Middle (6-8)	Middle (6-8)	Middle (6-8)	Middle (6-8)	Middle (6-8)
Intermediate (4-5)	Intermediate (4-5)	Intermediate (4-5)	Intermediate (4-5)	Intermediate (4-5)
Primary (K-3)	Primary (K-3)	Primary (K-3)	Primary (K-3)	Primary (K-3)

6. Scientific Application	7. People and Cultures	8. Decision Making	9. Resource Management	10. World Languages <i>Optional</i>
Understand and apply scientific concepts and methods	Understand interactions among people and cultures	Use information to make decisions	Manage resources for a household, community or government	Communicate in a language other than English
High School	High School	High School	High School	High School
Middle (6-8)	Middle (6-8)	Middle (6-8)	Middle (6-8)	Middle (6-8)
Intermediate (4-5)	Intermediate (4-5)	Intermediate (4-5)	Intermediate (4-5)	Intermediate (4-5)
Primary (K-3)	Primary (K-3)	Primary (K-3)	Primary (K-3)	Primary (K-3)



to provide adequate funding for the state's public education system while compensating for variations in property wealth among the state's hundreds of local school districts. Currently, Minnesota schools receive a higher proportion of state funding than do schools in many other states. During 1994–95, 52% of Minnesota's school operating expenses came from state sources, as compared to 47% nationally (Snyder, 1997).

The second major reform period began in the mid-1980s, with several pieces of successful legislation focused on school choice. In 1985, the Postsecondary Enrollment Options Act (PSEO) made it possible for Minnebegan during this time period as well. Although early efforts at reform were organized under the banner of Outcomes-Based Education (OBE), considerable resistance against OBE and its perceived intrusiveness on local instructional practices eventually shifted the initiative toward a focus on performance-based graduation standards.

In 1992, the Legislature approved a standards-based *Graduation Rule*, which included two distinct components: (1) the *Basic Standards*, representing the minimum skills required of all students for high school graduation; and (2) a sequence of *Preparatory Standards* for grades K–8 and *High* before they can receive their high school diplomas. In contrast to the measures of basic standards, "performance packages" are designed by classroom teachers to measure each student's attainment of the *Preparatory* and *High Standards*.

In 1997, additional legislation called for schools and districts to be evaluated on their ability to help students achieve the *Preparatory* and *High Standards* through the administration of a series of standardized assessments. Given in the 1997–98 school year for the first time, the *Minnesota Comprehensive Assessments* (MCAs) measure third and fifth graders' mastery of the *Preparatory Standards* in reading and mathematics.

sota high school juniors and seniors to enroll in courses or programs at eligible postsecondary institutions. By 1988, the state legislature had initiated the K-12 "open enrollment" options program, allowing all students the chance to attend schools outside of their resident district. In



The MCAs also measure fifth graders' writing skills. While no student consequences are attached to the MCA testing program, the scores show the public and policymakers how well schools and districts are doing at helping their students attain higher academic standards. The remain-

The remaining chapters of

1991, the ground-breaking legislation for the formation of charter schools was passed. Even before these initiatives, Minnesota had established itself as a proponent of school choice by allowing tax deductions for the educational expenses of all families, including tuition expenses of families with children in non-public schools.

Minnesota's present drive toward a results-oriented educational system

Standards for grades 9–12, which students are expected to achieve before leaving school.

The need to assess the minimum standards has since led to the establishment of *Basic Standards Tests* in reading, mathematics, and written composition for students at or above the eighth grade. With certain exceptions for students with limited English proficiency and students with disabilities, students must pass these exams this 1998 Yearbook categorize discussion around three interrelated components of our educational system (as shown in Figure 2.3).



CHAPTER 3 EDUCATIONAL INPUTS AND PROCESSES

INTRODUCTION

B ecause of its inherent value to our democracy and economic vitality, public education is one of the most important, and costly, enterprises of any state. In this chapter, we report on several of the characteristics of our schools and students that are brought together in the educational enterprise, and reflect on how the current status of each of these may be contributing to the results of our efforts.

The Relationship of Financial Resources and Achievement

Does the application of additional financial resources improve student success and achievement? This fundamental question, perennially posed by educational stakeholders at all levels, has no simple answer. Researchers vary widely in their conclusions about the effects of increased funding on achievement. Increased funding can buy smaller class size, higher teacher salaries, and more experienced teachers, all of which are associated with better outcomes (Greenwald, Hedges & Laine, 1996). But these factors, in and of themselves, do not assure improved student performance on the intended outcomes (Hanushek, 1989). The proper application of additional school finances appears to depend on three critical factors:

1. A clear articulation of the expected application of additional funds and the expected outcomes. Much of our inability to assess the impact of additional funding is based on the fact that the discretionary control exerted over additional funding by schools and districts often leads to a redirection of these appropriated funds to other areas in need. Such local reappropriation may well produce positive results, but in areas other than what was initially intended.

2. A careful consideration of the interactive effects among the various factors known to be associated with achievement. Many of the educational inputs known to be associated with higher levels of student achievement are interrelated, and must be carefully

> Minnesota's per pupil funding (adjusted for inflation) has increased over time, but at a slower rate than for the nation as a whole. During the 1959–60 school year, Minnesota spent \$2,177 per pupil, or 13% over the national average of \$1,920. In that year, Minnesota ranked tenth nationally in per pupil spending. By 1989–90, Minnesota's per pupil funding had fallen below the national average of \$5,899, and did not again exceed the national average until the 1994– 95 school year.

balanced if additional funds are to be applied effectively. For example, the positive effects of lowering class size in a school or district may be inadvertently diminished or lost if such a policy reduces the local agency's ability to hire more experienced teachers.

3. A perspective that views improvement in curriculum and instruction as the primary means through which positive changes can occur. Additional funding might best be understood as a secondary or enabling factor that can open the door for other changes more central to the mission of teaching and learning. Increased funding can only succeed when it leads to tangible improvements in how well teachers can teach and students can learn.

FINANCING PUBLIC EDUCATION IN MINNESOTA

Figure 3.1 (p. 12) reveals that Minnesota's per pupil funding (adjusted for inflation) has increased over time, but at a slower rate than for the nation as a whole. During the 1959–60 school year, Minnesota spent \$2,177 per pupil, or 13% over the national average of \$1,920. In that year, Minnesota ranked tenth nationally in per pupil spending. By 1989–90, Minnesota's per pupil funding had fallen below the national average of \$5,899, and did not again exceed the national average until the 1994–95 school year.

Examining the Sources of School Funding

A closer look at the sources of funding for our state's pre-K-12 educational system reveals some interesting differences between Minnesota and other states. Generally speaking, school revenues are predominantly gathered through three funding streams:

- Local sources, most commonly through property taxes
- State allocations
- Federal funding, which is typically targeted toward specific disadvantaged groups

Whenever school funding has come





to depend too heavily on locally generated funding sources, districts with high proportions of economically and educationally disadvantaged students have tended to have lower per pupil funding due to their lower property wealth. In recent years, many states have begun to fund proportionately more of their elementary and secondary programs through state allocations rather than local sources, in an attempt to alleviate this inequity.

The Minnesota and the national funding proportions for the 1994–95 school year, the last year for which comparative data for other states is available, are shown in Figure 3.2 (Snyder, 1997). Overall, Minnesota ranked 18th nationally on per pupil funding received from state allocations: 52.4% of Minnesota's funds came from state sources as compared to 46.8% nationally. In Minnesota, 39.4% came from local sources, as compared to 43.8% in other states. Private contributions were above the average; Minnesota was 14th out of 50 states. Private sources supported 3.7% of the cost of education in Minnesota, as compared to 2.7% nationally. Private sources include revenues from gifts, tuition, and fees from patrons. Finally, Minnesota received less support for education from the federal government than do most other states. Our federal contribution is 4.4% compared to 6.8% nationally, with Minnesota ranking 46th out of 50 (Snyder, 1997).

How Minnesota's Education Dollars are Spent

Increasingly, as Minnesota has moved toward statewide funding, it has addressed two types of disparities across districts: differences in local tax







25

bases and in the distribution of high need students across districts.

Differences in local tax bases have been addressed through the foundation formula, which is designed to provide a basic foundation of funding for all districts irrespective of local resources. It channels more state aid to districts with low residential and commercial tax bases.

Low levels of educational attainment are associated with factors such as poverty, limited English proficiency, and disabilities, and some districts in Minnesota have heavier concentrations of these factors than do others. Both federal and state funds have been used to address needs created

funds include compensatory education dollars (most heavily targeted to districts with heavy concentrations of poverty) and funding for students with limited English proficiency. To help prevent low achievement before it starts, state-funded learning readiness and first grade preparedness programs help students begin their education prepared to learn.

Like most other states, Minnesota is involved in a continuing debate as to whether the funding formulas adequately recognize disparities in local funding sources and the challenges faced by districts with high proportions of students with special needs. In any case, differences in funding levels across districts reflect, in part, state and federal efforts to address disparities in local funding capacities, the unequal distribution of at-risk students, and the drive to assure a basic education for all students. Differences in per pupil spending between districts (see Figure 3.3, below), particularly between Minneapolis/St. Paul and other districts, reflect the uneven distribution of students with special needs (see Figure 3.4, below right). It should be noted that we have not adjusted any figures for differences in costs of living.

Minnesota school expenditure data are reported in eleven categories (see Table 3.1, page 14). At a state level, regular classroom expenditures are allocated about half of the total expenditures. Exceptional instruction (non-regular instruction of students with special characteristics such as those with disabilities) account for approximately 15% of the total expenditures. When taken together, expenditures in the four instructional categories (regular, exceptional, vocational, and instructional support) account for approximately 70% of the total per pupil expenditure statewide, as compared to the average expenditures for administration (approximately 5%), and for build-





Note: Operating Expenditure excludes capital outlays, debt service, community service, and building construction. construction. neven distribution of high **Figure 3.4 Percentage of Students with F/R Lunch, and Students with LEP, by**

by this uneven distribution of high need students across districts. Federal Title 1 funding is targeted toward students from low-income families, while additional federal, state, and local dollars go toward supporting students with disabilities in special education programs. Certain state and federal funds are also dedicated to children of migrant workers and to Indian education.

Other state funds are targeted to help assure basic levels of education for lowachieving students. Under the general category of "Basic Skills" revenue, these





Table 3.1 1996-97 Per Pupil Operating Expenditures in Dollars and Percent of Total

	1							
	Region			Strata				
	Metro Area Region	Outstate Region	Mpls/St. Paul	Suburbs	Outstate: 2000+	Outstate: 2000-		
Demographics								
% Eligible for F/R Lunch	24	28	64	13	25	31		
% LEP	5	1	18	1	2	1		
% Special Education	10	11	12	10	11	11		
Mobility Rate	20	15	38	14	18	12		
Expenditure Category								
Administration	332 (5)*	345 (6)	350 (4)	327 (5)	303 (5)	385 (7)		
Support Services	208 (3)	136 (2)	267 (3)	192 (3)	133 (2)	138 (2)		
Regular Instruction	3017 (48)	2824 (49)	3505 (46)	2881 (48)	2780 (48)	2867 (50)		
Vocational Instruction	121 (2)	135 (2)	108 (1)	125 (2)	138 (2)	131 (2)		
Exceptional Instruction	993 (15)	803 (14)	1421 (18)	873 (14)	905 (15)	704 (12)		
Instructional Support	334 (5)	242 (4)	409 (5)	313 (5)	272 (5)	214 (4)		
Pupil Support	222 (3)	145 (3)	330 (4)	192 (3)	173 (3)	118 (2)		
Operations/Maintenance	530 (8)	510 (9)	632 (8)	501 (8)	505 (9)	514 (9)		
Food Service	259 (4)	252 (4)	301 (4)	247 (4)	235 (4)	269 (5)		
Pupil Transportation	345 (5)	349 (6)	471 (6)	310 (5)	345 (6)	352 (6)		
Other Operations	12 (0)	41 (1)	-65 (-1)**	34 (1)	37 (1)	44 (1)		
Total Operating	6373	5781	7730 ***	5994	5827	5737		

Note: The 1996-97 Average Per Pupil Expenditure for Minnesota was \$6,081. *Numbers in parentheses represent the percentage of each perpupil dollar spent in a category. ** A negative value represents an expenditure surplus. ***This figure is a weighted average of per pupil expenditures for Minneapolis (\$8344) and St. Paul (\$7701).

ing operations and maintenance (approximately 9%).

Table 3.1 (above) reveals differences in resource allocation between the metropolitan area and outstate Minnesota, as well as between districts in different strata. The outstate region spends a slightly greater percentage of each per pupil dollar for regular instruction, operations and maintenance, and administration (64%) than does the metro area (61%). This difference is offset by a lesser percentage of per-pupil expenditure by schools in the outstate region for support services, exceptional instruction, and instructional support (20%) as compared to spending on these categories in the metro area (23%).

The differences are greater among the four strata (Minneapolis/St. Paul, suburban, larger outstate districts (more than 2,000 students), and smaller outstate districts (2,000 students or less). Particularly evident is the fact that, compared to the other three strata, Minneapolis and St. Paul spend a smaller percentage (though not necessarily a smaller amount) on administration, regular instruction, and vocational instruction. Schools in Minneapolis and St. Paul offset this difference by spending a substantially greater percentage of per pupil dollars on exceptional instruction. The largest differences occur between Minneapolis/St. Paul schools and smaller outstate schools. Outstate schools spend 7% of each

per pupil dollar on administration, compared to only 4% spent for the same function by Minneapolis/St. Paul schools. Smaller outstate schools also spend 50% of each per pupil dollar on regular instruction, whereas Minneapolis/St. Paul schools spend 46% of each dollar for regular instruction. Additionally, the small outstate schools spend a substantially smaller portion of each dollar on exceptional instruction (12%), compared with 18% for Minneapolis and St. Paul schools. There is very little difference in the expenditure pattern of suburban districts and larger outstate districts.

TEACHER CHARACTERISTICS

When searching for the reasons



behind higher levels of student academic achievement, both researchers and the public can easily agree on one thing: quality teaching is essential to learner success. In order to provide a basis for discussion about what constitutes quality teaching, we must look at certain characteristics of the teaching force in Minnesota, and examine how our teachers compare to their peers across the nation.

LEVELS OF TEACHER EDUCATION AND EXPERIENCE

When examining the relationship between teaching and achievement, many researchers (Dolan & Schmidt, 1987; Ferguson, 1991; Wright, Horn & Sanders, 1997) agree that the quality of the teaching staff makes a difference in student achievement. What contributes to "teacher quality"? In their review of 60 studies examining the impact of various factors on student achievement, Greenwald, Hedges & Laine (1996) found that

increased levels of teacher education, teacher experience, and teacher salaries are associated with significant increases in student achievement. They also point out that there is vigorous debate about the size of the contribution made by each of these factors.

With regard to levels of teacher education, approximately 53% of the nation's teachers held a Master's degree in 1990, up from 40% in 1975 (American Federation of Teachers [AFT], 1997b). In 1996– 97, 42% of Minnesota's teaching force possessed a Master's degree or above.

Minnesota teachers had slightly more years of teaching experience (16 years) compared to the national average (15.2 years) in 1997. This gives Minnesota a rank of 18th out of the 50 states, in terms of average years of teacher experience (AFT, 1997b).

Table 3.2 shows differences in the educational attainment and years of experience among the population of fulltime teaching professionals serving Minnesota public schools during the 1996-97 school year. A greater proportion of male teachers (46%) held at least a Master's degree than their female counterparts (40%). The metropolitan area had a greater proportion of teachers with advanced degrees (52% of the teaching force) compared to outstate Minnesota (33%). This may be due to several factors, among them the easier access to advanced degree programs in the Twin Cities area.

The average length of teaching experience varies in interesting ways among Minnesota teachers. Male teachers in Minnesota have taught an average of three years longer than have female instructors (18 years vs. 15 years). Teachers in outstate Minnesota,

Tahla 3 2	1996-97 Minnesota	Teachers	Profile: Full-	time Tea	nchers (10)0% FTE)
Idvie J.Z	1770*7/ WIIIIICSULA	<i>icaciicis</i>	, , ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,			

	N	% of Teachers Holding BA or Higher	% of Teachers Holding MA or Higher	Average Years of Teaching Experience	Average Teacher Salary	Average Teacher Age	Number of Teachers Aged 55 or Over	Number of Teachers Aged 60 or Over
Total	44874	100	42	16	38232 *	44	6952	1817
Females Males	29671 14555	100 100	40 46	15 18	37905 39367	44 45	4078 2874	1237 580
Elementary	22758	100	41	16	38649	44	3422	1016
Secondary	20086	100	43	16	37989	44	3357	742
Metro Area	21518	100	52	15	41008	44	3534	923
Outstate	22158	100	33	17	35707	44	3302	853
Mpls/St. Paul	5081	100	52	15	43098	45	923	322
TC Suburbs	16437	100	52	16	40363	44	2611	601
Outstate: 2000+	10643	100	45	16	37916	44	1767	444
Outstate: 2000-	11515	100	22	17	33664	44	1535	409

* This figure varies slightly from the AFT average given on p. 16.



on average, have taught two more years Cost of living and the number of than metropolitan teachers. More specifically, teachers from districts having less than 2000 students edge out their counterparts from all other strata, with an average of 17 years of

The age distribution of Minnesota teachers has shown a dramatic change in the last ten years. In 1987-88, a plurality of teachers fell in the 35-44 age bracket. Today, the largest group of teachers is between 45 and 54 years old.

teaching experience.

TEACHER SALARIES

The nation's average teacher salary for 1996-1997 was \$38,436, representing an increase of 2.2% from the previous year. For the same period, the Minnesota average teacher salary was \$38,276, giving Minnesota a rank of 18th nationally. However, states vary considerably in their economic condition and cost of living. When cost-of-living differences are accounted for, Minnesota's adjusted average teacher salary rose to \$39,598, and our adjusted rank rose to 13th out of the 50 states (AFT, 1997Ь).

Table 3.2 (p. 15) shows that male teachers in Minnesota have a higher average salary than female teachers, possibly because male teachers may be more likely to hold an advanced degree, and because male teachers overall have a greater average number of years of experience. Elementary and secondary teachers' salaries are vir-tually equivalent, but there are considerable differences between metropolitan and outstate teachers.

Analyzing the data by region, we find that teachers in the Twin Cities average almost five thousand dollars more in annual salary than teachers outstate (\$41,008 vs. \$35,707).

teachers holding advanced degrees may account for these differences.

Age of Teaching Force

The age distribution of Minnesota teachers has shown a dramatic change in the last ten years. In 1987-88, a plurality of teachers fell in the 35-44 age bracket (40%). Today, the largest group of teachers (38%) is between 45 and 54 years old (Minnesota Department of Children, Families & Learning, 1997). Table 3.2 reveals that almost 7,000 of Minnesota's teachers are over 55 years old, with slightly over 1,800 of those at or above age 60. The number of teachers at or over 55 years of age is approximately 16% of the total professional teaching force in Minnesota. A larger percentage of these teachers are female, mirroring the larger proportion of females in the field in general.

While fairly evenly distributed between elementary and secondary settings, the greatest number of teachers in the upper age groups are found in the suburban districts surrounding Minneapolis and St. Paul.

STUDENT CHARACTERISTICS

Commonly, reports of this type include a lengthy section on student characteristics. We have placed extensive student characteristics data elsewhere (Chapter 5: Achievement) to provide context for the achievement information. That chapter contains data on rates of poverty, limited English proficiency, mobility, and disability (special education placement) for students taking the achievement tests. In this section, we merely underscore what many others have said about the changing nature of American and Minnesota schools.

From 1987 through 1994 the percentage of Black, Asian, and Hispanic students in Minnesota almost doubled (Minnesota Department of Children,

Families & Learning, 1997). The percentage of ethnic minority students in Minnesota increased from 6.1% in 1987 to 11.1% in 1994.

The changing composition of students is clearly evident when one examines the percentages of various ethnic minority groups at each grade (see Table 3.3). The best indicators of the growing minority population are grades kindergarten through eighth grade, grades in which dropping out seldom occurs. During these grades, changes in enrollment are principally affected by birth rates, immigration, and migration.

Shifting demographics may have an impact on state average test scores and graduation rates in the years to come. Minority students are characterized by higher rates of poverty (indicated by the percentage of students eligible for free or reducedprice lunch), limited English proficiency, and disability (indicated by placement in special education), as seen in tables 5.7 - 5.13 (pp. 39-42). If increasingly greater numbers of Minnesota's minority students are faced with these challenges, higher rates of achievement and graduation will require new strategies for in-

Minority students are characterized by higher rates of poverty, limited English proficiency, and disability. If increasingly greater numbers of Minnesota's minority students are faced with these challenges, higher rates of achievement and graduation will require new strategies for instruction and curriculum.

struction and curriculum.

Finding new strategies is an issue throughout the state, not just for the core cities, although the core cities carry the heaviest responsibility. Figure



_	Grades Largely Unaffected by Dropout							
Grade	Boys	Girls	Amer Ind	Asian	Black	Hispanic	White	Total
К	32126 (52%)	30000 (48%)	1361 (2%)	2940 (5%)	4593 (7%)	2093 (3%)	51139 (82%)	62126
1	32320 (52%)	30019 (48%)	1456 (2%)	2955 (5%)	4443 (7%)	2022 (3%)	51463 (83%)	62339
2	33057 (52%)	31087 (48%)	1412 (2%)	3154 (5%)	4398 (7%)	2003 (3%)	53243 (83%)	64131
3	32391 (51%)	31237 (49%)	1388 (2%)	2986 (5%)	4158 (6%)	1753 (3%)	53343 (84%)	63628
4	32731 (52%)	30763 (48%)	1362 (2%)	3137 (5%)	3972 (6%)	1690 (3%)	53333 (84%)	63494
5	32747 (52%)	30689 (48%)	1340 (2%)	2913 (5%)	3681 (6%)	1538 (2%)	53964 (85%)	63436
6	33737 (51%)	31903 (49%)	1406 (2%)	3076 (5%)	3460 (5%)	1552 (2%)	56146 (86%)	65640
7	34675 (51%)	32791 (49%)	1374 (2%)	2903 (4%)	3506 (5%)	1499 (2%)	58184 (86%)	67466
8	34095 (51%)	32431 (49%)	1389 (2%)	2957 (4%)	3172 (5%)	1521 (2%)	57487 (86%)	66526
				<u> </u>				

	Grades	Allecte	u by .	Diopo	uu	
_						-

Grade	Boys	Girls	Amer. Ind.	Asian	Black	Hispanic	White	Total
9	35491 (52%)	33248 (48%)	1389 (2%)	2965 (4%)	3546 (5%)	1507 (2%)	59332 (86%)	68739
10	35099 (51%)	33304 (49%)	1235 (2%)	2809 (4%)	3107 (4%)	1337 (2%)	59915 (88%)	68403
11	33065 (51%)	31663 (49%)	1045 (2%)	2456 (4%)	2506 (4%)	1074 (2%)	57647 (89%)	64728
12	32160 (51%)	31200 (49%)	900 (1%)	2218 (4%)	2387 (4%)	1100 (2%)	56755 (90%)	63360

3.5 shows the percentage of students in Figure 3.5 Percentage of Students in Each Ethnic Group, by Minnesota Region each region of the state by ethnic group. A large majority of black students, 68%, reside in the core cities. Fully one-fourth, however, attend suburban schools. A bare majority of Asian students, 52%, attend in the core cities, but one-third attend suburban schools, and 15% attend schools in outstate Minnesota. Most Hispanics (70%) attend outside the core cities, although a majority (57%) attend in the metropolitan area. Most American Indian students live outside the metropolitan area, with fully 40% attending in smaller districts outstate.

SCHOOL CHARACTERISTICS

Do some schools produce a higher level of student achievement than other schools that serve similar students? This question has intrigued researchers,







policy makers, and practitioners for decades. It served as the driving force behind the initial school effects research in the 1960s. Pioneers in the effective schools research outlined several correlates that were associated with "effective" schools. These early findings were quickly embraced by American educators and soon found their way into school improvement programs across the United States.

By the mid-1980s, however, there was mounting evidence that much of the effective schools research suffered from methodological limitations. First, findings cannot be reduced to a formula for school improvement. Second, findings cannot always be generalized to all schools.

Nonetheless, school-based factors do have a major impact on student learning. Wang, Haertel, and Walberg (1993) summarized the results of their meta-analysis by rating the importance of 28 different factors that influenced student learning. The most influential factors, based on their mean effect. were classroom management, defined as increased student engagement, decreased disruptive behaviors, and quality of instructional time. Other influential factors included studentteacher interaction, quantity of instruction, school culture, classroom climate, curriculum design, and classroom assessment.

Data from other studies also confirm

that school factors affect student achievement. Teddie and Stringfield (1993) conducted a study that found school, classroom, and teacher factors to account for 25 percent of the differences in student achievement. These findings are further supported by Zigarelli's (1996) empirical study, which tested the effect of each of the most commonly identified school effectiveness variables on student achievement. His analysis indicated that an achievement-oriented school culture, the principal's autonomy in hiring and firing teachers, and high teacher morale were the most important school characteristics associated with student achievement. These findings indicate that school factors

Level ¹	Enrollment: Pre-K-12	Teachers (FTE) ²	Pupil/Teacher Ratio	
National Average	44,840,481	2,598,220	17.3	
State Average				
Illinois	1,943,623	113,538	17.1	
Indiana	977,263	55,821	17.5	
Iowa	502,343	32,318	15.5	
Kansas	463,008	30,729	15.1	
M ich igan	1,641,456	83,179	19.7	
M innesota	835,166	46,971	17.8	
M issouri	889,881	57,951	15.4	
Nebraska	289,744	20,028	14.5	
N. Dakota	119,100	7,501	15.9	
Ohio	1,836,015	107,347	17.1	
S. Dakota	144,685	9,641	15	
Wisconsin	870,175	55,033	15.8	
Connecticut	517,935	36,070	14.4	
Maine	213,569	15,392	13.9	

Table 3.4 1995-96 National Pupil-Teacher R	<i>latios</i>
--	---------------

¹ Snyder, T.D. (1997). Digest of educational statistics (NCES 98-015). Washington, DC: U.S. Dept. of Education, National Center for Education Statistics.

² FTE refers to "Full-time Equivalent."



can make a difference in student academic achievement.

Salient questions still need to be answered. In Minnesota, one of the most often discussed school-based factors is the ratio of teachers to students. Specifically, where should we focus our energies, and to what degree do these factors influence student achievement? To begin to answer these questions, we must examine our current understanding of class size and its effect on achievement, and then compare class size in Minnesota schools to class sizes in other states.

CLASS SIZE

The association between reduced class size and higher academic achievement is not as clear as the discussion sometimes suggests. For example, Glass and Smith's (1978) meta-analysis of research on the relationship of class size and student achievement found that while reduced class size can be expected to produce increased academic achievement, the major benefits from reduced class size are obtained *as class*

size is reduced below 20 pupils. These findings, however, have been strongly criticized because of the seemingly arbitrary way in which studies were selected for analysis.

A compilation of studies examined by Robinson (1990) added an important set of qualifiers to the conclusions of Glass and Smith. Robinson stated that research does not support the expectation that smaller classes alone will result in greater academic gains for students. Rather, the effect of class size on student learning varies by: grade level (small classes are more effective in the primary grades); pupil characteristics, (for example, whether students are economically disadvantaged, from some ethnic minority, etc.); subject area (specifically, reading and mathematics in the primary grades); teaching methods; and other learning interventions. Unfortunately, this wide-ranging review failed to distinguish the best-designed studies from those using poor methodology, and therefore, the findings must be viewed as at least somewhat tentative.

The discussion surrounding class size and student achievement continues to provoke a great deal of interest. One theme that consistently presents itself is that lower class size alone seems unlikely to improve student achievement unless it is accompanied by improved classroom practices (Mueller, in press). A study conducted by Murnane and Levy (1996) highlights this point. In their study, not all schools that lowered class size succeeded in raising achievement. Those that succeeded viewed smaller class size as an enabling factor that allowed them to make major curricular and instructional changes. Combinations of changes, especially curriculum and instructional changes, are what teachers and administrators should focus on if they wish to max-imize the opportunities that smaller class sizes offer. Improved learning arises from the curriculum and instructional improvements that smaller class size makes possible, not from smaller classes per se.

The data represented in Tables 3.4 (page 18) and 3.5 (at left) are based on pupil/teacher ratios. Pupil/ teacher ratio is not the same as class size, however. Class size is based on the actual number of students a teacher has in his/her class or classes. Pupil/teacher ratio is based on the total number of licensed teaching staff in a single school compared to the total number of pupils-whether all of these licensed staff are teaching classes, or detailed for other teachingrelated tasks, such as curriculum development. Pupil/teacher ratios can often obscure the actual workload faced by a teacher. Class size-a more direct measure of classroom organization-is more important to academic achievement than pupil/teacher ratio (Boozer & Rouse, 1995). We are using pupil/teacher ratios in this report because class size information is lacking.

 Table 3.5
 1996-97 Minnesota Pupil-Teacher Ratios

Level	Enrollment K-12	Teachers (FTE)	Pupil/Teacher Ratio
State	827,490	48,168	17.2
Strata			
Mpls/St. Paul	92,723	5,463	17.0
Suburban	326,313	18,081	18.0
Outstate: 2000+	201,555	11,668	17.3
Outstate: 2000-	206,860	12,881	16.1
Metro Area vs. Outstate			
Metro Area	419,036	23,544	17.8
Outstate	408,415	24,548	16.6



The national pupil/teacher ratios in Table 3.4 (p. 18) are reported according to regions defined by the National Assessment of Educational Progress (NAEP). These regions include states that are clustered by a set of shared characteristics—region of the country, gender, race/ethnicity, parental education, type of school, and participation in federally funded Title I programs and the free/ reduced-price lunch component of the National School Lunch Program. Maine and Connecticut, two states that have student achievement levels similar to Minnesota's, are also included.

Table 3.6 shows pupil/staff ratios. "Staff" refers to licensed school personnel, including administrators, counselors, teachers, media specialists, speech clinicians, psychologists, etc. Student/staff ratios are not clear indicators of student achievement because they cannot isolate single variables in the highly complex process of learning development. They do, however, tend to illuminate resource allocation decisions made by school districts (i.e., how much of each dollar goes for building improvement, staffing, and curriculum and instruction).

Level	Enrollment K–12	Licensed Staff (FTE)	Pupil/Staff Ratio
State	827,490	55,092	15.0
Strata			_
Mpls/St. Paul	92,723	6,415	14.5
Suburban	326,313	20,695	15.8
Outstate: 2000+	201,555	13,303	15.2
Outstate: 2000-	206,860	14,679	14.1
Metro Area vs. Outstate			
Metro Area	419,036	27,110	15.5
Outstate	408,415	27,982	14.6

Table 3.6 1996-97 Minnesota Pupil-Staff Ratios





CHAPTER 4 PARTICIPATION: COURSEWORK, ATTENDANCE, AND GRADUATION

In this chapter, we have examined participation outcomes and indicators of student participation in education generally, or in particular educational programs. Are students in Minnesota high schools taking challenging coursework? What are the attendance patterns in Minnesota schools? What are the patterns that appear in Minnesota's graduation and dropout rates? This chapter addresses these questions, both for Minnesota students in general and for students in selected subgroups.

Secondary School Coursework

William Bennett (1988) states, in response to A Nation at Risk, "Common sense tells us, and education research confirms, that youngsters rarely learn what they do not study" (p. 23). Curriculum is important when addressing any educational enterprise. Patterns of high school coursework are interesting in their own right, but also for their relationship to other outcomes, such as later choice of college major (Jones, Bekhuis, & Davenport, 1985) and higher scores on college admissions tests (American College Testing Program, 1992, 1997). What do other states require of their students? What do our students study, and what variation is there across regions of the state and segments of our high school population? To partially address these questions, we consulted several sources, most of which focused on college preparatory courses.

MINNESOTA'S PROFILE OF LEARNING PERFORMANCE PACKAGES: A SNAPSHOT FROM PELICAN RAPIDS, MINNESOTA

LEARNING AREA: ARTS AND LITERATURE

Crystal Thorson, an elementary art teacher from Viking Elementary School in Pelican Rapids, transforms her classroom into an "Art Museum." The art museum is based on 30 – 35 objects representing a variety of time periods, cultures, media, and artists. While in the art museum, students look for works that best match the art vocabulary studied in class. Students demonstrate their understanding by logging their findings on a scavenger hunt checklist. Students must find straight lines, geometric shapes, warm colors, and rough textures in the artwork. Once they have identified the elements, students attempt to draw them. Parent volunteers are on hand to assist.

When students have completed this task they select a favorite art object and do a more in-depth analysis. Students gather information about their favorite artwork using an Art Detective Notes record sheet. The following figure represents one student's information gathering notes.

(cont'd on p. 23)

STATE COURSE REQUIREMENTS AND NATIONAL RECOMMENDATIONS

In their August 1998 report, the Council of Chief State School Officers (CCSSO, 1998) gave a comprehensive summary of course requirements necessary for high school graduation in all states and territories of the United States. Minnesota was listed as a "standards-based system," and had no statewide course requirements listed in any content area, although schools and districts do have requirements that vary by district. In place of course requirements at the state level, Minnesota has specified basic and high standards in its Graduation Rule, standards that every student is expected to meet.

Minnesota is one of a handful of states with no high school course graduation requirements at the state level and which leaves course requirements to districts. Most states have requirements in English (with most states requiring four years), mathematics (usually two or more years), social studies (usually three or more years), and science (usually 2 or more years). Some states have additional graduation requirements in the arts, foreign language, and computer technology. Table 4.1 (p. 22) shows the recommended course credit requirements in core academic areas recommended in A Nation at Risk (NCEE, 1983) and by the ACT Assessment program.

In contrast to most other states, Minnesota has expressed its statewide requirements, not in terms of courses taken, but in terms of what students should know and be able to do—the Graduation Standards. The *Basic*



Table 4.1 Recommended Course Credit Requirements

	Nation at Risk	ACT
Mathematics	3	31
Science	3	3
English	4	4
Social Studies	3 ²	
Social Sciences		3 ²
Foreign Language	2 ³	0
Computers	1/2	O ⁴

¹ ACT makes more specific suggestions concerning which math courses to take.

² ACT suggests three credits in social science, which includes social studies. A Nation at Risk just recommends social studies.

³A Nation at Risk recommends foreign languages only for college-bound students.

⁴ ACT places computer courses with mathematics.

National Commission on Excellence in Education (1983). A nation at risk: The imperative for educational reform. Washington, DC: Superintendant of Documents, U.S. Government Printing Office.

American College Testing Program (1997). ACT high school profile report high school graduating class 1997: State composite for Minnesota. Code 240-000, Iowa City, IA: Author.

Standards Tests and the performance packages associated with the Preparatory and High Standards give students the opportunity to demonstrate attainment of those standards. ACT CORE COURSE PREPARATION As shown in Table 4.1 (above), the ACT Assessment Program, the major college admissions testing program in Minnesota, asks students to report on completion of core academic courses as they enroll to take the college entrance test. ACT has found that taking the recommended ACT core sequence is associated with higher scores on the admissions test. The recommended core sequence includes four years of English and three years each of science, social science, and mathematics. Figure 4.1(below) shows the proportion of Minnesota test-takers completing the core over an elevenyear time span (ACT Assessment Program, 1992, 1997, 1998). From 1987-88 to 1997-98, the proportion of college-bound ACT test-takers completing a basic academic core has risen from 55% to 73%. This rise in preparation is extremely encouraging, but the distribution of that preparation over Minnesota's ethnic groups is not.

Figure 4.2 (p. 23) shows the percentage of 1997-98 Minnesota testtakers in each ethnic group who report having the core (ACT Assessment Program, 1998). Among presumably college bound students, there are marked differences across ethnic groups in the percentages of students

Figure 4.1 Percentage of Students Having Completed the ACT Recommended Core Academic Preparation for the Years 1987–98





Figure 4.2 Percentage of Minnesota ACT Test-takers Having the Recommended Core Academic Preparation, by Ethnic Group (1997-98)



Ethnic Group

completing a basic academic core. Not only do these data raise serious questions about the equality of preparation for college across ethnic groups, they also raise doubts as to whether the preparation of some students is consistent with their future educational plans.

COLLEGE-LEVEL COURSEWORK FOR HIGH SCHOOL STUDENTS

For Minnesota high school students, there is an array of three options within which a student can take advanced coursework for high school credit, and potentially, also for college credit: the Advanced Placement (AP) program, the International Baccalaureate (IB) program, and the Postsecondary Enrollment Options (PSEO) program. Advanced Placement and International Baccalaureate courses are taken at the high school; PSEO courses can be taken either at the high school or on the campus of a postsecondary institution.

Table 4.2 (p. 24) shows the growth of participation in AP programs. Since 1993, the number of participating schools has increased from 165 to 205. The number of students taking the exams has more than doubled from 4.438 to 9.369. The number of exams qualifying for a score of three or higher has increased over the period since 1991. The percentage of exams scored 3 or higher (qualifying for scholarship recognition and college credit on a scale of 1-5) hit a peak (66%) in 1994, dropped to 58% in 1995 and rose to 61% in 1997 (Minnesota Department of Children, Families, & Learning, 1998d). Undoubtedly, increased participation in AP programs has had some impact on overall test performance.

The number of schools and students participating in the International Baccalaureate program is smaller than for the AP program, just nine schools and 815 students in 1997. It too, however, is growing, and the percentage of exams achieving a score of 4 or better (qualifying for Honors status on a scale of 1–7) increased from 65% in 1996 to 71% in 1997.




Table 4.2 Growth in the Advanced Placement Program

In the case of both AP and IB programs, Minnesota post-secondary institutions award college credit to students who pass the various exams at certain levels (generally at a score of "3" for AP exams and "4" for IB tests). State funds also provide for scholarships to students who pass at these levels, for use in any Minnesota post-secondary institution.

Participation in the PSEO program has generally increased over time (see Figure 4.3, at right). From 1992 to 1995, the number of students taking PSEO courses on post-secondary campuses steadily increased from 5,457 to 6,671. After a slight decrease during the 1995-96 school year, the numbers again increased during the 1996-97 school year to 6,552 students. The decline in 1995-96 reflects the fact that students are increasingly taking PSEO courses at their own high school, rather than on post-secondary campuses. At present, more than half of all PSEO courses are taken at high schools as part of a

(cont'd from p. 23)

ARTS AND LITERATURE Task 2

The second task requires communicating ideas using visual arts. Crystal facilitates this process by introducing Faith Ringgold's "Narrative Story Quilt" artworks. After learning about the Story Quilts, students develop an idea for a story, using their favorite artwork as a springboard. Once the story has been formulated, students begin preliminary drawings which tell their story without words.

Before students complete their Story Quilt, they are taught a variety of art processes (printmaking, design & layout, mixed media techniques, etc.) to incorporate into their final Story Quilt.

The Story Quilts measure at least 18 by 18 inches. Each quilt has a border made of relief prints which include elements from their favorite artwork. Students create the design, choose the colors, and determine what other media (foil paper, origami paper, wallpaper, craft paper, magazine paper, etc.) that they will use to depict their story.

	1993	1994	1995	1996	1997
Schools	165	167	193	201	205
Candidates	4,438	4,890	7,278	8,465	9,369
Exams	5,794	6,491	9,401	11,169	12,641
Exams/Candidates	1.306	1.327	1.292	1.319	1.349

Minnesota Department of Children, Families, and Learning (1998). Advanced Placement and International Baccalaureate Programs. St. Paul: Author.

Figure 4.3 Number of Students Participating in Post Secondary Enrollment Options Programs on Post-secondary Campuses for the years 1992–97



The students are evaluated on the performance tasks using a feedback checklist. The scores consist of Y (yes, the student has met the performance requirement) or N (no, the student has not met the performance requirement). Crystal uses the checklists to monitor student progress on the standard. Once students complete all the performance tasks, Crystal will assign the students a score for the standard taking into account their previous performance. Below is a **partial** feedback checklist for Task 1. To help parents understand the new graduation requirements, Crystal set up an exhibit of student Story Quilts. The exhibit was used to guide parents through the tasks in the performance package. Crystal sees the graduation standards making the classroom more student centered, with a much stronger emphasis on individual student achievement. Although the paperwork associated with the performances has been burdensome, Crystal says she now knows her students better.

Partial Feedback Checklist: Task 1

Y = YES, MET PERFORMANCE STD. N= NO, DID NOT MEET PERFORMANCE STD.

Makes logic	al obser	vation	s ahou	t the :	artict's	time a	nd nla	ce froi	n the	Ň	Teach	er Evi	duation
artwork Identifies to	ols and	mater	ials tha	nt wou	ld logi	cally h	have be	en use	d by		- ØŢ		<u></u>
the artist Describes fe	elings n	elated	to the	artwo	rk dea	rly	×.	2	1 ⁴ 6 1	, R	43 <u>.5</u>	<u>*,4</u> 900 •	
Uses art terr	ns corre	ectly to	answe	er que	stions	с ^с	æ .	ś	1	2	· <u> </u>		
Identifies sti shape, color,	rong exa ; texture	ample: ?)	s of eac	ch art i	elemer	nt in ch	nosen a	rtwori	k (line ,	· •.			14.e

BEST COPY AVAILABLE

		Gra	ide	
	Third	Fifth	Eighth	Twelfth
Total	95	95	94	92
Girls	95	95	93	92
Boys	95	95	94	93
Asian	97	97	94	92
Black	94	94	90	88
Hispanic	93	93	89	87
American Indian	92	92	86	84
White	95	95	94	93
Special Education	95	94	91	90
Limited English	96	96	92	91
Metro Area	94	94	93	92
Outstate	96	96	95	93
Mpls./St. Paul	94	94	90	89
TC Suburbs	94	95	93	92
Outstate 2000 ·	96	96	94	92
Outstate 2000 '	96	96	95	93
Public/Not Charter	95	95	94	92
Public/Charter	94	94	91	79

 Table 4.3 Percentage of Students in Attendance for Third, Fifth, Eighth, and

 Twelfth Grades

Data Sources: For third, fifth, and eighth grades, attendance was calculated using all students in 1998. Twelfth grade attendance was calculated from the 1997 completion study file of the Minnesota Department of Children, Families, and Learning. 25% nationally (Shaughnessy, Nelson, & Norris, 1998). Only four states (Utah, Maryland, Massachusetts, and Delaware) enrolled a higher percentage of eighth graders in algebra. Hence, Minnesota's enrollment compares favorably to that in most other states.

Finally, the participation of high school students (grades 9 - 12) in advanced high school mathematics (defined as geometry, algebra 2, trigonometry, or calculus) and science courses (defined as chemistry, physics, or advanced or second-year science

Increasingly, high school students can elect to take college level work in Advanced Placement, International Baccalaureate, and Post-secondary Options programs, and they have been choosing to do so in growing numbers.

courses) is also monitored at the national level. In the latest year for which we found data, 1996 (Editorial Projects in Education, Inc., 1998), 45% of Minnesota high school students were reported as taking advanced math (Figure 4.4). Fifteen states reported more students doing so (including Illinois, Massachusetts, Nebraska, North Carolina, North

cooperative effort between the high school and a post-secondary institution. Social studies and language arts courses are most popular. Selections also include vocational/technical, math, science, foreign language, business, and health courses. Increasingly, high school students can elect to take college level work in Advanced Placement, International Baccalaureate, and PSEO programs, and they have been choosing to do so in growing numbers.

Another coursework indicator monitored on the national level is the percentage of eighth graders taking algebra (see Figure 4.4). In 1996, 34% of Minnesota eighth graders enrolled in algebra, as compared to

Figure 4.4 Percentage of Eighth Grade Students Taking Algebra and High School Students Taking Advanced Mathematics and Science





Dakota, Pennsylvania, South Dakota, Utah, West Virginia, and Wisconsin, all of whom reported more than 50% of their students taking upper-level mathematics). Sixteen states reported more students in advanced science (including Connecticut, Illinois, Kentucky, Massachusetts, Mississippi, Nebraska, North Dakota, Pennsylvania, Utah, West Virginia, and Wisconsin, all of whom report more than 30% as taking advanced science). In Minnesota, 28% were reported as doing so (see Figure 4.4, p. 25).

ATTENDANCE

One of the strongest foundations for school success is regular school attendance. During the 1997 fiscal year, the Minnesota Office of the Legislative Auditor performed an analysis of the *Basic Standards Test* scores in reading and mathematics (1998). Of the variables analyzed, attendance had the strongest relationship with average school test scores.

Table 4.3 (p. 25) shows the percentage of students in attendance at each of four grade levels. At grades three and five, the Minnesota average is 95%. At grade eight, it is 94%, and decreases to 92% by twelfth grade. Girls and boys have identical attendance rates at third and fifth grade, and there is only a one-percent difference at the eighth and twelfth grades.

Differences among ethnic groups are small in third and fifth grades but become more marked in the secondary grades. Asians have the highest attendance rates of any ethnic group in third and fifth grades (97%); in eighth and twelfth grades, they have the same attendance rate as the state overall (94% for eighth grade; 92% for twelfth grade). From 95% in third grade to 93% in twelfth grade, white students' attendance rates change very little as students get older. Rates for black students drop six percentage points between third and twelfth grades, from 94% to 88%. Attendance rates for Hispanic students also fall 6% between third and twelfth grades, from 93% to 87%. American Indian students begin from the lowest attendance rate, 92% in third grade, and suffer the largest decline, 8%, by twelfth grade.

Students receiving special education and students with limited English proficiency have rates within 1% of the state as a whole at the third and fifth grade levels. These special populations experience a decline in attendance from the fifth to the eighth grades. By twelfth grade, their attendance rates are 90% for Special Education students and 91% for limited English proficiency students, 1 – 2% below those of the state as a whole.

When categorizing the state's

districts into four groups—Minneapolis/St. Paul, the Twin Cities suburbs, large outstate districts (more than 2000 students), and small outstate districts (less than 2000 students), attendance rate declines are sharpest in the two core cities. Across these four regions, rates differ by at most 2% in third grade. By twelfth grade, however, there is a 4% difference between the Twin Cities and the small outstate districts.

Attendance rates for charter schools were also calculated. In third and fifth grades, the attendance rates for charter schools were one percent below that of other public schools (94% versus 95%). The charter school attendance rate declines 3% between fifth and eighth grades to 91%. The twelfth grade attendance rate is only 79%, although it is based on a very small number of twelfth grade charter school

Category	Number of Students	Number of Graduates	Number of Dropouts	Number Continuing	4-year Graduation Rate (%)	Dropout Rate (%)
Total	59699	46680	6758	6261	78	11
Girls	29298	23870	2801	2627	81	10
Boys	30401	22810	3957	3634	75	13
Asian	1784	1216	317	251	68	18
Black	2506	891	998	617	36	40
Hispanic	99 3	434	373	186	44	38
American Indian	1089	447	413	229	41	38
White	53327	43692	4657	4978	82	9
LEP	571	285	193	93	50	34
Special Ed.	5830	3403	1180	1247	58	20
Metro Area	26581	20138	3769	2674	76	14
Outstate	30583	26109	2297	2177	85	7
Mpls./St. Paul	5759	2704	2050	1005	47	36
TC Suburbs	20822	17434	1719	1669	84	8
Outstate:2000+	15215	12083	1516	1616	79	10
Outstate:2000-	15368	14026	781	561	91	5
Public/Charter	149	37	54	58	25	36
Public/Non- Charter	59550	46643	6704	6203	78	11

Table 4.4 Four Year Graduation and Dropout Rates



students in the 1997 high school completion study (Minnesota Department of Children, Families, and Learning, 1998b). Educational researchers have long studied the association between attendance and leaving high school ("dropping out"). Poor attendance often precedes dropping out. Indeed, the decline in attendance from fifth to twelfth grade, as shown in Table 4.3 (p. 25), may begin to reflect the

PROFILE OF LEARNING PERFORMANCE PACKAGE: A SNAPSHOT FROM CANNON FALLS, MINNESOTA

LEARNING AREA: DECISION MAKING

Mr. Lindow and his fellow teachers in Cannon Falls have teamed up to deliver the decision-making standard in 6th, 7th, and 8th grade using a modified version of the performance package, "Understanding Career Options."

In sixth grade, students take an interest and ability survey. Teachers guide students through the use of the selfassessment survey and the interpretation of the results. Teachers discuss the findings with their students, helping them to see connections among interest, ability, and career choices.

In seventh grade, students take part in another interest survey. Using the results of this survey and last year's survey, students select two possible career options to pursue in more depth. Seventh grade teachers build in class time that allows students to gather information on one of their career choices. Students typically use the media center to begin their information search. Books, periodicals, the Internet, and CD-ROMs make up a portion of the resources that students use in the development of their career file.

The next stage involves a reallife experience. Students **Techn** spend a half-day working with (shadowing) a person in their ogy th preferred profession. According to Mr. Lindow, "This reallife context is a critical element in understanding what the career is all about. It also serves as a reference in their career portfolio."

After completing the shadowing experience, students put together a presentation for the class. The presentation will include current, accurate information about the student's career option. It will identify the specific skills and abilities needed to do the job, and it should contain information regarding the preparation for the career (e.g. education and training). Students may present this information to the class in a variety of formats: speeches, workshops, brochures, power point presentations, videotaped productions, etc.

One student created a poster that outlined the specific skills and abilities of a retail sales clerk in her local community. The following are excerpts from her report:

RETAIL SALES CLERK

Skills and Abilities:

A retail sales person has several responsibilities. First, they need to be knowledgeable about the cash register and the cost of items in the store. Second, they must work hard to keep the store clean and orderly. Third, they need to be upbeat and friendly because over half the job is working with customers.

Education and Training:

Most employers will hire high school graduates, but the more education and training a person has the better his/her chances of getting a job.

Helpful subjects to take in school are math and English. A cashier's training course and/or computer skills would also be helpful.

Retail sales people should be in good health. Sales people are often on their feet all day and tend to do a lot of lifting.

Technology Used in the Business:

A cash register and calculator are the main items of technology that this retail business uses. A filing system keeps track of inventory.

I think a computer would be a good addition to the store. It would help keep things more organized.

Mr. Lindow stated that the students have responded positively to the performance package and the shadowing experience. And, although the middle level teachers at Cannon Falls have been doing a careers unit for some time, Mr. Lindow said the standard and performance package have given them a stronger focus and a common language.

The final portion of the Understanding Career Options performance package is completed in eighth grade. Students refer back to their career file and explore their other career option. This exploration is mainly conducted on the students' own time. Their findings may again be presented in a variety of formats: written, oral, audio, video, workshop, etc. An interesting question that students must address in their essay is the impact their career might

have on personal, family, and community life.

> The score a student receives for the Decision Making standard is determined by taking into account the level of achievement attained by the student on all the performance tasks described in the package. The overall scoring criteria for the standard consists of a four-point scale with the following values:

4 – *Exemplary:* Indicates evidence of student learning in all parts of the standard at a level that exceeds expectations by using and applying knowledge consistently, in new and insightful ways.

3 – *Proficient:* Indicates evidence of student learning in all parts of the standard at a consistently proficient level.

" 2 – *Novice:* Indicates evidence of student learning in all parts of the standard at an adequate level some or all of the time.

1 – *Emerging:* Indicates evidence of student learning in all parts of the standard at a superficial level some or all of the time.



declining attendance of some students that precedes dropping out. Stemming the decline in attendance during the secondary grades may be essential to cutting the dropout rate covered in the next section.

DROPOUT RATES

Table 4.4 (p. 26) shows the four-year graduation rate and the dropout rate for the state as a whole and for various groups and regions around the state. These estimates come from a four-year longitudinal study of students who were ninth graders in 1994, excluding students who transferred out of state or to a non-public school. For the state as a whole, 78% of the 1994 ninth graders in the study graduated from a Minnesota high school four years later. Eleven percent left high school. Somewhat surprisingly to us, fully 11% were still enrolled in high school but had not yet completed work for their diploma. This four-year high school completion rate, 78%, may not be comparable to that from other states, who include all students receiving a high school diploma, not just those finishing in four years. Census studies include, not only those graduating in more than four years, but also those who receive a high school equivalency degree.

Boys have a lower graduation rate

(75% vs. 81%) and a higher dropout rate (13% vs. 10%) than girls. Among the ethnic groups, Whites have the highest graduation rate (82%), followed by Asian students (68%), Hispanic (44%), American Indian (41%), and Black students (36%). Completion rates vary widely across the different regions of the state, from 47% in the two urban cities to a commendable 91% among the small outstate districts.

For the state as a whole, 78% of the 1994 ninth graders in the study graduated from a Minnesota high school four years later. Eleven percent left high school, and another 11% were still enrolled in high school but had not yet completed work for their diploma.

Dropout and graduation rates among blacks pose a paradox in light of educational attainment figures reported by the Census Bureau. The low completion rate and high dropout rate in Table 4.4 seemingly run contrary to national reports, which suggest that the high school graduation rates for black and white students are equalizing (News America Digital Publishing/Fox News Online, 1998). The Census Bureau counts an indi-

vidual as having a high school education if they have a high school diploma or an equivalency degree, and its surveys cover adults 25 years and older. In the last full census year, 1990, the Census Bureau reported that 76% of blacks and 82% of all Minnesotans aged 25 and over had attained a high school education, either a high school diploma or an equivalency degree (Snyder, 1997). More recently, in their 1997 educational attainment study, the Census Bureau (1998) estimated that the percentage of blacks with a high school education was as high as that for the state as a whole.

In this regard, fully 24% of black students have neither graduated nor dropped out, but are continuing their education after the end of the four years covered by the study (see Table 4.4). Because the data in Table 4.4 are limited to the four years following ninth grade and to the standard K-12 educational programs, the completion rates and dropout rates shown in the table may incompletely describe the high school educational attainment of black Minnesotans. The Census data suggest that many black Minnesotans may attain a high school education after the traditional four years and/or outside a conventional K-12 degree program.

BESTCOPY AVAILABLE





CHAPTER 5 ACHIEVEMENT

INTRODUCTION

Increasingly, the proof of success in Minnesota's K-12 educational system is framed in terms of student outcomes, particularly through various assessments designed to evaluate the extent to which our students are successfully learning and meeting our high academic expectations.

In this chapter, we begin by benchmarking the achievement of Minnesota students against that of students from other countries using data from international studies in reading, mathematics, and science. Then we turn our attention toward the most recent results from our nation's "Report Card," the National Assessment of Educational Progress (NAEP). These

data allow us to benchmark the achievement of Minnesota students against that of other states in the U.S. Next, we turn to a more detailed examination of achievement within the state using Minnesota's statewide tests: the Minnesota Comprehensive Assessment (MCA) in third and fifth grades and the Basic Standards Tests (BST) in eighth grade. Finally, we examine the performance of

college-bound students as reflected in the college admissions test most frequently taken by high school students in Minnesota, the ACT Assessment (ACT).

Comparing the Achievement of Minnesota Students to Other Nations

Many significant challenges present themselves when comparing Minnesota students' performance to that of other countries. National differences in curricular content, instructional approaches, language, and school organization make it difficult to design achievement measures that are truly comparable across countries. Of the subjects reviewed here—reading, mathematics, and science—reading poses the greatest challenges. Students from the differing countries are learning to read different languages spelled with varying alphabets or symbol systems. In the grades and subject areas where data are available, we try to answer the question, "How does the achievement of students in Minnesota compare to that of students in other countries?"

READING ACHIEVEMENT AMONG NATIONS

Starting in 1989, the United States participated along with 31 other nations in the International Reading Literacy Study sponsored by the International Association for the Evaluation of Educational Achievement (IEA), an independent international cooperative of research centers

MINNESOTA COMPREHENSIVE ASSESSMENTS

The newest statewide tests in Minnesota are the *Minnesota Comprehensive Assessments* (MCAs). Their content is based on the *Preparatory Standards* in the Graduation Rule.

At third grade, students are tested in reading and mathematics. At fifth grade, they take tests in reading, mathematics, and writing. For both grade levels, the reading and mathematics tests include a combination of multiple choice and open-ended items. For the writing portion, students must prepare a short composition.

The tests are scored using four possible proficiency levels:

• Level 1: Students demonstrate little or no evidence of the knowledge and skills required for satisfactory work in the subject area.

• Level II: Students demonstrate partial evidence of the knowledge and skills

necessary for satisfactory work in the subject area.

• Level III: Students demonstrate evidence of solid academic performance and competence in the knowledge and skills of the subject.

• Level IV: Students demonstrate superior performance in the subject area, beyond what is expected.

By demonstrating solid academic performance on the *Preparatory Standards*, students at Level III or above have demonstrated more than simple minimum competency. There is reason for concern about the performance of students at Level I.

In addition to a level, students are given a scale score whose statewide mean varies by subject area, but which stays at approximately 1400 for all students statewide.

and departments of education in more than 50 countries. In the United States, data were collected on approximately 7,200 students in the fourth grade and 3,800 students in the ninth grade at 332 public and private schools. The sample included schools distributed in 227 school districts across 31 states and the District of Columbia.

Generally speaking, American fourth grade students per-



formed extremely well. Only Finland outperformed American fourth graders (U.S. Department of Education, 1996). The fourth grade total reading achievement score for American students (547) was statistically lower than Finland (569), statistically equal to the achievement of Swedish children, and statistically higher than the remaining 24 nations participating in this portion of the study (see Table 5.1).

The ninth grade reading total achievement score for U.S. students was 535, which was essentially equal to the scores of 15 other countries. Once again, American students were significantly lower than only one country (Finland, whose students scored 560), and statistically higher than 14 nations (see Table 5.2).

DRAWING CONCLUSIONS FROM THE IEA STUDY FOR MINNESOTA **STUDENTS**

Literacy Study did not allow for the

The IEA International Reading

Finland 569

Nations with Average Scores

Significantly Higher than United

States

United Sta	ates	547	France	531
Sweden		539	Italy	529
			New Zealand	528
			Norway	524
			Iceland	518
			Hong Kong	517
			Singapore	515
			Switzerland	511
			Ireland	509
			Belgium - French	507
			Greece	504
			Spain	504
			Germany (West)	503
			Canada (B. Columbia)	500
			Germany (East)	499
			Hungary	499
			Slovenia	498
			N etherlands	485
			Cyprus	481
			Portugal	478
			Denmark	475
			Trinidad/Tobago	451
			Indonesia	394
			Venezuela	383

Nations with Average Scores

Significantly Lower than United States

reporting of scores at the individual state level; consequently we cannot directly say anything about the performance of Minnesota students in fourth and ninth grade as compared to those of students in other

> Despite the decline in scores from 1992 to 1994, only Maine's students clearly outscored Minnesota students, and the reading achievement of our fourth graders remains near the top nationally. Since American fourth graders' average scores on the IEA study were exceeded by only one country, we can be relatively sure that Minnesota students' performance in reading, at least at the fourth grade level, would be very competitive internationally.

Table 5.2 IEA Ninth Grade Reading Achievement: Average Scores

Nations with Average Scores Significantly Higher than United States		Nations with Average Sc Significantly Different fro States	Scores Not Nations with Average Sc from United Significantly Lower than U States		
Finland	560	France	549	Norway	516
		Sweden	546	Italy	515
		New Zealand	545	Ireland	511
		Hungary	536	Greece	509
		Iceland	536	Сургиз	497
		Switzerland	536	Spain	490
		Hong Kong	535	Belgium - French	481
		United States	535	Trinidad/Tobago	479
		Singapore	534	Thailand	477
		Slovenia	532	Philippines	430
		Germany (East)	526	Venezuela	417
		Denmark	525	Nigeria	401
		Portugal	523	Zimbabwe	372
		Canada (B. Columbia)	522	Botswana	330
		Germany (West)	522		
		Netherlands	514		



Table 5.1 IEA Fourth Grade Reading Achievement: Average Scores

Nations with Average Scores

Not Significantly Different from

United States

countries. But, as we shall see later, the results of this country's National Assessment of Educational Progress (NAEP) in fourth grade reading show Minnesota fourth graders' performance to be well above the national norm. Minnesota fourth graders score well above the United States average on NAEP, and the United States average in the IEA study is significantly exceeded by that of only one country. Therefore, we can be relatively sure that Minnesota students' performance would be competitive with those students from even the best countries participating in the IEA study, at least at the fourth grade level.

INTERNATIONAL MATHEMATICS ACHIEVEMENT

The Third International Mathematics and Science Study (TIMSS), also coordinated by IEA, is the largest comparative assessment study of mathematics and science education to date. Forty-five nations participated

Table 5.3 Minnesota Fourth Grade Mathematics Performance in TIMSS, Compared to the U.S. and Other Nations

Nations with Average Scores Significantly Higher than Minnesota		Nations with Average S Significantly Differe Minnesota	Scores Not nt from	Nations with Average Scores Significantly Lower than Minnesota		
Singapore	625	Austria	559	Scotland	520	
Korea	611	Slovenia	552	England	513	
Japan	597	Ireland	550	Cyprus	502	
Hong Kong	587	Hungary	548	Norway	502	
Netherlands	577	Australia	546	New Zealand	499	
Czech Republic	567	United States	545	Greece	492	
		Minnesota	542	Thailand	490	
		Canada	532	Portugal	475	
		Israel	531	Iceland	474	
		International Avg	52 9	Iran, Islamic Rep	429	
		Latvia (LSS)	525	Kuwait	400	

in the various components of TIMSS, which included student assessments, a curriculum content analysis, and questionnaires for both students and teachers. During the academic year of 1994-95, approximately 34,000 American students in grades 3-4, 7-8 and 12 participated. Additionally, SciMath^{MN} (a state partnership of Minnesota business, education, and

 Table 5.4
 Minnesota Eighth Grade Mathematics Performance in TIMSS, Compared to the U.S. and Other Nations

Nations with Average Scores Significantly Higher than Minnesota		Nations with Average Significantly Differe Minnesota	Scores Not nt from	Nations with Average Scores Significantly Lower than Minnesota		
Singapore	643	Slovak Republic	547	Latvia	493	
Korea	607	Switzerland	545	Spain	487	
Japan	605	Netherlands	541	Iceland	487	
Hong Kong	588	Slovenia	541	G reece	484	
Belgium -Flem ish	565	Bulgaria	522	Romania	482	
Czech Republic	564	Austria	539	Lithuania	477	
		France	538	Cyprus	474	
		Hungary	537	Portugal	454	
		Russian Federation	535	Iran, Islamic Rep	428	
		Australia	530	Kuwait	392	
		Ireland	527	Colombia	385	
		Canada	527	South Africa	354	
		Belgium - French	526			
		M in n eso ta	525			
		Sweden	519			
		T hailan d	522			
		Israel	522			
		International Avg	513			
		Germany	509			
		New Zealand	508			
		England	506			
		Norway	503			
		Denmark	502			
		United States	500			
		Scotland	498			

ERIC FullTaxt Provided by ERIC government pursuing statewide improvement in the teaching and learning of K-12 mathematics and science) sponsored nearly 5,000 Minnesota students to participate as a "mini-nation." Mini-nation status makes it possible to compare Minnesota results with the U.S. as a whole, and with other countries in the study.

Fourth Grade International Math Performance

To date, SciMath^{MN} has analyzed Minnesota fourth and eighth grade mathematics and science results. In mathematics, Minnesota fourth graders scored above the international average (529) of the 26 participating countries. Minnesota's average mathematics performance (542) was not statistically different from the U.S. (545) as a whole (Table 5.3, above). The Minnesota average score was lower than six nations, statistically equal to nine nations including the United States, and significantly higher than eleven other countries.

In the various content areas covered by the TIMSS math assessment, Minnesota fourth graders exceeded the international average in four of the six areas tested: (1) whole numbers; (2) data representation, analysis, and probability; (3) geometry; and (4) patterns, relations, and functions. In the other two areas (fractions and proportionality; and measurement, estimation, and number sense), the Minnesota fourth grade average was lower than the international average (SciMath^{MN}, 1998).

Eighth Grade International Math Performance

Minnesota eighth graders participating in TIMSS averaged a score of 525, above both the international average (513) and the U.S. average (500). They scored statistically lower than 6 nations, about the same as 23 nations and statistically higher than 12 nations (see Table 5.4, p. 31).

Minnesota eighth graders scored higher than the U.S. and international averages in the content areas of fractions and number sense; proportionality; algebra; and data representations, analysis, and probability. Our eighth graders scored lower than the international average, but above the U.S. average in the content areas of measurement and geometry. Minnesota was in the middle category of performance for all content areas.

DRAWING CONCLUSIONS FROM THE TIMSS MATH ASSESSMENT FOR MINNESOTA STUDENTS

Based on our performance in the TIMSS mathematics assessment and an analysis of our curriculum, SciMath^{MN} (1997) recommends a number of curricular changes (e.g., that algebra and geometry become part of our standards for eighth grade mathematics). A re-analysis of our mathematics curriculum seems warranted, particularly in seventh and eighth grades.

INTERNATIONAL SCIENCE

ACHIEVEMENT

Fourth Grade International Science Performance

The average Minnesota fourth grade score of 577 was well above the international average (524) of the 26 participating countries in the TIMSS

 Table 5.5
 Minnesota Fourth Grade Science Performance in TIMSS, Compared to the U.S. and Other Nations

Nations with A Significantly Minn	Average Scores Higher than lesota	Nations with Ave Not Significantly D Minneso	rage Scores Different from Data	Nations with Avera Significantly Low Minnesota	h Average Scores tly Lower than nnesota	
Korea	597	Minnesota	577	Netherlands	557	
		Japan	574	Czech Republic	557	
		United States	565	England	551	
		Austria	565	Canada	549	
		Australia	562	Singapore	547	
				Slovenia	540	
				Ireland	539	
				Scotland	530	
				Hong Kong	533	
				Hungary	53	
				New Zealand	53	
				Norway	530	
				International Avg	524	
				Latvia (LSS)	512	
				Israel	50	
				Iceland	50	
				Greece	493	
				Portugal	48	
				Сургаз	47	
				Thailand	473	
				Iran, Islamic Rep	410	
				Kuwait	40	

science assessment. Minnesota's average score was above, but not statistically different from, the U.S. average (565) as a whole. Only one country was statistically higher than Minnesota, Korea (597); four nations were statistically equivalent to Minnesota; and 21 nations were statistically lower than Minnesota (see Table 5.5). In the four content areas covered by the TIMSS science assessment, Minnesota fourth graders exceeded the international average in each area, as did the U.S. as a whole. In three of these areas-earth science, life science, and environmental issues/nature of science-Minnesota fourth-graders were among the top group of countries (SciMath^{MN}, 1998).

Eighth Grade International Science Performance

Minnesota eighth graders' overall scale score in science performance (565) was significantly above the U.S. (534) and international averages (516). Only eighth grade students in Singapore (607) significantly outscored Minnesota students in the TIMSS science assessment. Fourteen countries' scores were not significantly different from Minnesota; and Minnesota was statistically higher than 26 countries, including the United States, and statistically higher than the international average (see Table 5.6, p. 33). The TIMSS science scores show the power of instructional alignment with Minnesota's science curriculum. In the five eighth grade science content areas (earth science, life science, chemistry, physics, and environmental science), Minnesota eighth graders exceeded the international and the national average in every area, and in two of these Minnesota students scored among the highest performing groups.

DRAWING CONCLUSIONS FROM THE TIMSS SCIENCE ASSESSMENT FOR MINNESOTA STUDENTS

Minnesota's methods for teaching elementary science are reflected in our



Nations with Average Scores Significantly Higher than Minnesota		Nations with Averag Not Significantly D from Minneso	e Scores ifferent ta	Nations with Average Scores Significantly Lower than Minnesota		
Singapore	607	Czech Republic	574	Sweden	535	
		Japan	571	United States	534	
		Korea	565	Germ any	531	
		Bulgaria	565	Canada	531	
		M innesota	565	Norway	527	
		N etherlands	560	New Zealand	525	
		Slovenia	560	T hailan d	525	
		Austria	558	Israel	524	
		Hungary	554	Hong Kong	522	
		England	552	Switzerland	522	
		Belgium -Flem ish	550	Scotland	517	
		Australia	545	Spain	517	
		Slovak Republic	544	International Avg	516	
		Russian Federation	538	France	498	
		Irelan d	538	Greece	497	
				Ireland	494	
				Romania	486	
				Latvia (LSS)	485	
				Portugal	480	
				Denmark	478	
				Lithuania	476	
				Belgium - French	471	
				Iran, Islam ic Rep	470	
				Cyprus	463	
				K u w ait	430	
				Colom bia	411	
				South Africa	326	

 Table 5.6
 Minnesota Eighth Grade Science Performance in TIMSS, Compared to

 the U.S. and Other Nations

overall outstanding performance in the TIMSS science assessment. The traditional Minnesota elementary science program uses a hands-on structure to emphasize the children's natural curiosity both inside and outside the classroom. Our state's traditional approach to teaching science emphasizes a considerable amount of class discussion and laboratory work, which may help explain the relatively strong performance of Minnesota students in science (SciMath^{MN}, 1998).

Comparing the Achievement of Minnesota Students to The Nation

The National Assessment of Educational Progress (NAEP) is the major system of measurement designed to give a picture of student achievement across the United States. Often referred to as "The Nation's Report Card," it tests a stratified random sample of students at the fourth, eighth and twelfth grades in several subjects to measure performance for the nation at each grade. In addition to these measures, state-by-state comparisons are reported for reading, mathematics, and science at the fourth and/or eighth grade levels. This section examines the performance of Minnesota fourth and eighth graders in these three content areas, in comparison to their peers nationally.

NAEP uses three performance levels to describe a student's mastery of the subjects being tested: Basic, Proficient, and Advanced. The Proficient Level (the one used in this section to benchmark Minnesota students' performance) is described as representing "solid academic performance." Students reaching this level have "demonstrated competency over challenging subject matter, including subjectmatter knowledge, application of such

46

knowledge to real world situations, and analytical skills appropriate to the subject matter" (Shaughnessy, C. A., Nelson, J. E., & Norris, N. A., 1998, p. 40).

The interpretation of NAEP proficiency levels has some of the ambiguity associated with letter grades in school. As parents, we would like our children to get "A's" and "B's" on their report cards, but we know that it would be unrealistically demanding to expect all of our children to get "A's" in every subject every year. Similarly, we would like all of our children to score at or above the "Proficient" level in the NAEP system, but at the same time we know how unrealistic and demanding that would be. Rather than expecting all children in Minnesota to score at or above the Proficient level, we compare the percentage of children in Minnesota who reach that level to the percentage in other states as a way to benchmark achievement in Minnesota against that in other states. In short, "How does the performance of students in Minnesota compare to that of students in other states?"

Fourth Grade Reading Achievement

Figure 5.1 (p. 34) displays the results of the NAEP Reading "Report Card" for fourth graders in both 1992 and 1994, the most recent years for which such data are available. As a whole, the performance of Minnesota fourth graders surpassed the national average in both years. However, when we consider student reading performance over time, Figure 5.1 shows Minnesota's fourth grade reading performance *declining* slightly during this two-year period. By 1994, Minnesota ranked 14th among the participating states in its overall mean scale score of 218, although differences between our average and those of the higher ranked states were not statistically significant (with the exception of Maine and its mean scale score of 228). This decline of scores between



1992 and 1994 reflected a national trend.

Using the percentage of students achieving at the Proficient Level, Figure 5.2 reveals the performance of selected subgroups in the NAEP 1994 Reading Assessment in both Minnesota and the nation as a whole. Overall, a greater percentage of Minnesota fourth graders achieved proficiency than fourth graders nationally. However, when gender and ethnicity are taken into account, scores from the various subgroupings are all statistically similar to their national counterparts.



Figure 5.2 1994 NAEP Grade 4 Reading: Percent Proficient by Subgroup



other states. Despite the decline in scores from 1992 to 1994, only Maine's students clearly outscored Minnesota students, and the reading achievement of our fourth graders remains near the top nationally.

Fourth Grade Mathematics Achievement

Figure 5.3 shows the mean scale score of fourth grade students in the 1992 and 1996 NAEP Mathematics Assessment, for both Minnesota and the nation. Over the four-year period covered by this assessment, achievement in Minnesota and the nation improved. Scaled scores in Minnesota rose from 228 to 232, with the proportion of students scoring at the Proficient level increasing by 3%.

DRAWING CONCLUSIONS FROM THE NAEP FOURTH GRADE READING ASSESSMENT

There are disappointing trends in these data. First, there is the decline in scores from 1992 to 1994. We strongly hope that the upcoming 1998 NAEP reading assessment will show a reversal of this trend. Second, Minnesota's ethnic groups did *not* consistently outscore their counterparts nationwide. Nevertheless, our major purpose in examining these data was to benchmark the performance of Minnesota students against those from



Figure 5.3 1992-96 NAEP Grade 4 Mathematics Results for Minnesota and the Nation



Figure 5.4 1996 NAEP Grade 4 Mathematics: Percent Proficient by Subgroup



the nation. Specific student groupings are broken out by gender, ethnicity, and socioeconomic status (i.e., eligible for free and reduced lunch vs. not eligible). Both boys and girls in Minnesota outscore their counterparts nationally. All Minnesota subgroups had a larger proportion of students scoring at or above the Proficient level than their national counterparts, though most of these differences are considered statistically insignificant (with the exception of students eligible for free and reduced lunch).

Figure 5.5 1990-92-96 Grade 8 Mathematics Results for Minnesota and the Nation

Minnesota fourth grade students scored well above the national average both years. In 1996, none of the 44 states which participated in the Grade 4 State NAEP mathematics test administration had a scaled score mean above 232 (although Maine and Connecticut tied with Minnesota), and only Connecticut had more fourth graders at the Proficient Level—31% as compared to 29% in Minnesota.

The success of this performance by Minnesota fourth graders in 1996 is further detailed in Figure 5.4, which shows the comparative performance of various Minnesota student subgroups against their national counterparts. While both boys and girls showed significantly higher percentages of students achieving proficiency,

> At both fourth and eighth grade, the achievement of Minnesota students compared extremely well with that of other states in the U.S., but in mathematics, comparing favorably to other states does not make the performance internationally competitive. Based on American students' performance in the TIMSS assessment, a re-analysis of our mathematics curriculum seems warranted.



no ethnic group performed significantly different than its national counterpart.

Eighth Grade Mathematics Achievement

Figure 5.5 shows NAEP data for eighth graders over the period from 1990 to 1996. Minnesota students score above the national average in all three years tested. In 1996, the Minnesota student average of 284 met or exceeded that of all the other 41 participating states.

Figure 5.6 (page 36) shows the percentage of public school eighth graders performing at the Proficient level in the 1996 NAEP Mathematics Assessment, for both Minnesota and DRAWING CONCLUSIONS FROM THE NAEP MATHEMATICS ASSESSMENTS

No state clearly outperformed Minnesota students at either fourth or eighth grades. At both fourth and eighth grade, the achievement of Minnesota students compared extremely well to that of other states in the U.S. However, this does not mean that Minnesota's scores are highly competitive internationally.

Eighth Grade Science Achievement

The State Level NAEP Science Assessment was first administered in 1996 to eighth grade students in 40 states and U.S. jurisdictions. It enlisted the use of a new testing framework that featured many more



constructed-response questions and hands-on tasks to allow students to exhibit their abilities in integrating scientific concepts and in conducting scientific investigations. Minnesota's public school eighth graders achieved an average NAEP science scale score of 159, well above the national average scale score of 148, and significantly exceeded by only two other participating states (Maine and North Dakota).

Figure 5.7 displays the percentage of students within certain subpopulations that achieved at or above the Proficient Level of performance in the 1996 NAEP Science Assessment. As in the case of the 1996 NAEP Mathematics Assessment, each Minnesota eighth grade subgroup outperformed its national counterpart; however, only in the case of boys was this difference considered statistically significant.

DRAWING CONCLUSIONS FROM THE NAEP SCIENCE ASSESSMENTS

Minnesota's performance in the NAEP science assessment is exceptional, with each student subgroup outperforming its national counterpart, and the state's average scale score significantly exceeded by only two other participating states. Future administrations of the NAEP science assessment will allow us to examine trends in our state's overall levels of science achievement.



Student Subgroup

Note: "ns"= not significant.

GENERAL CONCLUSIONS

Our ability to compare Minnesota student performance in reading to that of students in other states is limited by the fact that NAEP's State Level Reading Assessment has only been conducted on the achievement of fourth graders, with results available only for 1992 and 1994. Although Minnesota's overall mean scale score dipped slightly between these two administrations of the NAEP Reading Assessment, it is worth remembering that our state's 1994 overall mean scale Minnesota's public school eighth graders achieved an average NAEP science scale score of 159, well above the national average scale score of 148, and significantly exceeded by only two other participating states (Maine and North Dakota). In the most recent NAEP assessment, only eighth grade students in Singapore significantly outscored Minnesota students.

score was significantly lower than that of only one other participating state (Maine), and was, in fact, statistically higher than 13 other participating jurisdictions. We are concerned, however, about the decline in scores from 1992 to 1994. Because of the emphasis on reading fostered by the Minnesota Graduation Standards, we hope to find that this decline has been reversed when NAEP releases findings from the 1998 administration of its state-level reading assessment.

While there continues to be concern about the unevenness of mathematics performance across different ethnic groups, the NAEP math findings show

Figure 5.7 1996 NAEP Grade 8 Science: Percent Proficient by Subgroup



49

several encouraging trends. First, Minnesota students scored well above the national average at both fourth and eighth grades and are virtually at the top of the states that have participated in the state-by-state comparisons. Second, scores have been increasing in Minnesota. It must be remembered, however, that in mathematics, being among the top states in the U.S. does not place Minnesota in the very top tier internationally.

Trend information in the NAEP Science Assessment is not yet available, since the state-level test was administered for the first time in 1996. Nevertheless, the overall performance of Minnesota eighth graders in the 1996 NAEP Science Assessment was impressive, when compared to the rest

significantly higher mean scale scores, and every student subgroup in Minnesota outperformed its counterpart nationally.

STUDENT PERFORMANCE IN THE MINNESOTA ACHIEVEMENT TESTING PROGRAMS

The international and national studies provide a comparison of Minnesota student performance to that of students in other states and countries. Because the studies include only a sample of Minnesota students, these data do not provide a detailed look at achievement within the state. We now turn to data from Minnesota's

of the nation. Only two states reported statewide tests to look at performance across regions and segments of Minnesota.

> In 1997-98, Minnesota began statewide testing in grades 3, 5, and 8 for all Minnesota students. The third and fifth grade examinations, called the Minnesota Comprehensive Assessments, or MCAs, measure reading and mathematics performance in third grade, and reading, mathematics, and writing performance in fifth grade. At both grade levels, the tests are aligned with the High Standards articulated in the Profile of Learning. The reading and mathematics portions contain both multiple-choice and short answer items, whereas the fifth grade writing test asks for a sample of the student's writing.

MINNESOTA'S PROFILE OF LEARNING **PERFORMANCE PACKAGES: A SNAPSHOT** FROM GRAND RAPIDS, MN LEARNING AREA: SCIENTIFIC APPLICATION

Last year, middle school science teacher, Cheryl Smith, and her colleagues at Grand Rapids Middle School, implemented Tasks 2 and 3 from the performance package, "Weather," developed by the Department of Children, Families, and Learning (CFL).

After four weeks of studying weather, experimenting with instrumentation, and recording data, the students were asked to develop a question for further study. One student proposed this question and hypothesis (see darker box at center):

Once the question and hypothesis were approved, the students had to design a plan for data collection. This same student proposed the following data collection procedure:

Data Collection Design

She would collect temperature data for the months of December and January using the weather station in her science class and her father's weather station at home. The data would be collected at the same time every day. For previous years, she would also use weather data collected by the Forestry Experiment Laboratory at the Itasca Community College campus.

Question

How does the average temperature in December and January in Grand Rapids of this El Nino year compare with the average for the 5 most recent El Nino years and the 5 most recent non-El Nino years?

Hypothesis

"I think the order of averages will be, from highest to lowest: a) this year's average, b) the 5 other El Nino years' averages, and c) the 5 non-El Nino years' average temperatures. I think they will be in this order because this year's El Nino is predicted to be the largest one on record, and the other El Nino years are supposed to be warmer than non-El Nino years."

Once she had the daily temperatures for her years of study, she would calculate the monthly averages. This information would be recorded on weather data tables and eventually graphed.

Next the data collection plan had to be approved by the teacher. A checklist provided the students with an extra measure of assistance in the development of their plan.

Much of the research for the students' field study had to be conducted on their own time—"A big challenge for many students," Cheryl observed. Students were required to use three outside sources in their quest for the answer to their question. Cheryl commented that this was particularly difficult for some students who didn't have access to the Internet or transportation to the local library. In contrast, some students were very resourceful. "I was surprised at the number of students who made

calls to news stations, airports, regional weather stations, or visited our university extension site." To further assist students in accessing resources, Cheryl built in research time during the school day. A progress checklist helped keep students on task and on the right track. It also helped Cheryl monitor the progress of her students.

The end product was a written report with appropriate graphs, charts, and written documentation. Students were expected to synthesize the information into a report that defended or refuted (cont'd on p. 38)

50

SCIENTIFIC APPLICATION (cont'd from p. 37)

their particular weather hypothesis.

The report submitted by this student contained several observation tables documenting the average temperatures for her years of study. Using this information she created several computer generated bar graphs. She also included line graphs documenting humidity, atmospheric pressure, and

Conclusion

break all records."

"I conclude that my data refutes my

hypothesis. My original hypothesis was that

this year's average would be the highest,

followed by the other El Nino years, then

the non-El Nino years. This was not so. The

warmest temperature was an El Nino year,

predictions that this year's El Nino would

1982-83, but it was not this year, despite the

temperature. The inset contains an *excerpt* from her conclusions:

Since Cheryl was **piloting** Tasks 2 and 3 of the performance package on

"Weather," she evaluated each student's performance based on the four-point scale developed by CFL to score the entire content standard (see box on p. 27). (Typically, the guidelines on the checklists evaluate work using "Y" or "N" (Y=yes, the student met the performance standard; N=no, the

In eighth grade, students take the multiple-choice Basic Standards Tests (BSTs), which cover reading and mathematics and are aligned with the basic standards in the Minnesota Graduation Rule. The eighth grade test is the student's first chance to demonstrate mastery of the high school basic requirements. Any student correctly answering at least 75% of the items meets the high school requirement set by the Minnesota Board of Education for reading or mathematics. Students who do not meet the minimum graduation standard will have additional opportunities to retake the test in later grades.

The BSTs have a clear passing score, 75%. The MCAs, however, use proficiency levels similar to NAEP assessments. (See sidebar text on p. 29 for definitions of the various levels of student performance in the MCA testing program.) These proficiency student did not meet the performance standard.)

The performance requirements for Task 3 included such things as: summarizing field study information accurately, supporting or refuting the hypothesis, and providing justification for conclusions about the hypothesis, using collected data and appropriate

scientific concepts and principles.

Cheryl commented that at first, the students were overwhelmed by the performance package. The content standard, with its accompanying performance package, pushed

the students beyond what had previously been expected of them. The solution was to break everything down into small steps. The checklists in the performance package made this possible, and it placed more of the responsibility for learning on the students. This created a path that students could follow to succeed.

ð>

levels are similar to grades in that they describe where students' achievement falls on a continuum; also, as with letter grades, all students will not reach the same point on the continuum in each test every year. That is, while we might wish all our children earned A's in every subject every year, it would be unrealistically demanding to expect it. Similarly, we genuinely wish all children could score at Level III on the tests, but this is a level of accomplishment that cannot be realistically expected of all children. In fact, school districts in the metropolitan area have found that Level III performance is equivalent to very high performance (ranging from the 68th to the 73rd percentile) on other norm-referenced achievement tests administered by these districts (J. Angermeyer, personal communication, Nov. 2, 1998). Continuing the analogy with grades, C's are passing grades representing an

acceptable level of performance, but they are not all we might wish for our children. Level II is an acceptable level of performance, but it is less than we might hope.

In this section, we are trying to evaluate the performance of students across various segments and regions of Minnesota. After presenting statewide data, we turn to issues of ethnic and gender differences.

Throughout the education literature, achievement test scores are correlated with student poverty (eligibility for free or reduced lunch), mobility (frequent school or residence changes), and limited English proficiency. Therefore, in accordance with the 1998 Minnesota Omnibus Education Act, Subdivision 1 and to provide context for the test scores, our tables include data on the percentage of testtakers who are in poverty, the percentage who recently moved into the district (mobility), and the percentage who are categorized as having limited English proficiency. Also in accordance with Minnesota statute, Appendix A includes additional data on all students except those with limited English proficiency, all students except those new to their district, and all students except those receiving special education. Finally, to comply with the reporting recommendations of the 1997 amendments to the federal Individuals with Disabilities Education Act (IDEA), we have reported separate data on students in special education.

THIRD GRADE *MINNESOTA COMPRE-HENSIVE ASSESSMENT* RESULTS IN READING AND MATHEMATICS

Tables 5.7 and 5.8 (p. 39) show results for all third grade students in the MCA reading and mathematics assessments. Tables A.1 through A.6 (see Appendix A) show the data for all students except those with limited English proficiency, those new to their district since January 1977, and those



in special education.

More than 60,000 students took the tests, or 93% of the third graders enrolled at the time of testing. Seventy-seven percent of tested third graders scored at or above Level II in the MCA reading test; 82% were at or above Level II in mathematics. On both tests, 35% of the students scored at or above Level III.

FIFTH GRADE *MINNESOTA COMPRE-HENSIVE ASSESSMENT* RESULTS IN READING, MATHEMATICS, AND WRIT-ING

Tables 5.9 through 5.11 show the fifth grade MCA results in reading,

Table 5.71998 Grade 3: MinnesotaComprehensive AssessmentResults in Reading for all PublicSchool Students Tested	Сатедогу	Number Tested	% At or Above Level III	% At or Above Level II	Mean Scale Score	% Enrolled Students Tested	% LEP Students in Score	% Special Ed Students in Score	% New to District Since 1/1/97	% F/R · Students in Score	
	Total	60577	35	77	1410	93	4	11	10	31	
		Girls	29792	41	82	1435	94	4	7	10	32
		Boys	30663	30	73	1386	92	5	15	10	31
		Asian	2847	17	52	1302	93	62	5	13	70
		Black	3692	11	46	1264	90	4	14	16	82
		Hispanic	1486	16	54	1300	86	38	10	18	70
		American Indian	1204	15	56	1303	89	0+	18	17	75
		White	50542	39	83	1434	94	0+	11	9	24
		Special Ed	6696	12	41	1248	82	2	_	10	44
		LEP	2612	4	34	1222	87	-	6	13	87
		Metro Area	32683	37	76	1410	92	7	10	10	30
		Outstate	27759	34	79	1410	93	2	12	10	34
		Mpls/St. Paul	7792	18	51	1298	89	22	9	11	69
		TC Suburbs	24846	42	84	1445	93	2	10	10	17
		Outstate: 2000+	13700	34	79	1409	92	3	12	8	31
		Outstate: 2000 -	14059	34	79	1411	94	1	12	11	36
		Public/Charter	276	21	52	1302	82	13	12	56	62
		Public/Not	60301	35	78	1410	93	4	11	10	31
		Non-public	1315	43	88	1455	-	_	_		

52

Table 5.8 1998 Grade 3: Minnesota Comprehensive Assessment Results in Mathematics for all Public School Students Tested

Note: LEP=Limited English Proficiency; F/R=Eligible
for free or reduced-price lunch; 0+ indicates a .4% or
less. All percentages and Mean Scale Scores are
rounded to the nearest whole number.



mathematics, and writing for all public test, and all students except those in school students tested. Tables A.7 through A.15 (see Appendix A) show results for all students except those with limited English proficiency, all students who attended school in their district for at least one year before the

special education.

More than 60,000 students (95% of all those enrolled) took the MCA fifth grade assessments. Thirty-eight percent demonstrated the solid academic

performance expected at Level III or above in reading. The corresponding figures for mathematics and writing were 31% and 42%. In reading, mathematics, and writing, 79%, 80%, and 80% respectively, scored at or above Level II.

Table 5.9	<i>1998 Grade 5:</i> Minnesota Comprehensive Assessment <i>Results in</i> <i>Reading for all Public School</i>	Category	Number Tested	% At or Above Level III	% At or Above Level II	Mean Scale Score	% Enrolled Students Tested	% LEP Students in Score	% Special Ed Students in Score	% New to District Since 1/1/97	% F/R Students in Score
	Students Tested	Total	60492	38	79	1419	95	4	13	9	29
		Girls	29484	43	83	1444	96	3	8	9	29
		Boys	30958	34	76	1395	94	4	17	9	29
		Asian	2786	22	59	1323	97	52	8	11	65
		Black	3271	13	46	1254	93	4	20	16	77
		Hispanic	1305	16	54	1290	91	34	17	17	66
		American Indian	1165	15	58	1292	90	0+	21	13	73
		White	51088	42	84	1443	96	0+	13	8	22
		Special Ed	7794	10	39	1230	89	3	_	9	42
		LEP	2154	4	33	1201	93		12	13	88
		Metro Area	31575	40	78	1423	95	6	12	9	27
		Outstate	28633	36	80	1415	95	1	14	8	31
		Mpls/St. Paul	7009	21	54	1301	94	20	15	10	66
		TC Suburbs	24566	45	85	1458	96	1	12	8	16
		Outstate: 2000+	14054	38	80	1419	94	2	14	8	28
		Outstate: 2000 -	14579	35	80	1412	96	1	13	9	34
Note: LEP=Lim	ired English Proficiency: E/R-Eligible for	Public/Charter	233	26	59	1322	94	13	24	51	50
free or reduced-	price lunch; 0+ indicates a .4% or less.	Public/Not	60259	38	79	1420	95	4	13	9	29
All percentages : nearest whole n	and Mean Scale Scores are rounded to the umber.	Non-public	1334	45	88	1461	_		_	_	_

:

Table 5.10

1998 Grade 5: Minnesota Comprehensive Assessment Results in Mathematics for all Public School Students Tested

Note: LEP=Limited English Proficiency; F/R=Eligible for Note that a summer Lington Hometricy, Proceeding to for free or reduced-price lunch; 0+ indicates a .4% or less. All percentages and Mean Scale Scores are rounded to the nearest whole number.

Category	Number Tested	% At or Above Level III	% At or Above Level II	Mean Scale Score	% Enrolled Students Tested	% LEP Students in Score	% Special Ed Students in Score	% New to District Since 1/1/97	% F/R Students in Score
Total	60362	31	80	1395	95	4	13	9	29
Girls	29305	30	80	1393	95	3	8	9	29
Boys	30995	32	79	1397	95	4	17	9	29
Asian	2779	19	63	1315	96	52	9	11	65
Black	3233	7	41	1211	92	4	20	16	77
Hispanic	1293	11	52	1262	90	34	17	17	66
American Indian	1157	10	55	1273	90	0+	21	14	73
White	51008	35	84	1419	95	0+	13	8	22
Special Ed	7790	11	47	1242	89	3	_	9	42
LEP	2149	4	40	1207	93	_	12	13	88
Metro Area	31419	34	79	1401	95	6	13	9	27
Outstate	28660	28	80	1389	95	1	14	8	31
Mpls/St. Paul	6948	16	54	1281	93	20	15	10	66
TC Suburbs	24471	39	86	1435	95	1	12	8	16
Outstate: 2000+	14022	29	81	1394	94	2	14	7	28
Outstate: 2000 -	14638	28	80	1385	96	1	13	9	34
Public/Charter	234	18	60	1294	95	14	23	50	50
Public/Not Charter	60128	31	80	1395	95	4	13	8	29
Non-public	1329	33	89	1420	_	_	—		_

53

EIGHTH GRADE BASIC STANDARDS

TESTS IN READING AND MATHEMATICS

Tables 5.12 (below) and 5.13 (p. 42) show the eighth grade Basic Standards Test results in reading and mathematics students except those in special for all students tested. Tables A.16 through A.21 (see Appendix A) show

the results for all students except those with limited English proficiency, all students except those new to the district since January 1, 1997, and all education.

Over 64,000 students participated in the testing, 96% of all eighth graders enrolled. Sixty-eight percent of the 64,000 test-takers met the state's minimum standard for high school graduation in reading. Seventy-one percent met the standard in mathemat-

% New

Table 5.11 1998 Grade 5: Minnesota **Comprehensive Assessment** Results in Writing for all Public School Students Tested

Category	Number Tested	% At or Above Level III	% At or Above Level II	Mean Scale Score	% Enrolled Students Tested	% LEP Students in Score	% Special Ed Students in Score	% New to District Since 1/1/97	% F/R Students in Score
Total	60364	42	80	1393	95	4	13	9	29
Girls	29420	52	87	1496	96	3	8	9	29
Boys	30891	32	74	1296	94	4	17	9	29
Asian	2757	35	76	1325	96	52	8	11	65
Black	3184	21	57	1131	91	4	19	16	77
Hispanic	1274	25	64	1202	89	33	17	16	66
American Indian	1139	19	61	1145	88	0+	20	13	73
White	51123	45	83	1426	96	0+	12	9	22
Special Ed	7607	15	51	1041	87	3	_	9	42
LÉP	2088	18	60	1141	90	—	12	12	88
Metro Area	31443	44	81	1413	95	5	12	9	27
Outstate	28642	39	79	1372	95	1	13	8	31
Mols/St. Paul	6884	29	65	1227	93	20	14	10	66
TC Suburbs	24559	49	85	1465	96	1	12	8	16
Outstate: 2000+	14055	40	80	1378	94	2	13	7	28
Outstate: 2000 -	14587	38	79	1366	96	1	13	9	34
Public/Charter	231	31	64	1207	94	13	21	50	50
Public/Not	60133	42	80	1394	95	3	13	8	29
Onarter Non-public	1334	43	81	1409	—	_	-	_	—

Note: LEP=Limited English Proficiency; F/R=Eligible for free or reduced-price lunch; 0+ indicates a .4% or less. All percentages and Mean Scale Scores are rounded to the nearest whole number.

1998 Grade 8: Basic Table 5.12 Standards Test Results in Reading for all Public School Students Tested

Note: LEP=Limited English Proficiency; F/R=Eligible for free or reduced-price lunch; 0+ indicates a .4% or less. All percentages and Mean Scale Scores are rounded to the nearest whole number.

.

	6)
EI	R	[C]
▲ Full Tex	t Provide	d by ERIC

Total 64403 68 31.19 96 2 12 7 Girks 31146 71 31.76 96 2 7 7 Boys 32416 66 30.79 96 3 16 7 Asian 2769 48 27.83 93 36 7 10 Black 2684 32 24.38 89 5 23 16 Hispanic 1256 39 25.89 88 27 16 18 Arrenican Inclian 1134 38 26.06 88 0+ 24 14 Write 55098 73 32.01 97 0+ 11 6 LEP 1579 16 21.58 85 - 12 14 Special Ed 7530 27 23.18 87 2 - 11 Metro Area 31121 68 31.23 95 4 12	24 24 24 62 72 62
Garks 31146 71 31.76 96 2 7 7 Boys 32416 66 30.79 96 3 16 7 Asian 2769 48 27.83 93 36 7 10 Back 2684 32 2438 89 5 23 16 Hspanic 1256 39 25.89 88 27 16 18 Arrenican Indian 1134 38 2606 88 $0+$ 24 14 White 55098 73 3201 97 $0+$ 11 6 LEP 1579 16 21.58 85 $$ 12 14 Special Ed 7530 27 23.18 87 2 $$ 11 Metro Area 31121 68 31.23 96 1 12 7 Mpls/St. Paul 6215 41 2632 89 17 17 9	24 24 62 72 62
Boys 32416 66 30.79 96 3 16 7 Asian 2769 48 27.83 93 36 7 10 Back 2684 32 24.38 89 5 23 16 Hspanic 1256 39 25.89 88 27 16 18 Armerican Indian 1134 38 26.06 88 0+ 24 14 White 55098 73 32.01 97 0+ 11 6 LEP 1579 16 21.58 85 12 14 Special Ed 7530 27 23.18 87 2 11 Metro Area 31121 68 31.23 95 4 12 7 Quissare 32805 68 31.28 96 1 12 7	24 62 72 62
Asian 2769 48 27.83 93 36 7 10 Back 2684 32 24.38 89 5 23 16 Hspanic 1256 39 25.89 88 27 16 18 Arrenican Indian 1134 38 26.06 88 04 24 14 Write 55098 73 32.01 97 04 11 6 LEP 1579 16 21.58 85 12 14 Special Ed 7530 27 23.18 87 2 11 Metro Area 31121 68 31.13 95 4 12 7 Outstate 32805 68 31.28 96 1 12 7 Mpls/St. Paul 6215 41 2632 89 17 17 9	62 72 62
Back 2684 32 24.38 89 5 23 16 Hisparic 1256 39 25.89 88 27 16 18 Arrenican Indian 1134 38 26.06 88 0+ 24 14 White 55098 73 32.01 97 0+ 11 6 LEP 1579 16 21.58 85 12 14 Special Ed 7530 27 23.18 87 2 11 Metro Area 31121 68 31.13 95 4 12 7 Questate 32805 68 31.28 96 1 12 7 Mpis/St. Paul 6215 41 2632 89 17 17 9	72 62
Hispanic 1256 39 25.89 88 27 16 18 Arrenican Indian 1134 38 2606 88 0+ 24 14 White 55098 73 3201 97 0+ 11 6 LEP 1579 16 21.58 85 12 14 Special Ed 7530 27 23.18 87 2 11 Metro Area 31121 68 31.13 95 4 12 7 Questate 32805 68 31.28 96 1 12 7	62
Arrenican Indian 1134 38 2606 88 0+ 24 14 White 55098 73 3201 97 0+ 11 6 LEP 1579 16 21.58 85 12 14 Special Ed 7530 27 23.18 87 2 11 Metro Area 31121 68 31.13 95 4 12 7 Questate 32805 68 31.28 96 1 12 7 Mpls/St. Paul 6215 41 2632 89 17 17 9	
Write 55098 73 3201 97 0+ 11 6 LEP 1579 16 21.58 85 12 14 Special Ed 7530 27 23.18 87 2 11 Metro Area 31121 68 31.13 95 4 12 7 Outstate 32805 68 31.28 96 1 12 7 Mpls/St. Paul 6215 41 2632 89 17 17 9	64
LEP 1579 16 21.58 85 — 12 14 Special Ed 7530 27 23.18 87 2 — 11 Merro Area 31121 68 31.13 95 4 12 7 Ourstate 32805 68 31.28 96 1 12 7 Mpls/St. Paul 6215 41 2632 89 17 17 9	18
Special Ed 7530 27 23.18 87 2 — 11 Metro Area 31121 68 31.13 95 4 12 7 Outstate 32805 68 31.28 96 1 12 7 Mpis/St. Paul 6215 41 2632 89 17 17 9	89
Merro Area 31121 68 31.13 95 4 12 7 Outstate 32805 68 31.28 96 1 12 7 Mpls/St. Paul 6215 41 2632 89 17 17 9	41
Outstate 32805 68 31.28 96 1 12 7 Mpls/Sc Paul 6215 41 26.32 89 17 17 9	22
Mpls/St. Paul 6215 41 2632 89 17 17 9	26
	64
TCSuburbs 24906 75 32.33 9/ 1 10 6	12
Outstate 2000+ 16122 69 31.34 96 2 12 6	23
Outstate: 2000- 16683 68 31.22 97 0+ 12 8	29
Public/Charter 182 43 26.12 98 1 26 45	51
Public/Not 64221 68 31.20 96 2 12 7	24
Non-public 4153 83 34 — — — —	_

Table 5.13

13 1998 Grade 8:B Standards Test M Mathematics fo	asic Results in r all Public	Сатедогу	Number Tested	% Passing	Mean Number Correct	% Enrolled Students Tested	% LEP Students in Score	% Special Ed Students in Score	% New to District Since 1/1/9	% F/R Students in Score
School Students	s Tested	Total	64397	71	53.74	96	3	12	7	24
		Girls	31131	70	53.55	96	2	7	7	25
		Boys	32362	73	54.24	96	3	16	7	24
		Asian	2775	53	48.55	93	36	7	9	62
		Black	2694	26	38.49	89	6	23	15	72
		Hispanic	1239	38	43.53	87	27	15	18	62
		American Indian	1139	39	44.18	88	0+	24	14	64
		White	55051	76	55.38	97	0+	11	6	19
	LEP	1584	23	37.78	85	_	11	14	89	
		Special Ed	7523	29	39.65	87	2		11	41
		Metro Area	31075	70	53.34	95	4	12	7	22
		Outstate	32838	72	54.20	96	1	12	7	26
		Twin Cities	6229	41	43.76	89	17	17	9	64
		TC Suburbs	24846	77	55.74	97	1	10	6	12
		Outstate: 2000+	16145	72	54.41	96	2	12	6	23
		Outstate: 2000-	16693	71	53.99	96	0+	12	8	29
Note: LEP=Limited English Prof	iciency: F/B=Eligible	Public/Charter	178	40	42.37	96	1	27	44	51
for free or reduced-price lunch; 0	+ indicates a .4% or	Public/Not	64219	71	53.77	96	3	12	7	24
to the nearest whole number.	cale scores are rounded	Non-public	4153	82	57.00	_	_	_	_	_

ics. As with the MCA, results varied widely across regions of the state.

The large number of eighth grade students failing to meet the high school minimum on the Basic Standards Tests remains one of the state's most pressing educational problems. By high school, students should move beyond basic skills to more challenging outcomes in core subjects, in vocational training, and in the arts.

EQUITY AND EXCELLENCE ACROSS GENDER AND **ETHNICITY**

For the past several decades, "equity" and "excellence" have been guiding ideals in education. Schools have sought higher levels of excellence as demonstrated by better student performance. At the same time, they have sought to distribute that excellence more equitably across students of color, gender, and rich and poor students. Both equity and excellence are essential to achieving society's diversity goals. We now turn to a consideration of how equitably excellence has been achieved across

gender and ethnicity.

ACHIEVEMENT BY GENDER

Figures 5.8 through 5.10 compare the performance of girls and boys in third, fifth, and eighth grades respectively. These three figures graphically portray the data for girls and boys from Tables 5.7 through 5.13 (pp. 39-42). In reading, the direction of the difference is consistent across grades; girls outperformed boys at every grade

level. For instance, in third grade, 82% of girls and 73% of boys scored at or above Level II; 41% of girls and 30% of boys scored at or above Level III. In fifth grade, 83% of girls and 76% of boys scored at or above Level II; 43% of girls and 34% of boys scored at or above Level III. In eighth grade, 71% of girls and 66% of boys met the high school minimum standard in reading. These differences are attributable, in part, to the higher propor-

Figure 5.8 Grade 3 Percentage of Students at or above Level II and Level III for Reading and Mathematics, by Gender

Figure 5.9 Grade 5: Percentage of Students at or above Level II and Level III for Reading, Mathematics, and Writing, by Gender

tion of boys in special education. As seen in Tables 5.7 through 5.13 (pp. 39–42), roughly twice as many boys as girls are in special education and classified as having a disability.

Results are less consistent across grades in mathematics. In third grade, 82% of both boys and girls scored at or above Level II; 34% of girls and 36% of boys scored at or above Level III. In fifth grade, essentially the same percentage of girls as boys scored at or above Level II (80% for girls vs. 79% for boys) and Level III (30% for girls vs. 32% for boys). In eighth grade, more boys met the high school minimum standard in mathematics: 73% versus 70%. Gender differences in mathematics tend to be smaller than in other subject areas on the statewide tests.

The largest gender difference is in fifth grade writing where 87% of girls and 74% of boys scored at or above Level II; 52% of girls and 32% of boys displayed the solid mastery of challenging material characteristic of Level III. Again, some of this difference is attributable to the higher percentage of boys with a disability and placed in

special education.

ACHIEVEMENT BY ETHNIC GROUP

Figures 5.11 through 5.13 (p. 44) show the statewide achievement test scores by ethnic groups in third, fifth, and eighth grades respectively. They graphically portray ethnic data from Tables 5.7 through 5.13. These ethnic differences are generally larger and more consistent across subject matter than the gender differences of Figures 5.8 through 5.10. In interpreting these differences, one must remember that minority test-takers include: (a) a high percentage of economically disadvantaged students eligible for free and reduced lunch in all minority ethnic groups, (b) a large percentage of students with limited English proficiency among Asian and Hispanic students, and (c) a large percentage of students in special education in all non-Asian minority groups. These percentages are shown in Tables 5.7 through 5.13.

Across all grades and subject matters, the minority ethnic groups scored lower than the state as a whole. Black students exhibited the lowest performance, but also the highest rates of poverty and placement in special education. Asian students outperformed the other ethnic minority groups at fifth and eighth grades. In third grade, however, the scores of Asian students are more comparable to those of Hispanic and American Indian students.

(cont'd on p. 45)

Figure 5.10 Grade 8: Percentage of Students Meeting High School Minimum Standard for Reading and Mathematics, by Gender

Grade 3: Percentage of Students at or above Level II and Level III for Reading and Mathematics, by Ethnicity

Grade 5: Percentage of Students at or above Level II and Level III for Reading Mathematics, and Writing, by Ethnicity

Ethnic Group

Figure 5.13

Grade 8: Percentage of Students Meeting Basic Standard for Reading and Mathematics, by Ethnicity

Any approach to higher achievement must encompass all students. The issues discussed below apply to all students, but they tend to be more prevalent among minority and economically disadvantaged students. Achieving equity of outcomes among Minnesota's ethnic groups poses the difficult challenge of achieving equity among groups with very different rates of disabilities, English proficiency, and poverty. A program to address such differences goes beyond the scope of this report; nevertheless, these data point to factors that any such program should consider. For the sake of brevity, we will limit this discussion to trends in our data, even though this leads to a largely incomplete discussion of differences among groups. Some related issues concerning attendance and course-taking in high school were addressed in Chapter Four.

First, large differences appear by third grade, the earliest grade for which statewide achievement data are collected. Any solution will need to encompass early childhood development through programs at the preschool and early elementary levels. Early childhood development must mesh seamlessly with the primary grades by laying the social and cogni-

Figure 5.14 Grade 3: Percentage of Students (with and without LEP) at or above Level II for Reading and Mathematics, by Ethnicity

tive foundations for later school learning.

The second factor is highlighted by Figures 5.14 through 5.16 (pp. 45– 46). The figures show data on Asian and Hispanic students when those with limited English proficiency are and are not included. Among Asian students, those with limited English proficiency constitute from 36% (eighth grade) to 62% (third grade) of the students tested. Among Asian students, the differences with statewide data disappear at third and fifth grade in all subject areas when only English proficient students are considered, and the differences between Asian and statewide results become small at eighth grade. Similar trends are seen in the data of Hispanic students, although the trends are not as strong. All of this highlights the role of English proficiency and English language learning in addressing the problems faced by many Asian and Hispanic students.

At first, it may seem that language development is not critical to understanding the performance of American Indian and Black students, few of whom are classified as having limited English proficiency. However, Asian and Hispanic students' achievement is closely tied to their level of English proficiency, which highlights the importance of language development to achievement. Researchers such as Hart and Risley (1995) have emphasized the importance of early language stimulation for economically disadvantaged students and the relationship of that stimulation to later school performance.

Finally, at all grade levels and subject matters in Tables 5.7 through 5.13, a higher percentage of American Indian,

Figure 5.15 Grade 5: Percentage of Students (with and without LEP) at or above Level II for Reading, Mathematics, and Writing, by Ethnicity

Figure 5.16 Grade 8: Percentage of Asian and Hispanic Students (with and without LEP) Meeting Basic Standard for Reading and Mathematics

Ethnic Group

Charter Schools

Black, and Hispanic students are new to their districts. Such mobility has long been associated with a disrupted educational experience and low achievement. This problem can be addressed only through combining efforts to reduce student moves, particularly midyear changes of schools or districts, with adjustments in school programs that better accommodate such mobility.

In a document like this, we would not presume to make anything more than a cursory attempt at discussing solutions for the differences in achievement across ethnic groups. In the end, however, it simply comes down to the fact that the large ethnic and racial differences in reading, writing, and math across all grades are simply incompatible with society's diversity goals.

CHARTER AND NON-PUBLIC SCHOOLS AND SCHOOL CHOICE

Increasingly, Minnesota offers students educational choices through open enrollment, post-secondary options, private schools, home schooling, and charter schools. Tables 5.7 through 5.13 (pp. 39–42) report statewide achievement test data on public charter schools and non-public schools. Minnesota passed the nation's first charter school legislation in 1991. Charter schools are legally and financially independent public schools. They constitute an exciting and rapidly expanding segment of Minnesota's public schools. Preliminary estimates from the Department of Children, Families, and Learning (1998a) show that enrollment grew from 2390 to 3708 between 1997 and 1998, a 55% increase.

Tables 5.7 to 5.13 show test results for Minnesota charter schools. The achievement levels of charter schools vary widely. Overall, however, performance of students in charter schools lags behind that of other public schools outside the urban areas of Minneapolis and St. Paul. For instance, Table 5.12 shows that 43% of charter students met the state's minimum high school graduation standard in reading, as compared to 68% for the state as a whole. Table 5.13 shows that 40% met the state's minimum high school graduation standard in mathematics, as compared to 71% for the state as a whole. At least in part, these results can be attributed to student characteristics. More than half of eighth grade charter school students (51%) are economically disadvantaged

students, eligible for free or reduced lunch, as compared to 24% statewide. And more than twice as many are students with disabilities (27% as compared to 12% statewide). It is also important to note that Minnesota charter schools often receive less funding per pupil unit than do traditional public schools (Mandala, 1998). Charter schools do not receive any excess levies that regular districts can access through voter approval.

Nevertheless, the achievement levels in charter schools pose a challenge to parents and to the agencies which charter those schools. School choice places a heavy responsibility on parents and students for the selection of schools and programs that are best suited to the abilities and goals of the student. All parents and students need to ask questions and carefully read materials about any school, public or private, before making a selection. Parents and students should also consult the Graduation Standards button on the web site of the Department of Children, Families, & Learning for information about the school's achievement levels and other contextual factors. The web site's address is http://cfl.state.mn.us . Some

> . . . achievement levels in charter schools pose a challenge to parents and to the agencies which charter those schools.

charter schools serve large numbers of at-risk students, and therefore may have programs best suited to such students.

The second challenge is to sponsoring agencies. These agencies are charged with the responsibility of reviewing the charter school's performance at least every three years as part of the charter renewal process. The charter school legislation (Minnesota

Statute 120.064, 1997) makes clear that the review is to be results-oriented, and advises, "A charter school must design its program to at least meet the outcomes adopted by the state Board of Education for public school students." Charter schools must adequately prepare their students for both the Graduation Rule's Preparatory and High Standards. In charter schools that do not extend through the twelfth grade, the sponsoring agency must consider how students transferring to other public and private schools for high school graduation will meet the required standards.

Non-Public Schools

Minnesota has 514 non-public schools; most of these serve relatively small numbers of children compared to their public school counterparts. Consequently, while more than 20% of the schools in Minnesota are nonpublic schools, non-public schools enroll only 10% of our state's student population.

All public schools are required to participate in statewide testing, and must test all students, with certain exceptions for students with disabilities and those with limited English proficiency. In contrast, non-public schools participate in testing on a voluntary basis. Furthermore, given the scarcity of available data on nonpublic schools' student demographics, there is currently no way to determine whether the test takers are representative of non-public school students in terms of family income and other factors associated with achievement. Nor can we determine whether differences in public and non-public student performance are related to differences in student characteristics beyond the control of schools, such as poverty, limited English proficiency, disabilities, etc.

Comparisons between public and non-public school performance are questionable at best, and therefore some earlier reports have omitted the non-public school data. We have included the data in Tables 5.7 through 5.13. Conclusions based on comparisons between public and nonpublic schools are very tentative.

At third and fifth grade, the scores of non-public school students tend to be roughly comparable to those of public school students in the suburban schools, which in turn, tend to be higher than those for public school students as a whole. There are exceptions, notably in the fifth grade writing data, where the non-public scores are more comparable to those of public schools generally, and lower than the scores of suburban public schools. At

> While ACT college admissions scores for Minnesota declined through the 1980s, they have risen through the 1990s. Of the states with 50% or more students taking the ACT, only one state had an overall average scale score higher than Minnesota in the 1998 ACT administration.

eighth grade, the scores of the nonpublic school students are higher than the scores of public school students, including suburban students, for both mathematics and reading.

THE PERFORMANCE OF MINNESOTA STUDENTS IN COLLEGE ADMISSIONS TESTING

Tables 5.7 through 5.13 show data on all students for grades three, five, and eight. But what about Minnesota's college-bound students as they near the end of high school? Of the two college admissions tests, the ACT Assessment (ACT) and the Scholastic Assessment Test (SAT; formerly known as the Scholastic Aptitude Test), far more Minnesota high school seniors and juniors take the former. Therefore, ACT test results more completely reflect the performance levels of Minnesota students bound for

60

two- and four-year colleges.

The decline in college admissions test scores through the 1970s and into the 1980s has been of national concern. In 1988, the Minnesota Legislative Auditor's Office reported that "Contrary to the national trend, Minnesota's ACT scores have continued to decline. Minnesota's composite ACT score in 1988 dropped to its lowest point in 21 years" (p. 15).

While ACT college admissions scores for Minnesota declined through the 1980s, they have risen through the 1990s. Figure 5.17 shows the trend since the last year of the auditor's report. Scores in Minnesota continued to decline for one more year. But since 1989-90, scores have increased slowly. This upward trend cannot be attributed to a declining number of test takers, since the number of Minnesota students taking the test rose steadily throughout the period. In the 1997-98 administration of the test, the national average score was 21.0, while state ACT averages ranged from 18.7 (in Mississippi) to 22.3 (in Wisconsin) for states in which at least 50% of graduates took the test. Of these states, only Wisconsin had a higher mean than Minnesota (22.3 vs. 22.2).

In 1997-98, 37,100 students in Minnesota took the ACT, 20,628 females and 16,472 males. These figures clearly attest to the increasing participation of women in undergraduate higher education, an increase that has been noted elsewhere, most recently in *The 1997 Minnesota High School Follow-up Study* (Minnesota Department of Children, Families, & Learning, 1998c). The figures reflect success at increasing the participation of women in higher education.

Figure 5.18 displays Minnesota student performance on the ACT for the 1997–98 year, with data broken down by gender and ethnicity. The general trends seem strongly related to

Figure 5.17 Minnesota and National ACT Composite Scores, 1988–98

the percentage of test takers from the various ethnic groups identified in Chapter Four as having taken ACT's recommended core course requirements. That is, those groups with higher percentages of students taking the recommended core course requirements generally have higher average ACT scores than those groups in which lower percentages of students complete the recommended coursework.

With the emphasis on minimum competency produced by the *Basic Standards Tests*, there has been concern that the education of high ability students may be neglected. The rising ACT scores shown in Figure 5.17 and the increasingly higher course preparation of college-bound students discussed in Chapter 4 should help to relieve this concern.

School Year/Number of Students Tested (N)

Figure 5.18 1997–98 Minnesota ACT Composite Scores, by Gender and Ethnicity

Subgroup

In Minnesota's efforts at educational reform, there have been three key goals: greater choice, equity, and excellence. Our findings and conclusions are organized around these three issues.

EDUCATIONAL CHOICES

Since the 1980s, Minnesota has tried to widen the range of alternatives available to students. Students are no longer limited to a school in the district where they reside and can attend a public school of their choice under the state's open enrollment policy. High school students can also choose to attend a post-secondary institution under the Post-secondary Enrollment Options program. The 1991 charter school legislation opened a new type of public school operating under less restrictive regulations than other public schools, and therefore with more freedom for innovation. Tax deductions for tuition lighten the financial cost, at least somewhat, and therefore widen access to the private school system. These options not only enhance choice, they also increase the responsibility of students and parents to choose wisely among schools as well as among programs within schools. Consequently, school choice cannot be separated from parental involvement.

The data on charter schools covered in this initial report lead us to conclude that the low achievement scores, low attendance rates in secon-dary grades, and low graduation rates of some charter schools will pose a challenge to their sponsoring agencies. These agencies are charged with the responsibility of reviewing the charter school's performance at least every three years as part of the charter renewal process. The charter school legislation makes clear that the review is to be results-oriented, and advises, "A charter school must design its program to at least meet the outcomes adopted by the state Board of Education for public school students" (Minnesota Statute 120.064, Subdivision 10 [1997]).

Although charter schools vary in their missions, the composition of their student bodies, and their outcomes, their collective performance was generally below that of other public schools. Table 4.3 shows lower attendance in charter schools at eighth and twelfth grade. Although the

MINNESOTA'S PROFILE OF LEARNING PERFORMANCE PACKAGES: A SNAPSHOT FROM BRAINERD, MINNESOTA

Students use trigonometric equations, spreadsheets, and graphs to demonstrate their ability to relate real-world situations and mathematical models. Don Karlgaard, a high school math teacher in Brainerd, facilitates this process by using Biorhythms, a performance task he authored to meet a portion of the Algebraic Patterns content standard.

Students prepare for this performance task by studying mathematical functions and trigonometry. Mr. Karlgaard and his students refer to the Biorhythms performance assessment checklist throughout their study of trigonometry to clarify content and performance expectations. In addition, Mr. Karlgaard instructs students in the use of spreadsheets and graphing calculators.

Biorhythm Overview

At the end of the 19th century, a German

physician named Wilhelm Fleiss tried to establish a mathematical relationship between an individual's date of birth and the date of death and/or illnesses. Since then, others have expanded and adjusted the original idea, and the concept is now known as "biorhythm."

According to this idea, there are three biological patterns that begin at birth and affect disposition during the course of a lifetime. These patterns do not determine what will happen on a particular day, only how the individual is likely to feel as events occur. There is a physical cycle with a 23-day period, an emotional cycle with a 28-day period, and an intellectual cycle with a 33-day period.

On your date of birth, each pattern starts at zero and begins to rise in a positive phase, during which the energies and abilities associated with the cycle are high. Then, gradually declining, the patterns cross the zero point midway through their complete periods, and continue into a negative phase in which capabilities are low. Then, increasing amounts of energy are picked up as the negative phase turns upward until, at the end of the patterns, the zero point is recrossed into the positive phase, and the cycle begins again. (Original description credited to Robert Andre, Waianae, Hawaii, in *Minnesota Mathematics Magazine*, Winter, 1995.)

After this brief overview, students are assigned research on biorhythms, specifically, how to analyze and interpret results. When the assignment is completed, Mr. Karlgaard leads the class in a discussion of the best methods and procedures used for data analysis and interpretation. After students have had time to process this information, Mr. Karlgaard assigns the Biorhythm performance task.

number of students on which the dropout figures are based is small, Table 4.4 shows low completion and high dropout rates for these schools. The achievement levels shown in Chapter 5 are generally below those of other public schools.

To some extent, these outcome figures for charter schools are understandable in light of the large percentage of students new to the schools, the high percentage of students of poverty (eligible for free and reduced lunch), and the high percentage of students with disabilities as seen in Tables 5.7 -5.13.

Nevertheless, in the charter renewal process, chartering agencies should carefully review outcome data for each school on a case-by-case basis. The review should by no means be limited to the kinds of input, process, and outcome information covered by this report. For charter schools whose programs terminate prior to twelfth grade, charter school students will need to transfer prior to completion of high school. The charter renewal process should consider any needs created by such transfers.

EDUCATIONAL EQUITY

Equity can be defined in various ways. It can be defined as equality of opportunity or equality of outcome. Equity can be defined in terms of gender, ethnicity, economic status, or disability status. In this section, we focus on gender and ethnicity.

Gender

On the outcomes covered in this report, differences by gender do not consistently favor boys or girls. For the most part, the third, fifth, and eighth grade boys had achievement levels at least as high as girls on the mathematics tests covered in this report. In our opinion, these differences in mathematics achievement are small.

In third, fifth, and eighth grades,

achievement differences on the reading tests consistently favor girls. In our opinion, these differences are also small. The largest difference is on the fifth grade writing test, and by virtue of its size, is the male/female difference of most concern. Girls graduate from high school at higher rates than boys. The most recent high school follow-up survey suggests that, in contrast to historical trends, girls also progress to two- and four-year colleges at rates comparable to boys. Indeed, this latter result is a major success attributable, in part, to past decades of gender equity initiatives in schools.

The large differences among various ethnic groups in reading, writing, and mathematics across all grades are incompatible with society's diverse goals and with our state's drive toward higher standards for all students. Any approach toward addressing the inequity of achievement among Minnesota's minority students must take into account various factors: lower attendance, higher rates of poverty, greater mobility, and higher proportions of students with disabilities, and limited English proficiency.

ETHNICITY

Ethnic diversity has been a major theme of our society and of our schools for decades. As a percentage of total enrollment, minority children will continue to increase. The breadth and scope of ethnic differences in graduation rates and achievement is simply intolerable. So long as such pervasive differences exist, society is unlikely to achieve its diversity goals. But responsibility for the differences cannot be laid solely at the feet of schools. It will take a genuine familyschool-community partnership to address the problems faced by minority children and their schools. And if it is

a true partnership, each partner—the schools, families, and the community—must accept and be willing to be accountable for its share of the responsibility.

Minority students more commonly live in conditions of poverty. Many of them have limited English proficiency that must be addressed through English language instruction. Tables 5.7 - 5.13 show that minority students taking the achievement tests are far more likely to be new to their district than are Minnesota students generally. We must attempt to reduce the mobility of minority students and minimize the educational disruptions caused by such mobility. Schools need the assistance of parents in reducing mobility, particularly midyear transfers between schools and districts.

In terms of participation outcomes, attendance rates for minority students are lower than for the state as a whole, particularly in the secondary grades. The attendance rates of Native American students are particularly alarming, because they are low in the elementary grades and fall off more sharply than that of other groups in the secondary grades. Attendance is another area where families and communities must assist the schools.

Minority students also experience higher dropout rates and lower fouryear high school graduation rates. Census data suggest that many black Minnesotans may be acquiring a high school diploma outside the traditional four-year time span or acquiring a high school equivalency degree after dropping out. Based on these and other findings, it would seem that we need to track some students beyond four years, and bring the adult basic education programs into the accountability and public reporting system to better understand the attainment of a high school education by some students.

Among college-bound students who

took the ACT college admissions test, fewer minority students had taken the core academic course preparation: four years of English, three of mathematics, three of science, and three of social studies. Parents, teachers, and counselors need to encourage post-secondary education, but also adequate high school preparation consistent with the student's post-secondary plans.

Ethnic differences are equally pervasive in the achievement data. They appear in the youngest grade covered by this report, third grade. They extend through the fifth and eight grades, through all subject areas tested, and into the ACT college admissions data. For students with limited English proficiency, English

language instruction will be a key. For eight. others, early educational interventions will be critical. But there is no simple solution. As we turn to a consideration of excellence, it is clear that excellence, at least as measured by the indicators of this report, is by no means equitably distributed across ethnic groups in Minnesota's society. These inequalities pose a major societal challenge.

EXCELLENCE

In the report above, we have treated other states and nations as benchmarks against which to judge the performance of Minnesota students. Good data are available only in reading, mathematics, and science. Even in these areas, data are available only for selected grades, and none above grade

The International Reading Literacy Study found that only one country had a mean score significantly greater than that of the United States (U.S. Department of Education, 1996). This finding held at both fourth and eighth grade. While this study contained no direct comparison of Minnesota students to those of other countries, the National Assessment of Educational Progress (Campbell, Donahue, Reese, & Phillips, 1996) has reported that Minnesota fourth graders scored well above the US average in reading, and no state had a mean significantly above that of Minnesota. Combining the national and international data, it seems most

MATH APPLICATIONS (cont'd from p. 49)

Students are required to:

• Develop a mathematical model to illustrate and investigate Fliess' idea, using their birth dates.

 Develop trigonometric equations to model all three patterns.

 Create a spreadsheet to calculate values for graphing the cycles for the last month and the next two months.

· Use the graph to communicate relationships between the cycles and their recollection of good and bad experiences last month.

• Use the graphs to predict high and low times for the upcoming two months.

Mr. Karlgaard uses this task in place of his regular trigonometry test and gives students approximately one week to complete the assignment. The High Standards, explains Mr. Karlgaard, provide students with the "big picture," an opportunity to apply what they have learned at substantially different levels than what is required of them on a regular paper and pencil test. Mr Karlgaard believes that students understand and learn more when they see and make connections between mathematics and real-life situations.

The following is an excerpt from a student's Biorhythm project calculations.

FIND THE NUMBER OF DAYS OLD YOU WERE AT THE END OF 1997:

Birthday: December 6, 1981

TOTAL	DAYSA	LIVE:		CHECK:	
1981	25	1989	365	365	
1982	365	1990	365	x 16	age
1983	365	1991	365	5840	
1984	366	1992	366		
1985	365	1993	365	5840	
1986	365	1994	365	+ 4	leap years in 16 yrs.
1987	365	1995	365	5844	
1988	366	1996	366		
		1997	365	5844	
			1337 000	+ 25	days older than 16 yrs.
		TOTAL: 5869		5869	TOTAL

(cont'd on p. 52)

reasonable to conclude that the reading direct comparison of Minnesota of Minnesota's fourth-grade students is competitive with that of even the best nations around the world and the highest scoring states within the United States.

Thanks to Minnesota's participation in SciMath^{MN} (1997–98), we have a

MATH APPLICATIONS (cont'd from p. 51)

Students used the days in each biological cycle and their total number of days alive to develop the equations for their biological patterns. Each student has a unique Biorhythm graph.

The following excerpt was taken from a student's interpretation of the Biorhythm graph (below).

Prediction

Using the Biorhythm graph, I will be able to determine a positive or negative relationship between the biological patterns and the good or bad days that I have had.

Physical

At the beginning of January, I remember being very tired. In fact, I could not keep my eyes open in class. What I found interesting was that the Biorhythm graph also indicated a low physical cycle around this time. Toward the beginning of February, I should be full of energy according to my physical biorhythm graph. I can also predict that around March 15th I

students with those of other countries in mathematics and science. At both fourth and eighth grade, six other countries had mean scores significantly above the mean for Minnesota students. While the mathematics achievement of Minnesota fourth and eighth

graders has compared very favorably to that of other states in the National Assessment of Educational Progress, the achievement levels are less impressive when benchmarked against the performance of students in other countries. A re-analysis of our mathematics curriculum seems warranted,

will have a very high physical energy level.

Emotional

During the beginning of January my Biorhythm graph indicated that I would feel emotionally low. This, however, was very inaccurate. My emotions were very high during this part of the month; I could not have been happier. Looking ahead, I should expect to reach emotional highs around the 20th of February and March. I should have emotional lows around February 8th and March 6th.

Intellectual

My intellectual biorhythm seems to fit how I was feeling in January. I did not feel very intellectually "with it." I should reach a higher intellectual level around February 5th and March 8th. And, I should have an intellectual low point around the 22nd of February and March.

Evaluation

Students were evaluated on a 3-point scale:

E - excellent, S - satisfactory, or N - needs improvement for each criterion on the biorhythm checklist. Since the completion of this project, the scale for evaluating tasks has been changed to a 2-point scale: Y (yes) has met the performance task or N (no) - has not met the performance task. The performance criteria are listed below:

 Mathematical equations: incorporate correct calculations.

• Trigonometric equations: are appropriate translations of biorhythmic patterns.

· Spreadsheet and graph: show evidence of appropriate mathematical procedures.

 Communication: clearly shows connection of the mathematics of the task to your life.

Once the student has completed all the performance tasks required in the content standard (Math Applications) they will be evaluated on the four-point scale developed by the Department of Children, Families and Learning.

BEST COPY AVAILABLE

particularly in seventh and eighth grades.

In science, as in reading, achievement levels seem high. At fourth grade, only Korea had a mean score significantly above that of Minnesota. At eighth grade, only Singapore's students had a mean score significantly . Reports that American students' higher than that of Minnesota students. In the last National Assessment of Educational Progress, two states had scores significantly higher than Minnesota, Maine and North Dakota. In fourth and eighth grade science, Minnesota student achievement seems to rank near that of the best countries in the world and close to that of the top performing states within the United States.

Given that fifteen years have now elapsed since A Nation at Risk, where do we now stand on three of the issues fueling the call for educational reform, the three most directly addressed by this report?

• National reports of declines in educational achievement

Clearly the decline in college admissions test scores on the ACT, the most frequently administered college admissions test, has been arrested. Scores have been climbing during the 1990s, even as more students are taking the exams. More college bound students are completing a core academic curriculum. On the National Assessment of Educational Progress (NAEP), mathematics scores have increased over the three administrations during the 1990s. Reading scores declined between 1992 and 1994, but this decline involves only two data points early in the decade. Given the increased emphasis on reading in the Graduation Standards,

we should forego conclusions until the 1998 NAEP results are in. Further evidence concerning declines or increases in achievement will come from comparing future performance on the MCA and BST against this year's baselines.

mathematics and science achievement test scores are lower than those of students from other industrialized countries

Sadly, data on Minnesota students (as distinct from American students generally) are lacking above eighth grade. In fourth and eighth grade science, only one country at each grade clearly scored above Minnesota students. We conclude that the achievement of Minnesota students is competitive with all but one participating country. In mathematics, six countries clearly performed better than Minnesota students, while about a dozen were clearly lower. This performance is not up to Minnesota's high ambitions in mathematics. Like SciMath^{MN}, we call for continued reform of curriculum and instructional methods in this area. A good place to start is with the evaluations of the U.S. curriculum arising from the SciMath^{MN} (1997–98) study.

• Educational equity for female, minority, and disadvantaged students

On the array of outcomes examined in this report, there is no consistent pattern of lower performance for males or females. Female performance was lower on some outcomes, such as the eighth grade Basic Standards Test in mathematics, but such lower performances by females were counterbalanced by higher scores on other

indicators, such as the Basic Standards Test in reading. The increased proportion of females attending two- and four-year undergraduate programs is one of the decade's educational success stories.

Large differences in achievement, high school course-taking patterns, and graduation rates remain among Minnesota's ethnic and racial groups. These differences are incompatible with our shared vision of a diverse society. The responsibility, however, cannot be laid solely at the feet of schools. Addressing the problem will require a concerted effort by schools, students, parents, and the wider community.

Looking back over the past two decades of educational reform, one can see clear progress in ACT scores, in the course preparation of college bound students, and in the undergraduate higher education participation of females.

CONCLUSION

From the available data, we conclude that Minnesota students achieve at high levels in reading and science when benchmarked against other states or other countries. In mathematics, they perform at high levels when benchmarked against other states, but at more modest levels as compared to other countries.

Within this overall high level of performance, however, not all groups of students are scoring well, or graduating at acceptable rates. Statewide test scores leave room for improvement at all grades tested. This year's achievement scores provide the baseline from which that improvement will be gauged.

BEST COPY AVAILABLE

APPENDIX A

MINNESOTA COMPREHENSIVE ASSESSMENTS AND BASIC STANDARDS TESTS WITH RESULTS BROKEN OUT BY CATEGORY

								0/17 1	A/ NT	
Table A.1	1998 Grade 3: Minnesota Comprehensive Assessment Results in Reading for all Public	Category	Number Tested	% At or Above Level III	% At or Above Level II	Mean Scale Score	% Enrolled Students Tested	% Special Ed Students in Score	% New to District Since 1/1/97	% F/R Students in Score
	those with Limited English	Total	57277	- 37	80	1419	94	11	10	29
	Proficiency	Girls	28277	42	84	1444	94	8	10	30
		Boys	28939	32	76	1396	93	15	10	29
		Asian	1096	36	80	1424	95	5	16	40
		Black	3538	11	46	1266	91	15	16	82
		Historic	921	24	71	1363	91	11	18	58
		American Indian	1200	15	56	1303	89	18	17	75
		White	50410	40	83	1434	94	11	9	24
		Special Ed	6534	13	42	1251	82	_	10	43
		Merro Area	30066	20	80	1425	94	10	10	26
		Outstate	27040	34	80	1413	94	12	11	36
		Mols/St Paul	5879	23	59	1330	91	11	11	63
		TC Subarbo	24187	43	85	1448	94	10	9	17
			12000	25	90 90	1/1/	03	13	Ŕ	20
		Outstate: 2000+ Outstate: 2000-	13818	34	80	1413	94	12	11	36 36
		PublidCharter	229	26	57	1362	82	11	63	61
		Public/Not Charter	57048	37	80	1420	94	11	10	29

Table A.2 1998 Grade 3: Minnesota Comprehensive Assessment Results in Reading for all Public School Students Tested Except those New to Their District Since January 1, 1997

Caregory	Nırırber Tested	Inntar %Ator Above Issed Level III		MtanScale Score	%Encled Sackers Tested	%LEP Sudensin Sare	%Special Edi Saudentsin Saure	%F/R Sadensin Sare
Tad	53996	36	78	1415	94	- 4	- 11	
Gats	26635	41	83	1440	95	4	7	30
Boys	27305	31	74	1391	93	4	15	30
Asian	2488	16	52	1304	93	63	6	70
Black	3081	11	46	1263	91	4	15	82
Hispanic	1212	17	56	1306	88	38	11	68
American Indan	998	16	56	1306	89	0+	17	74
White	46176	40	83	1436	95	0+	11	2
Special Ed	6045	13	42	1251	83	3	_	42
LEP	266	5	35	1227	89	-	7	87
Meno Ana	29081	38	77	1416	94	7	10	28
Oleer	24821	34	80	1413	94	2	12	32
MpHs/St. Paul	6709	19	52	1303	91	23	10	θ
TCSituts	22262	43	85	1450	95	2	10	15
Quesar: 2000+	12434	34	79	1412	94	2	12	30
Oasze: 2000-	12387	34	80	1414	95	1	12	35
Ritio/Chater	111	15	41	1259	79	23	19	66
Ritto/Nr Chater	53885	36	78	1415	94	4	11	30

<i>1998 Grade 3:</i> Minnesota Comprehensive Assessment <i>Results in</i> <i>Reading for all Public</i>	Category	Number Tested	% At or Above Level III	% At or Above Level II	Mean Scale Score	% of Enrolled Students Tested	%LEP Students in Score	% New to District Since 1/1/97	% F/R Students in Score
School Students Tested	Total	53193	38	82	1431	95	5	10	
Except those in Special Education	Girls Bars	27318	43 24	95 70	1450	95 95	4	10 10	31 20
	DOys	2,019	Л	19	1412	り	J	10	50
	Asian	2698	17	53	1308	94	62	13	70
	Black	3159	12	51	1287	93	5	17	81
	Hispanic	1331	17	59	1317	88	39	17	70
	American Indian	1165	15	58	1329	90	0+	17	74
	White	44925	43	87	1455	96	0+	9	22
	LEP	2450	5	35	1229	87	_	14	87
	Metro Area	28957	39	81	1430	95	7	10	29
	Outstate	24088	37	84	1433	95	2	10	32
	Mpls/St. Paul	6871	20	55	1316	93	24	11	69
	TCSuburbs	22086	45	88	1466	95	2	8	16
	Outstate: 2000+	11877	37	84	1432	95	3	8	30
	Outstate: 2000 -	12211	37	84	1433	96	1	11	34
	Public/Charter	234	24	56	1327	85	12	62	64
	Public/Not Charter	52959	38	82	1432	95	5	10	30
	Non-public	1315	43	88	1455	_	_	_	_

Table A.41998 Grade 3: Minnesota
Comprehensive
Assessment Results in
Mathematics for all Public
School Students Tested
Except those with Limited
English Proficiency

Table A.3

Category	Number Tested	%At or Above Level III	%At or Above Level II	Mean Scale Score	Enrolled Students Tested	% Special Ed Students in Score	%New to District Since 1/1/97	%F/R Students in Score
Total	57261	37	84	1412	94	12	10	29
Girls	28177	36	84	1408	94	8	10	30
Boys	29012	38	84	1417	93	15	10	29
Asian	1086	37	84	1423	94	5	16	40
Black	3513	8	48	1202	91	15	16	82
Hispanic	926	21	73	1325	90	11	18	58
American Indian	1188	16	67	1292	88	19	17	75
White	50342	40	87	1432	94	11	9	24
Special Ed	6576	15	56	1250	83	_	10	43
Metro Area	30041	39	84	1420	94	11	10	26
Outstate	27045	34	85	1404	94	13	10	33
Mols/St. Paul	5862	24	63	1303	91	11	11	63
TCSuburbs	24179	43	89	1448	94	10	9	17
Outstate: 2000+	13162	34	85	1403	93	13	8	30
Outstate: 2000 -	13883	34	85	1406	91	13	11	36
Public/Charter	256	22	61	1279	91	13	60	61
Public/Not Charter	57005	37	84	1413	94	12	9	29

Table A.5	1998 Grade 3: Minnesota Comprehensive Assessment Results in Mathematics for all Buttics School Students Testad	Caregry	Namber Tesed	%Ator Above Level III	%Ator Above Level II	ManSale Same	%Enroled Suctors Tesed	%LEP Sudensin Sane	%Specal Ed Sudens in Some	%F/R Sudensin Score
	Except Those New to Their	Taal	53979	- 36		1408	94	- 4	11	
	District Since	Girls	2659	35	83	1404	94	4	7	30
	January 1, 1997	Boys	27344	37	84	1413	94	4	15	29
		Asian	2457	20	65	1303	92	ଔ	6	70
		Black	3063	9	48	1203	91	4	15	82
		Hispanic	1209	15	60	1264	88	37	11	68
		Arresian Indan	989	17	67	1293	89	0+	18	74
		White	46130	40	88	1435	95	0+	11	22
		Special Ed	6082	15	56	1250	84	3		42
		LEP	2253	7	49	1208	89	—	7	87
		MenoArea	29004	38	82	1412	94	7	10	7
		Ouszac	24831	34	85	1404	94	2	12	2
		Mfd/St. Paul	6746	20	60	1284	91	23	10	Ð
		TCSibuts	22258	44	89	1451	95	2	10	15
		Oussate 2000+	12382	33	84	1401	93	2	12	30
		Ousae 2000-	12449	34	85	1407	95	1	13	34
		Ritio/Chater	127	17	55	1244	91	20	21	20
		Rublio/Net Chanter	53852	36	83	1409	94	4	11	4

Table A.61998 Grade 3:MinnesotaComprehensive AssessmentResults in Mathematics for allPublic School Students TestedExcept Those in SpecialEducation

	Category	Number Tested	% At or Above Level III	%At or Above Level II	Mean Scale Score	% Enrolled Students Tested	% LEP Students in Score	% New to District Since 1/1/97	% F/R Students in Score
đ	Total	53123	38	- 86	1423	95	- 5	10	30
	Girls	27232	36	85	1414	95	4	10	31
	Boys	25831	40	87	1433	95	5	10	29
	Asian	2670	20	66	1308	93	62	13	70
	Black	3149	9	52	1220	92	5	17	81
	Hispanic	1322	15	62	1274	87	38	19	69
	American Indian	974	18	72	1317	90	0+	17	74
	White	44886	42	91	1451	96	0+	9	22
	IEP	2438	7	50	1211	87		14	87
	Metro Area	28912	40	84	1424	95	7	10	29
	Outstate	24063	36	88	1422	95	2	10	32
	Mpls/St. Paul	6833	21	63	1296	92	24	11	69
	TC Suburbs	22079	45	91	1463	95	2	10	16
	Outstate: 2000+	11829	36	87	1418	94	3	8	29
	Outstate: 2000-	12234	37	89	1425	96	1	11	34
	Public/Charter	252	21	61	1285	91	11	60	60
	Public/Not Charter	52871	38	86	1424	95	5	10	30

~										
Table A.7	<i>1998 Grade 5:</i> Minnesota Comprehensive Assessment <i>Results in Reading for all</i> <i>Public School Students</i>	Category	Number Tested	% At or Above Level III	%At or Above Level II	Mean Scale Score	Enrolled Students Tested	% Special Ed Students in Score	% New to District Since 1/1/97	%F/R Students in Score
	Tested Except Those with Limited English Proficiency	Total	57672	40	81	1429	95	13	9	27
		Girls	28234	44	85	1453	96	8	9	27
		Boys	29417	35	78	1405	95	18	9	27
		Asian	1340	41	84	1445	97	6	12	.38
		Black	3141	14	47	1258	94	20	16	77
		Hispanic	861	23	68	1350	93	17	17	55
		American Indian	1162	15	58	1293	90	21	13	73
		White	50959	42	84	1443	96	13	8	22
		Special Ed	7535	12	42	1235	89		9	41
		Metro Area	29443	42	81	1438	96	13	9	24
		Outstate	27952	37	81	1419	95	14	8	31
		Mpls/St. Paul	5422	26	61	1333	94	16	11	61
		TCSuburbs	24021	46	86	1461	96	12	8	15
		Outstate: 2000+	13612	38	82	1425	95	14	8	27
		Outstate: 2000 -	14340	36	80	1414	96	14	9	34
		Public/Charter Public/Not Charter	188 57484	31 40	65 81	1357 1429	94 95	22 13	59 8	51 <i>2</i> 7

Table A.8 1998 Grade 5: Minnesota **Comprehensive Assessmer** Results in Reading for all **Public School Students** Tested Except Those New to Their District Since January 1, 1997

nt	Caregory	Narthe Tested	%Ator Above Level III	%Ator Above Level II	ManSale Soure	%of Findled Studens Tested	%LEP Sudens in Same	%Special Ed Students in Score	%F/R Sudensin Sare
0	Total	54605	39	80	1425	96	3	13	27
	Gats	26650	44	84	1450	96	3	8	27
	Boys	27936	35	77	1401	95	4	17	28
	Asian	2489	23	59	1327	97	52	9	65
	Black	2743	14	47	1257	94	4	20	77
	Hsparic	1080	18	56	1298	94	34	17	64
	American Indian	1012	16	58	1296	91	0+	21	71
	White	47149	43	84	1446	96	0+	B	21
	Special Ed	7073	13	42	1235	90	3	_	41
	LEP	1881	4	34	1206	96	_	13	89
	MerroAna	28413	41	80	1430	96	5	13	- 25
	Oustate	25959	3 7	81	1419	95	1	13	30
	Mphs/St. Paul	6122	22	55	1307	95	21	15	67
	TCSuburbs	22291	46	86	1464	96	1	12	14
	Questae: 2000+	12856	38	81	1423	95	2	14	27
	Oustae: 2000-	13108	36	81	1416	96	1	B	33
	Addis/Chaner	101	28	55	1308	94	23	32	43
	Ablio/Na Chanter	54504	3 9	80	1425	96	3	13	27

Table A.91998 Grade 5: Minnesota
Comprehensive Assessment
Results in Reading for all Public
School Students Tested Except
Those in Special Education

Category	Number Tested	%At or Above Level III	%At or Above Level II	Mean Scale Score	% of Enrolled Students Tested	% LEP Students in Score	% New to District Since 1/1/97	%F/R Students in Score
Total	52032	42	- 85	1449	- %	4	9	27
Girls	26729	46	87	1465	97	3	9	28
Boys	25279	39	83	1432	96	4	9	27
Asian Black Hispanic American Indian White	2555 2624 1080 922 44678	23 16 19 18 46	62 54 61 67 89	1337 1291 1319 1331 1470	97 95 91 93 97	51 5 34 0+ 0+	11 17 13 8	65 76 66 72 21
LEP	1895	5	37	1217	93		14	8
Metro Area	27331	44	84	1452	%	6	9	26
Outstate	24490	40	86	1445	%	1	8	29
Mpls/St. Paul	5817	24	61	1336	96	22	10	66
TC Suburbs	21514	49	90	1483	97	1	9	15
Outstate 2000+	12010	42	87	1449	96	2	7	26
Outstate 2000 -	12480	39	86	1441	96	1	9	32
Public/Charter	164	35	70	1385	94	10	58	52
Public/Not Charter	51868	42	85	1449	96	4	9	27

Table A. 101998 Grade 5: MinnesotaComprehensive AssessmentResults in Mathematics for allPublic School Students TestedExcept Those with LimitedEnglish Proficiency

Category	Number Tested	% At or Above Level III	% At or Above Level II	Mean Scale Score	% Enrolled Students Tested	% Special Ed Students in Score	% New to District Since 1/1/97	% F/R Students in Score
Total	57528	32	81	1403	95	13	9	27
Girls	28063	31	81	1400	%	9	9	27
Boys	29440	33	81	1406	95	17	9	27
Asian	1331	34	83	1416	%	7	12	39
Black	3104	7	42	1215	93	20	16	77
Hispanic	855	16	65	1314	92	17	17	54
American Indian	1154	10	55	1273	90	21	14	73
White	50879	35	85	1419	%	13	8	22
Special Ed	7529	11	48	1246	89		9	41
Metro Area	29280	36	82	1414	95	13	9	24
Outstate	27972	29	81	1 <i>3</i> 93	95	14	8	31
Mpls/St. Paul	5367	20	59	1306	93	16	10	61
TC Suburbs	23913	39	87	1438	95	12	8	15
Ourstate: 2000+	13586	30	82	1398	94	14	7	27
Ourstate: 2000 -	14386	28	80	1387	96	13	9	34
Public/Charter	189	21	65	1315	94	21	58	51
Public/Not Charter	57339	32	81	1404	95	13	8	<i>2</i> 7

<i>Table A.11</i>	<i>1998 Grade 5:</i> Minnesota Comprehensive Assessment <i>Results in</i> Mathematics for all Public	Gægny	Nørter Tesed	%Arar Above Level III	%Ator Above Level II	ManSale Some	%Enailed Sudens Tested	%LEP Sudensin Sace	VoSpecal Ed Sudensin Scare	%F/R Sudensin Sane
	School Students Tested	Tazi	54502	32	81	1401	95	3	13	27
Except Those New to Their	Gets	26509	31	81	1399	96	3	8	27	
	District Since January 1, 1997	Boys	27971	33	81	1403	95	4	17	27
		Asian	2486	19	64	1318	97	53	9	65
		Black	2712	7	42	1217	B	3	20	77
		I-Ispanic	1075	12	54	1270	B	34	17	64
		American Indan	999	57	57	1278	90	0+	20	72
		White	47099	35	85	1422	96	0+	13	21
		Special Ed	7070	11	49	1248	90	3	_	40
		LEP	1877	4	41	1211	96		13	89
		Mano Ana	28276	35	80	1409	95	5	13	25
		Orver	25993	29	81	1393	96	1	13	30
		Mfls/St Pari	6077	17	56	1289	94	21	15	67
		TCS.hubs	22199	40	87	1441	96	1	12	B
		Oassan: 2000+	12843	30	82	1397	95	2	14	27
		Outate: 2000-	13150	28	81	1390	96	1	ß	33
		Ritic/Caster	103	17	62	(140)	95	23	29	44
		Ritio/Na Chatar	5439)	32	81	236	95	3	B	27

Table A. 121998 Grade 5: Minnesota
Comprehensive
Assessment Results in
Mathematics for all Public
School Students Tested
Except Those in Special

Education

Category	Number Tested	%At ar Above Level III	%At ar Above Level II	Mean Scale Score	% Enralled Sudents Tested	% LEP Students in Score	% Newto District Since 1/1/97	% F/R Sudents in Score
Total	51887	34	85	1419	96	4	9	27
Girls	26546	33	84	1411	96	3	9	28
Boys	25308	36	86	1428	96	4	9	27
Asian	2544	20	65	1326	97	51	11	65
Black	2597	8	47	1238	94	5	17	76
Hispanic	1072	13	57	1285	91	34	17	65
American Indian	919	11	63	1299	92	0+	B	72
White	44589	38	89	1441	96	0+	8	20
LFP	1888	5	43	1218	93	<u> </u>	14	88
Metro Area	27167	37	84	1424	96	6	9	26
Outstate	24512	32	86	1414	96	1	8	29
Mpk/St. Paul	5769	19	60	1308	95	22	10	66
TCSuburbs	21398	42	90	1456	96	1	9	15
Oustate: 2000+	11981	33	86	1419	95	2	7	26
Oustate: 2000-	12531	31	85	1410	97	1	9	32
Public/Charter	168	23	67	1335	96	11	57	53
Athlio/Not Charter	51719	34	85	1420	96	4	8	27

Note: LEP=Limited English Proficiency; F/R=Eligible for free or reduced-price lunch; 0+ indicates a .4% or less. All percentages and Mean Scale Scores are rounded to the nearest whole number.



····						67	WT I	0/ 11		
Table A.13	1998 Grade 5: Minnesota Comprehensive Assessment Results in Writing for all	Category	Number Tested	% At or Above Level III	% At or Above Level II	Mean Scale Score	% Enrolled Students Tested	% Special Ed Students in Score	%New to District Since 1/1/97	% F/R Students in Score
	Public School Students Tested Except Those with	Total	57597	43	81	1404	95	13	9	27
	Limited English Proficiency	Girls	28191	53	88	1506	96	8	9	27
		Boys	29385	33	75	1307	95	17	9	27
		Asian	1336	50	87	1479	97	7	12	38
		Black	3064	21	57	1136	91	20	16	76
	Hispanic	858	31	71	1285	92	17	16	55	
	American Indian	1136	20	61	1145	88	20	13	73	
		White	50997	45	83	1426	96	12	8	22
		Special Ed	7363	15	51	1047	87		9	40
		Metro Area	29357	46	82	1431	95	12	9	24
		Outstate	27967	40	80	1377	95	13	8	30
		Mols/St. Paul	5345	32	67	1257	93	15	10	61
		TCSuburbs	24012	49	85	1469	96	12	8	15
		Outstate: 2000+	13617	41	80	1385	95	13	7	27
		Outstate: 2000 -	14350	39	79	1370	96	13	9	34
		Public/Charter Public/Nor Charter	187 57410	34 43	66 81	1248 1405	93 95	20 13	57 8	51 <i>2</i> 7

TableA.14

.

Table A.141998 Grade 5: Minnesota
Comprehensive Assessment
Results in Writing for all
Public School Students
Tested Except Those New to
Their District Since
January 1, 1997

1998 GRADE 5: Mnnesota Comprehensive Assessment Results in WATHING for all Public School Students Except those New to their District since January 1, 1997

Caregory	Narte Tested	%Atar Above Level III	%Ator Above Level []	MenSak Sare	%Findled Sachris Tested	%LP Sudensin Sare	%Special Ed Suctents in Score	%F/R Sartensin Sare
Total	54511	43	81	1403	- 95	3	13	27
Gits	26607	53	88	1505	96	3	8	27
Boys	27886	33	74	1305	95	4	17	27
Asian	2467	35	76	1331	96	52	9	65
Black	2674	22	57	1138	92	3	20	77
Hsperic	1065	25	65	1215	92	33	16	64
Aminican Indan	988	19	61	1150	89	0+	20	71
White	47187	45	83	1431	96	0+	12	21
Special Fed	6909	15	51	1050	87	3	_	40
LEP	1835	24	62	1150	94		12	89
MeroArea	28310	45	82	1424	95	5	12	25
Quar	25971	40	80	1380	96	1	13	30
Mpts/St. Paul	6033	30	6	1237	93	21	15	67
TCSiturts	22277	50	86	1475	96	1	12	13
Oustare 2000+	12872	41	80	1384	95	2	13	27
Oustar: 2000-	13099	39	80	1375	96	1	13	33
Rittic/Chatter	102	28	Q	1158	94	22	27	45
Rublic/Not Charter	54409	43	81	1403	95	3	B	27

Note: LEP=Limited Ergish Protocory; F/R=Highte for free or reduted hand All processes and Man Sale Scores have been

runded to the nearest whole number; "O+" indicates a .4% or less.



Table A.15	<i>1998 Grade 5:</i> Minnesota Comprehensive Assessment <i>Results in</i> <i>Writing for all Public</i>	Category	Number Tested	% At or Above Level III	% At or Above Level II	Mean Scale Score	Fnrolled Students Tested	%LEP Students in Score	% New to District Since 1/1/97	% F/R Students in Score
	School Students Tested	Total	52078	46	85	1747	96	4	9	27
	Except Those in Special Education	Girls Boys	26690 25361	55 36	90 79	1531 1358	97 96	3 4	9 9	28 27
	Asian Black Hispanic American Indian White	2526 2571 1063 908 44840	37 25 28 24 49	78 64 69 68 87	1357 1212 1261 1221 1474	96 93 90 91 97	51 4 33 0+ 0+	11 16 13 8	64 76 65 71 21	
		LEP	1844	20	65	1181	91		13	88
		Metro Area Outstate	27273 24595	49 43	85 84	1467 1424	% %	6 1	9 8	26 29
		Mpls/St. Paul TC Suburbs Outstate: 2000+ Outstate: 2000-	5740 21533 12063 12532	33 53 44 42	72 89 84 84	1299 1512 1431 1418	95 97 96 97	21 1 2 1	10 9 7 9	66 15 26 32
		Public/Charter Public/Not Charter	168 51910	36 46	71 85	1 307 1447	96 96	11 4	55 8	54 27
Table A.16	<i>1998 Grade 8:</i> Basic			%		-	%	Special	% New to	0% E/D

.16 1998 Grade 8: Basic Standards Test Results in Reading for all Public School Students Tested Except Those with Limited English Proficiency

Category	Number Tested	% Meeting H.S. Standard	Mean Number Correct	% Enrolled Students Tested	% Special Ed Students in Score	% New to District Since 1/1/97	% F/R Students in Score
Total	62824	69	31.43	96	12	7	22
Girls	30419	72	32.01	96	7	7	23
Boys	31595	68	31.03	96	16	17	22
Asian	1768	66	31.05	97	5	10	45
Black	2539	33	24.70	89	24	- 15	71
Hispanic	912	48	28.04	92	14	17	52
American Indian	1132	38	26.06	88	24	14	64
White	55011	73	32.02	97	11	6	18
Special Ed	7347	28	23.33	88	_	11	40
Metro Area	29875	70	31.54	96	12	7	20
Outstate	32475	69	31.37	96	12	7	26
Mpls/St. Paul	5184	47	27.46	90	18	9	58
TC Suburbs	24691	75	32.39	97	10	6	12
Outstate: 2000+	15867	70	31.48	96	12	6	22
Outstate: 2000 -	16608	68	31.26	97	12	8	29
Public/Charter	180	43	26.19	98	26	46	50
Public/Not Charter	62644	69	31.44	96	12	7	23

Note: LEP=Limited English Proficiency; F/R=Eligible for free or reduced-price lunch; 0+ indicates a .4% or less. All percentages and Mean Scale Scores are rounded to the nearest whole number.



Table A.171998 Grade 8: BasicStandards Test Results in

Reading for all Public School Students Tested Except Those New to Their District Since January 1, 1997

Category	Number Tested	% Meeting H.S. Standard	Mean Number Correct	% Enrolled Students Tested	% LEP Students in Score	% Special Ed Students in Score	% F/R Students in Score
Total	59732	69	31.38	96	2	11	23
Girls	28949	72	31.96	97	2	6	23
Boys	30098	67	30.98	96	3	15	23
Asian	2507	49	28.09	93	37	7	61
Black	2268	33	24.64	90	5	20	72
Hispanic	1028	42	26.44	90	26	16	60
American Indian	977	39	26.15	90	0+	21	64
White	51660	73	32.14	97	0+	10	18
LEP	1381	16	21.85	86		12	89
Special Ed	6449	25	22.84	87	3		40
Metro Area	28971	69	31.35	96	4	11	21
Outstate	30485	69	31.44	97	1	11	25
Mpls/St. Paul	5653	42	26.57	90	16	13	63
TC Suburbs	23318	76	32.51	97	1	10	11
Outstate: 2000+	15115	70	31.52	96	1	11	22
Outstate: 2000 -	15370	69	31.35	97	0+	11	28
Public/Charter	100	36	24.99	96	2	25	53
Public/Not Charter	59632	69	31.39	97	2	11	23

Table A.181998 Grade 8: BasicStandards Test Results in
Reading for all Public School
Students Tested Except Those
in Special Education

						TIZ BI	
Category	Number Tested	Meeting H.S. Standard	Mean Number Correct	% of Enrolled Students Tested	% LEP Students in Score	% New to District Since 1/1/97	% F/R Students in Score
Total	55411	74	32.37	97	2	7	22
Girls Boys	28470 26683	74 74	32.50 32.29	97 97	2 3	7 7	23 21
Asian Black Hispanic American Indian White	2579 2055 1061 857 48859	50 38 43 46 78	28.40 26.19 26.92 28.03 33.04	94 92 88 90 98	35 7 26 0	10 16 19 13 6	61 70 60 61 16
LEP	1396	17	22.18	85		15	89
Metro Area Outstate	26919 28181	74 74	32.27 32.48	97 98	4 1	7 7	20 24
Mpls/St. Paul TC Suburbs Outstate: 2000+ Outstate: 2000 -	4932 21987 13805 14376	47 80 75 74	27.81 33.27 32.51 32.44	91 98 97 98	18 1 2 0	9 6 6 7	62 11 20 27
Public/Charter Public/Not Charter	98 55313	52 74	28.59 32.38	100 97	2 2	47 7	44 22

Note: LEP=Limited English Proficiency; F/R=Eligible for free or reduced-price lunch; 0+ indicates a .4% or less. All percentages and Mean Scale Scores are rounded to the nearest whole number.



BEST COPY AVAILABLE

Table A.19	<i>1998 Grade 8:</i> Basic Standards Test <i>Results in</i> <i>Mathematics for all</i> <i>Students Tested Except</i>	Category	Number Tested	Meeting H.S. Standard	Mean Number Correct	% Enrolled Students Tested	% Special Ed Students in Score	% New to District Since 1/1/97	% F/R Students in Score
	Those with Limited English	Total	62813	72	54.15	96	12	7	23
	Proficiency	Girls	30400	71	53.97	97	7	7	23
		Boys	31544	74	54.63	96	16	7	22
		Asian	1767	69	53.81	97	5	10	45
		Black	2547	27	39.13	89	24	15	72
		Hispanic	900	46	46.68	91	14	17	52
		American Indian	1137	39	44.18	88	24	14	64
		White	54963	76	55.39	97	11	6	18
		Special Ed	7344	30	39.98	88		11	40
		Metro Area	29821	72	54.03	96	12	7	20
		Outstate	32510	72	54.33	96	12	7	26
		Mpls/St. Paul	5192	45	45.44	90	18	9	58
		TC Suburbs	24629	77	55.84	97	10	6	12
		Outstate: 2000+	15890	73	54.64	96	12	6	22
		Outstate: 2000 -	16620	72	54.05	97	12	8	29
		Public/Charter	176	40	42.51	96	27	44	50
		Public/Not Charter	6263/	/2	54.18	96	12	7	23

Table A.201998 Grade 8: BasicStandards Test Results in
Mathematics for all

Students Tested Except Those New to Their District Since January 1, 1997

Catempy	Number	% Meeting	Mean	% Enrolled	%LEP	% Special Ed	%F/R
Category	Tested	Standard	Correct	Tested	Score	Students in Score	Score
Total	59741	72	54.14	- 96	2		23
Girls	28949	71	53.95	97	2	6	23
Boys	30051	74	54.62	96	3	15	23
Asian	2513	54	48.98	94	37	7	61
Black	2278	28	39.17	91	5	20	73
Hispanic	1016	41	44.18	89	26	16	60
American Indian	978	41	44.74	90	0+	21	64
White	51629	77	55.62	97	0+	10	18
LEP	1385	23	38.23	86	_	12	89
Special Ed	6456	28	39.15	82	3	_	40
Metro Area	28941	71	53.81	96	4	11	21
Outstate	30517	73	54.49	97	1	11	25
Mpls/St. Paul	5673	42	44.26	90	16	13	63
TC Suburbs	23268	78	56.14	97	1	10	11
Outstate: 2000+	15141	74	54.76	96	1	11	22
Outstate: 2000 -	15376	72	54.23	97	0+	11	28
Public/Charter	100	35	40.59	97	2	25	53
Public/Not Charter	59641	72	54.16	96	2	11	23

Note: LEP=Limited English Proficiency; F/R=Eligible for free or reduced-price lunch; 0+ indicates a .4% or less. All percentages and Mean Scale Scores are rounded to the nearest whole number.



Table A.211998 Grade 8: Basic StandardsTest Results in Mathematics for
all Students Tested ExceptThose in Special Education

Category	Number Tested	% Meeting H.S. Standard	Mean Number Correct	% of Enrolled Students Tested	% LEP Students in Score	% New to District Since 1/1/9	% F/R Students in Score
Total	55375	77	55.85	- 97	3	7	22
Girls	28461	74	54.97	97	23	7	23
Boxs	26657	81	56.89	97		7	21
Asian	2582	56	49.77	94	35	10	61
Black	2069	32	41.60	92	7	16	69
Hispanic	1053	42	45.64	88	27	19	60
American Indian	863	48	47.82	91	0+	14	61
Vinte I FP	40000 1405	25	39.26	86		16	89
Miro Area	26871	75	55.42	97	4	7	20
Ouistate	28193	78	56.27	98	1	7	24
Metro Area TC Suburbs Ouistate: 2000+ Ouistate: 2000-	4947 21924 13815 14378	47 82 79 78	46.54 57.43 56.48 56.06	91 98 97 98	18 1 2 0+	9 6 7	62 11 20 27
Public/Charter	94	49	45.94	96	2	46	44
Public/Not Charter	55281	77	55.87	97	3	7	22

Note: LEP=Limited English Proficiency; F/R=Eligible for free or reduced-price lunch; 0+ indicates a .4% or less. All percentages and Mean Scale Scores are rounded to the nearest whole number.





APPENDIX B CONTENT AND PERFORMANCE STANDARDS



HOW TO OBTAIN MORE INFORMATION ON CONTENT (CURRICULA) AND PERFORMANCE STANDARDS

NATIONAL COUNCIL OF TEACHERS OF MATHEMATICS (NCTM)	The National Council of Teachers of Mathematics sets content standards for mathematics in their 1989 publication <i>Curriculum and Evaluation Standards</i> <i>for School Mathematics</i> . This publication details the mathematics content a student should know and be able to demonstrate. A revision of these stan- dards is due October 1998. This will be a draft and is free to NCTM mem- bers and will be available to non-members for a nominal fee for shipping and handling, \$7.50. You may order the standards online (<u>http://www.nctm.org/</u> <u>standards2000</u>), by telephone (888-220-7952 or 703-620-9840 ext. 2103), by fax (703-476-2970), by email (<u>standards2000-draft@nctm.org</u>), or by mail (NCTM / Drawer A / 1906 Association Drive / Reston, VA 20191-1593 / Attn: Standards 2000).
NATIONAL COUNCIL OF TEACHERS OF ENGLISH (NCTE)	The National Council of Teachers of English and the International Reading Association (<u>http://www.ira.org</u>) set content standards for English in their 1996 publication, <i>Standards for the English Language Arts</i> . The cost of the publication is \$13.00 to members and \$18.00 to nonmembers. NCTE can be reached online (<u>http://www.ncte.org/standards</u>), by telephone (800-369- 6283 or 217-328-3870), by fax (217-328-9645), by email (<u>standards@ncte.org</u>), or by mail (NCTE / 1111 West Kenyon Road / Urbana, IL 61801).
NATIONAL RESEARCH COUNCIL	The National Research Council, an arm of the National Academy of Science created standards in science. These standards have been adopted and promoted by the National Science Teachers Association (<u>http://www.nsta.org</u>). The 1996 publication, <i>National Science Education Standards</i> , is available for \$19.95 (prepaid) plus \$4.00 shipping and handling (\$0.50 more for each additional copy). Discounts apply for multiple orders. Orders can be sent to the National Academy Press by telephone (202-334-3313 or 800-624-6242), by mail (National Academy Press / 2101 Constitution Avenue, NW / Washington, DC 20418), or online (<u>http://www.nap.edu/bookstore</u>). Note that books ordered online merit a 20% discount.
NATIONAL COUNCIL FOR THE SOCIAL STUDIES (NCSS)	The National Council for the Social Studies developed social studies stan- dards in 1994. Their publication, <i>Expectation of Excellence: Curriculum</i> <i>Standards for Social Studies</i> , is available at \$15.00 to non-members and \$12.75 to members. There is a \$4 charge for shipping and handling. There are multiple copy discounts. To order the publication you can contact NCSS by telephone (800-683-0812), by fax (301-843-0159), by mail (NCSS Publications / P.O. Box 2067 / Waldorf, Maryland 20604-2067), or online (http://www.ncss.org/bookstore/standards.html).
NATIONAL CENTER FOR HISTORY IN THE SCHOOLS	There is also a set of standards for history (both United States history and world history). The National Center for History in the Schools (<u>http://</u> <u>www.sscnet.ucla.edu/nchs</u>) originated the standards in 1994 and subsequently revised then in 1996. The standards are \$15.95 for nonmembers and \$15 for members. To order the standards contact the UCLA Store which can be accessed by telephone (310-206-0788), by fax (310-825-0382), by mail



CENTER FOR CIVIC EDUCATION

NATIONAL COUNCIL FOR GEOGRAPHIC EDUCATION

American Council on the Teaching of Foreign Languages (ACTFL)

CONSORTIUM OF NATIONAL ARTS EDUCATION ASSOCIATIONS

The American Alliance for Theatre & Education The Music Educators National Conference The National Art Education Association The National Dance Association.



(UCLA Book Zone / 308 Westwood Plaza / Ackerman Union / Los Angeles, CA 90024-1645), or by e-mail at: (<u>bookorder@asucla.ucla.edu</u>). Note that these standards can also be ordered via NCSS's online bookstore (<u>http://www.ncss.org/bookstore/standards.html</u>).

Civics and government also have a set of standards. These were created by the Center for Civic Education. The publication is entitled *National Standards for Civics and Government*. For 1 – 9 copies the publication is \$14.00 each. If you order 10 or more copies, the cost goes to \$12.00 per book. The Center for Civic Education can be contacted by phone (818-591-9321), fax (818-591-9330), e-mail (center4civ@aol.com), mail (Center for Civic Education / 5146 Douglas Fir Rd. / Calabasas, CA 91302-1467), or online (http://www.civiced.org). Note that these standards can also be ordered via NCSS's online bookstore (http://www.ncss.org/bookstore/standards.html).

Geography for Life: The National Geography Standards are available from the National Council for Geographic Education. Single copies are available for \$7 each with a \$2.50 charge for shipping and handling for the first book. There are multiple copy discounts on books and shipping and handling. NCGE can be contacted by mail (National Council for Geographic Education / Leonard 16A / Indiana University of Pennsylvania / Indiana PA 15705), by telephone (724-357-6290), by email (NCGE-ORG@grove.iup.edu), or online (http://www.ncge.org). Note that these standards can also be ordered via NCSS's online bookstore, (http:// www.ncss.org/bookstore/standards.html).

The American Council on the Teaching of Foreign Languages (<u>http://</u><u>www.actfl.org</u>), and a host of other organizations developed a set of standards for foreign language. Standards for Foreign Language Learning: Preparing for the 21st Century. It is available at \$20.00 per copy for less than ten books and \$15.00 per copy for ten or more books. These prices include shipping and handling. You can order by telephone (800-627-0629 or 913-843-1221), by fax (913-843-1274), or by mail (National Standards Report / P.O. Box 1897 / Lawrence, KS 66044)

The Arts Standards were developed by the Consortium of National Arts Education Associations (http://artsedge.kennedy-center.org/cs/design/ standards) which consisted of the American Alliance for Theatre & Education (http://www.aate.com), the Music Educators National Conference (http:// www.menc.org), the National Art Education Association (http://www.naeareston.org), and the National Dance Association (http://www.aahperd.org) . National Standards for Arts Education: What Every Young American Should Know and Be Able to Do in the Arts, is available for \$20 (nonmembers) or \$16 (members) and can be ordered from MENC by mail (Music Educators National Conference / 1806 Robert Fulton Drive, Reston, VA 20191), by telephone (800-828-0229), by fax (888-275-MENC), or online (http:// www.menc.org/publication/books/order.html).

Please Note: Web addresses and e-mail addresses (contained within parentheses) are underlined in this section. All of the underlined text, without the parentheses, should be entered on your computer without spaces between letters or between symbols. For space reasons, we were unable to ensure that all addresses would appear unbroken and on a single line of text. Information and addresses provided in this section were correct when this report went to press.





1998 MINNESOTA EDUCATION YEARBOOK GLOSSARY

Achievement test	An examination that measures the extent to which a person has acquired certain information or mastered certain skills, usually as a result of specific instruction.
ACT Assessment Program	The ACT assessment program measures educational development and readiness to pursue college-level coursework in English, mathematics, natural science, and social science. Student performance on the tests does not solely reflect innate ability and is influenced by a student's educational preparedness.
ACT core academic courses	These are courses that the ACT Assessment program suggests students complete prior to high school graduation. The courses include: four years of English, three years of science, three years of social studies and three years of mathemat- ics. The English portion of the test consists of punctuation 13%, basic gram- mar 16% and sentence structure 24%. Rhetorical skills include strategy 16%, organization 15%, and style 16%. The math portion consists of pre-algebra 23%, elementary algebra 17% intermediate algebra 15%, coordinate geometry 15%, plane geometry 23%, and trigonometry 7%. The reading portion consists of passages from social studies 25%, natural sciences 25%, prose fiction 25% and humanities 25%. The science portion consists of data representation 38%, research summary 45%, and conflicting viewpoints 17%. Web site: <u>http://www.act.org/</u>
Administration–Expenditure Category	Expenditures for the school board and for the office of the superintendent, principals, and any other line administrators who supervise staff.
Advanced Placement Programs (AP)	AP gives highly motivated students an opportunity to take college-level courses and exams while still in high school. There are now 32 different AP courses to choose from, in 18 different subject areas, offered by approximately 14,000 high schools worldwide. In 1998, AP reached a milestone — more than a million exams were taken by about half a million students. The College Board adminis- ters the exams. AP examination grades are reported on a 5-point scale as follows: 5 – extremely well qualified; 4 – well qualified; 3 – qualified; 2 – possible qualified; 1 – no recommendation. A score of 3 or above will receive college credit or advanced placement. Web site: <u>http://www.collegeboard.org/ap</u>
Assurance of Mastery revenue	Districts that have identified direct instructional services to assure that K-8 pupils master learner outcomes in communications and math are eligible for state aid. Other district revenue must match the state aid. This revenue, along with limited English proficient revenue and assurance of mastery revenue, is included in the targeted need revenue category.
At-risk students	Those students in danger of failing to complete their education with the skills necessary for a modern technological society.
Average Daily Attendance (ADA)	The aggregate attendance of a school during a reporting period (normally a school year) divided by the number of days school is in session during this period. Only days on which the pupils are under the guidance and direction of teachers should be considered days in session.
Average Daily Membership (ADM)	The aggregate enrollment of a school during a reporting period (normally a school year) divided by the number of days school is in session during this period. Pupils need not be in attendance to be counted in ADM, but they must be in membership.



Bachelor's degree	A degree granted for the successful completion of a baccalaureate program of studies, usually requiring at least 4 years (or equivalent) of full-time college-leve study.	
Basic standards	These standards represent <i>one</i> of the two components of the Minnesota Gradua- tion Rule established in 1992. The Basic Standards represent the minimum skills required for a high school diploma in Minnesota.	
Charter Schools	Publicly funded schools that are granted a high degree of autonomy from existing rules and regulations. Depending upon state law, teachers, parents, or other would-be educators can apply for permission to open a school. The "charter" may be granted by, for example, the local school board, the state board of education, or a public institution of higher education, depending upon the state. Some states also allow existing public or nonsectarian private schools to convert to charter status. Charter schools have the potential to control their own budget, staffing and curriculum, but their autonomy varies from state to state. They must attract students and achieve the results agreed to in their charters, or their contracts can be revoked.	
Choice options	The school choice options in Minnesota include the Postsecondary Enrollment Option, open enrollment or charter schools.	
Class size	The number of students a teacher has in his/her class at a given time.	
Compensatory funds (also known as Compensatory Educa- tion Revenue)	Based on a complex formula which provides additional funding for districts with students eligible to receive free lunch and/or reduced priced lunch based on October 1st enrollments of the previous fiscal year. Compensatory revenue increases as the percent of students eligible for free and reduced lunch increases. The percent is capped, however.	
Completion rate	Refers to the percent of students who complete high school in four years.	
Content Standards	Content standards define what students should know and be able to do in key academic subjects at specific grades.	
Curriculum	A school's master plan for selecting content and organizing learning experiences for the purpose of changing and developing learners' behaviors and insights. A curriculum is characterized by its scope (breadth of content) and sequence (organization of content).	
Dropout rate	The percentage of students that leave high school before receiving their diploma. Students who transfer to a non-public high school or to a public high school in another state are <u>not</u> counted as a dropout.	
Educational Accountability	Is a systematic method to assure those inside and outside the education system that schools and students are moving toward desired goals. In Minnesota, it is a statewide system that is applicable, with appropriate assessment accommoda- tions, to all students, including those with disabilities and limited proficiency in English.	
Educational attainment	The highest grade of regular school attended and completed.	
Enrollment	The total number of students registered in a given school unit at a given time, generally in the fall of a year.	
Equity	Refers to equal treatment, justice.	
Ethnicity	The cultural heritage of a particular group.	



.

Exceptional Instruction– Expenditure Category	Expenditures for instruction of students who, because of atypical characteris- tics or conditions, are provided educational programs that are different from regular instructional programs. Includes expenditures for special instruction of students who are emotionally or psychologically disabled, or mentally retarded; for students with physical, hearing, speech, and visual impairments; and for students with special learning and behavior problems.
Federal funding	Federal funding is the percentage of revenues from the federal government, whether paid directly or through another governmental unit. It includes all federal appropriations, grants, and contracts received by districts. The funds are typically targeted toward specific minority and disadvantaged student populations.
First grade preparedness funds	For the 1996-97, 1997-98 and 1998-99 school years, certain school sites are eligible for funding to operate full day kindergarten programs or half day programs for four year olds to develop reading and other skills necessary to succeed in school. School sites with the highest concentrations of pupils eligible for free and reduced lunch are eligible for funding. The funding is the amount equal to .53 times pupils enrolled in the program times the general education formula allowance.
Food Support -Expenditure Category	Expenditures for the preparation and serving of meals and snacks to students.
Foundation formula (also known as the General Education Fund- ing Program)	The general education funding program is the method by which school districts receive the majority of their financial support. It is designed to provide a basic foundation of funding for all districts irrespective of local resources. It also channels more state aid to districts with low residential and commercial tax bases.
Free lunch/reduced lunch	The eligibility requirements are based on household size and total household income. Household size includes every child and adult in the household, whether related or unrelated. Every person who shares housing and/or expenses is considered to be part of your household for this purpose. To qualify, a total household income should not exceed the following amounts. Household size to total monthly household income: 1/\$1,242; 2/\$1,673; 3/ \$2,105; 4/\$2,537; 5/\$2,968; 6/\$3,400; 7/\$3,832; 8/\$4,263. For each additional household member add \$432. (Application for educational benefits 1998-99, Free or reduced price school meals - State and Federally Funded Programs for Schools)
Full-time-equivalent (FTE)	School staff members are counted using FTE values. For example, a full-time staff member is counted as 1.0 FTE; one employed only half time is counted as .5 FTE.
Graduation rate	For the purposes of this report, graduation rate refers to the proportion of public school ninth graders who graduate from high school four years later. Ninth grade students who transfer to a non-public school or to a public school in another state are excluded from the calculations.
IDEA	Individuals with Disabilities Education Act, the federal law that oversees the provision of a free and appropriate public education to students with disabili- ties.
IEA	International Association for the Evaluation of Educational Achievement is an independent international cooperative of research centers and departments of education in more than 50 countries.



-

Instructional Support-Expenditure Category

International Baccalaureate (IB)

Limited English Proficiency (LEP)

Local sources

Master's degree

Mean score

Metro Area

Mini-nation

Minnesota Comprehensive Assessments (MCA)

Mobility



The match between learning goals, learning activities, and assessment. Alignment is critical if teaching is to be effective and learning is to be maximized.

- Expenditures for activities intended to help teachers provide instruction, not including expenditures for principals or superintendents. Includes expenditures for assistant principals, curriculum development, libraries, media centers, audiovisual support, staff development, and computer-assisted instruction.

The International Baccalaureate Diploma Program is a rigorous pre-university course of studies, leading to examinations, that meets the need of highly motivated secondary school students between the ages of 16 and 19 years. Designed as a comprehensive two-year curriculum that allows its graduates to fulfill requirements of various national education systems, the diploma model is based on the pattern of no single country but incorporates the best elements of several. Each examined subject is graded on a scale of 1 (minimum) to 7 (maximum). The award of the diploma requires students to meet defined standards and conditions including a minimum total of 24 points and the satisfactory completion of the extended essay, Theory of Knowledge course (TOK) and CAS (creativity, action, service) activities. The maximum score of 45 includes three points for the combination of the extended essay and work in TOK. IB diploma holders gain admission to selective universities throughout the world, including University of Minnesota, Oxford, Yale, and Sorbonne. Formal agreements exist between the IBO and many ministries of education and private institutions. Some colleges and universities may offer advanced standing or course credit to students with strong IB examination results. The program is available in English, French, and Spanish. (Web site: http://www.ibo.org).

A student with limited English Proficiency is defined as one whose primary language is not English and whose score on an English reading or language arts test is significantly below the average score for students of the same age. This definition is used by the Minnesota legislature; however, it may vary across school districts.

The percent of revenues from local sources, including property taxes, fees, county apportionment, etc.

A degree awarded for successful completion of a program generally requiring 1 or 2 years of full-time college-level study beyond the bachelor's degree.

What is normally meant by the word average. The total of the scores divided by the number of scores.

Refers to school districts located in Minneapolis, St. Paul, and the seven county metro area.

The label given to individual states that participated in the TIMSS study. States were offered the opportunity to assess a state-representative sample of their students at the same time as the U.S. National TIMSS study. Colorado, Illinois and Minnesota joined in this program. Web site: <u>http://www.ed.gov/</u> <u>NCES/timss/brochure.html</u>

These tests are given at the third and fifth grade levels to evaluate student progress on the Preparatory Standards and to measure the success of schools and districts in improving achievement over time.

The number of times a student moves from school to school or district to district in a given year (frequent school or residence changes).

National Assessment of Educational Progress (NAEP) NAEP is often called the "nation's report card." It is the only regularly conducted survey of what a nationally representative sample of students in grades 4, 8, and 12 know and can do in various subjects. The project is mandated by Congress and carried out by the National Center for Education Statistics at the U.S. Department of Education. Beginning in 1990, the survey was expanded to provide state-level results for individual states that choose to participate. The policy defines three NAEP achievement levels basic, proficient and advanced. The definitions for each level follow. A basic achievement level denotes partial mastery of prerequisite knowledge and skills that are fundamental for proficient work at each grade. A proficient achievement level represents solid academic performance for each grade accessed. Students reaching this level have demonstrated competency over challenging subject matter, including subject-matter knowledge, application of such knowledge to real-world situations, and analytical skills appropriate to the subject matter. An advanced achievement level signifies superior performance.

The NAEP scores have been evaluated at certain performance levels. In reading a score of 300 implies an ability to find, understand, summarize and explain relatively complicated literary and informational material. A score of 250 implies an ability to search for specific information, interrelate ideas, and make generalizations about literature, science and social studies materials. A score of 200 implies an ability to understand, combine ideas, and make inferences based on short uncomplicated passages about specific or sequentially related information. A score of 150 implies an ability to follow brief written directions and carry out simple, discrete reading tasks. Scale ranges from 0 to 500. In 1994, the NAEP reading achievement levels were as follows: For Grade 4, basic achievement is a score of 208-237, proficient achievement is 238-267 and advanced achievement is above 268. For Grade 8, basic achievement is a score of 243-280, proficient achievement is 281-322 and advanced achievement is above 323. For Grade 12, basic achievement is a score of 265-301, proficient achievement is 302-345 and advanced achievement is above 346.

The NAEP scores have been evaluated at certain performance levels. In math performers at the 150 level know some basic addition and subtraction facts, and most can add two-digit numbers without regrouping. They recognize simple situations in which addition and subtraction applies. Performers at the 200 level have considerable understanding of two digit numbers and know some basic multiplication and division facts. Performers at the 250 level have an initial understanding of the four basic operations. They can also compare information from graphs and charts, and are developing an ability to analyze simple logical relations. Performers at the 300 level can compute decimals, simple fractions and percents. They can identify geometric figures, measure lengths and angles, and calculate areas of rectangles. They are developing the skills to operate with signed numbers, exponents, and square roots. Performers at the 350 level can apply a range of reasoning skills to solve multi-step problems. They can solve routine problems involving fractions and percents, recognize properties of basic geometric figures, and work with exponents and square roots. Scale ranges from 0 to 500. In 1996, the NAEP mathematics achievement levels were as follows: For



	Grade 4, basic achievement is a score of 214-248, proficient achievement is 249-281 and advanced achievement is above 282. For Grade 8, basic achievement is a score of 262-298, proficient achievement is 299-332 and advanced achievement is above 333. For Grade 12, basic achievement is a score of 288-335, proficient achievement is 336-366 and advanced achievement is above 367.
Open enrollment	Public-school-choice programs allow families to choose the public schools their children attend. Intradistrict programs limit a family's choice to some or all of the public schools in their own district. Open-enrollment programs allow families to choose schools outside the district in which they live.
Operations and Maintenance (Expenditure Category)	Expenditures for operation, maintenance, and repair of the district's buildings, grounds and equipment. Includes expenditures for custodians, fuel for buildings, electricity, telephones and repairs.
Other Operations (Expenditure Category)	Expenditures for general fund operating programs necessary to a district's operations but not able to be assigned to other programs. These can include federally funded community education services for students, property and liability premiums, principle and interest on non-capital obligations, and nonrecurring costs such as judgements and liens.
Outcomes	The desired results of an educational system
Outcomes-Based Education (OBE)	Is a structure at a school and district level that stresses clearly defined out- comes, criterion-referenced measures of success, and instructional strategies. These outcomes are directly related to student abilities and needs, flexible use of time and learning opportunities, recognition of student success, and modification of programs on the basis of student results. Web site: <u>http://</u> www.hrdc-drhc.gc.ca/hrdc/corp/stratpol/arbsite/research/r964sm e.html)
Outstate	Refers to the school districts located outside the seven county metro area. For some purposes, they are divided into districts that have enrollments of 2000 students or less (2000-), or enrollments of greater than 2000 students (2000+).
Performance Standards	Performance standards define in what ways and how well students must demonstrate their knowledge and skills to be considered competent.
Per-pupil expenditure or per- pupil spending	The State's annual total spending on public K-12 education divided by its total number of students. An adjusted amount makes the number compa- rable by taking into account how much it costs school districts in different regions to recruit and employ teachers with similar qualifications.
Postsecondary enrollment options (PSEO)	This program allows high school juniors and seniors to enroll in classes at postsecondary institutions at public expense and receive both high school and college credit for their courses. The Minnesota program is two fold: To promote rigorous academic pursuits and to provide a variety of options to high school students.
Proficiency levels on the Minnesota Comprehensive	There are four achievement levels that represent the expectations for academic success in Minnesota:
Assessments (MCA)	• Level I: Students at this level demonstrate evidence of limited knowledge and skills necessary for satisfactory work in the High Standards in the elementary grades.
	• Level II: Students at this level demonstrate evidence of partial knowledge and skills necessary for satisfactory work in the High Standards in the elementary grades.



	 Level III: Students at this level demonstrate evidence of solid academic performance and competence in the knowledge and skills necessary for satisfactory work in the High Standards in the elementary grades.
	• Level IV: Students at this level demonstrate evidence of advanced academic performance, knowledge and skills that exceed the level necessary for satisfactory work in the High Standards in the elementary grades.
Profile of Learning	The second component of the Minnesota standards-based Graduation Rule. It is a taxonomy of preparatory standards (K-8 th grade) and high standards (9-12 th grade) that students are expected to achieve before leaving high school.
Pupil Support (Expenditure Category)	Expenditures for all non-instructional services provided to students, not including transportation and food. Includes expenditures for counseling, guidance, health services, psychological services, and attendance and social work services.
Pupil Transportation (Expendi- ture Category)	Expenditures for transportation of students, including salaries, contracted services, fuel for buses, and other expenditures.
Pupil/staff ratios	Pupil/staff ratios are based on the total number of pupils in attendance (ADA) at a school compared to the total number of licensed school personnel (FTE) (e.g. administrators, counselors, teachers, media specialists, speech clinicians, psychologists, etc.) in that school.
Pupil/teacher ratios	Are based on the total number of pupils in attendance (ADA) at a school compared to the total number of licensed teaching staff (FTE) in that school.
Regular Instruction (Expendi- ture Category)	Expenditures for elementary and secondary classroom instruction, not including vocational instruction and exception instruction. Includes salaries of teachers, classroom aides, coaches, and expenditures for classroom supplies and textbooks
Results-oriented educational system	Same as Outcomes Based Education.
Scale score	A scale score provides a common scale for different forms of a test used at a given grade or across age/gender levels.
Scholastic Assessment Test (SAT)	Formerly known as the Scholastic Aptitude Test, the SAT is commonly used as a college entrance exam.
School Accreditation Processes	The awarding of credentials to schools in particular the award of membership in one of the regional associations of educational institutions that attempt to maintain certain quality standards for membership.
School climate	The social system and culture of the school, including the organizational structure of the school and values and expectations within it.
School improvement programs	Programs with the intent to improve school quality.
SciMath ^{MN}	Founded in 1993. A state partnership of Minnesota business, education, and government pursuing statewide improvement in the teaching and learning of K-12 mathematics and science based on the national mathematics and science education standards. SciMath ^{MN} 's vision is to increase the educational achievement and participation of all Minnesota students in science and mathematics to help them meet the complex challenges of their future.
Site-based management	Governance arrangements designed to give the people closest to students the ability to make decisions about their education. Typically, teachers, parents,



.

and administrators at the school site are given more say over such matters as staffing, budgets, curriculum, and instructional materials. But the level of autonomy granted to individual schools, who is involved in making the decisions, and whether they are focused on student learning vary widely.
Direct instructional activities or special learning experiences designed prima- rily for students identified as having exceptionalities in one or more aspects of the cognitive process or as being underachievers in relation to general level or model of their overall abilities. Such services usually are directed at students with physical, emotional, cognitive learning disabilities. Programs for the mentally gifted and talented are also included in some special education programs.
Often described as the positive and/or negative consequences that are placed on students, schools or districts as the result of student achievement data. The terms "low stakes" and "high stakes" express the varying levels of risk being placed on those responsible for the expected results.
The knowledge or skill level necessary for a particular rating or grade on a given dimension of achievement. It is used as a basis of comparison. See content standards and performance standards.
The percent of revenues from the Minnesota state government.
The purpose of a Learning Readiness program is to provide all eligible children adequate opportunities to participate in child development programs that enable the children to enter school with the necessary skills and behavior as well as the family stability needed for them to progress and flourish. Learning Readiness is offered in 345 school districts in Minnesota. The cost per child for Learning Readiness varies depending on the level of participa- tion. The average statewide cost is \$382 per child.
In most of this report, student poverty refers to students eligible for free or reduced lunch. Other indicators are possible; e.g. students from families receiving aid for Families with Dependent Children.
Expenditures for central office administration and central office operations not included in district and school administration. Includes expenditures for business services, data processing, legal services, personnel office, printing, and the school census.
The amount of education a teacher has. The major distinction is between teachers having Bachelor's Degrees and those having Master's Degrees.
Number of years in the teaching profession.
The annual pay received.
TIMSS is a study of classrooms across the country and around the world. It is the largest international comparative study of educational achievement to date. The National Center for Educational Statistics (NCES) of the U.S. Department of Education, the National Science Foundation (NSF) and the Canadian Government funded the international TIMSS project to assess school achievement in mathematics and science in nearly 50 countries. TIMSS studied student outcomes, instructional practices, curricula, and cultural context. TIMSS provides a comparative international assessment of educational achievement in mathematics and science, and the factors that contribute to achievement. Web site: http://www.ed.gov/NCES/timss/ brochure.html



Title I (Federally funded program)	Title I of the Elementary and Secondary Education Act (ESEA), as restruc- tured by the Improving America's Schools Act (IASA) of 1994, has as its primary focus to help disadvantaged students acquire the same knowledge and skills in challenging academic standards expected of all children. By the beginning of the 2000-2001 school year, Title I requires that each State develop or adopt a set of high-quality yearly student assessments that measure performance in at least mathematics and reading/language arts. Such assess- ments are to be aligned with the State content standards and be used to monitor progress toward achievement goals for accountability purposes. In a key change from previous law, States now use the same assessment that is used for all children to measure whether students served by Title I are achieving the State standards. There is no longer any requirement for a separate assessment for Title I students. Web page: <u>http://www.ed.gov/legislation/ESEA/Title I</u>
Total Operating (Expenditure Category)	The total of the following categories: administration, support services, regular instruction, vocational instruction, exceptional instruction, instruc- tional support, pupil support, operations and maintenance, food support, pupil transportation and other operations. This figure includes all expendi- tures incurred for the benefit of elementary and secondary education during the school year, except for capital and debt service expenditures.
Vocational Instruction (Expen- diture Category)	Expenditures in secondary schools for instruction that is related to job skills and career exploration. Includes expenditures for home economics, as well as industrial, business, agriculture, and distributive education.
Vouchers	Vouchers enable families to use public tax dollars to pay for their children's education at a public or private school of their choice. Voucher programs may or may not include private religious schools.



. •

1998 MINNESOTA EDUCATION YEARBOOK: LIST OF REFERENCES



American College Testing (ACT) Program (1992). ACT high school profile report H.S. graduating class 1992: State composite for Minnesota. (Code 240-000). Iowa City, IA: Author.

American College Testing Program (1997). ACT high school profile report H.S. graduating class 1997: State composite for Minnesota. (Code 240-000). Iowa City, IA: Author.

American College Testing Program (1998). ACT high school profile report H.S. graduating class 1998: State composite for Minnesota. (Code 240-000). Iowa City, IA: Author.

American Federation of Teachers (AFT). (1997a). Making standards matter 1997: An annual fifty-state report on efforts to raise academic standards. Washington, DC: Author.

American Federation of Teachers (AFT). (1997b). Survey & analysis of salary trends 1997. Washington, DC: Author.

Bennett, W. J. (1988). American education: Making it work (U.S. Secretary of Education). Washington, DC: U.S. Government Printing Office.

Bond, L., Roeber, E., & Connealy, S. (1998a). Annual survey of state student assessment programs: Volume II. Washington, DC: Council of Chief State School Officers.

Bond, L., Roeber, E., & Connealy, S. (1998b). Trends in state student assessment programs. Washington, DC: Council of Chief State School Officers.

Boozer, M., & Rouse, C. (1995, June). Intraschool variation in class size: Patterns and implication. (Working Paper No. 344; ERIC Document #ED 385935). Washington, DC: National Bureau of Economic Research, Industrial Relations Section.

Campbell, J. R., Donahue, P. L., Reese, C.M., & Phillips, G. W. (1996). *NAEP 1994 reading* report card for the nation and the states. Washington, DC: Office of Educational Research and Improvement, U.S. Department of Education.

Council of Chief State School Officers (1998). Key state education policies on K-12 education: Standards, graduation, assessment, teacher licensure, time and attendance. Washington, DC: Author.

Dolan, R. C., & Schmidt, R. M. (1987). Assessing the impact of expenditures on achievement: some methodological and policy considerations. *Economics of Education Review*, 6(3), 285-299.

Editorial Projects in Education Inc. (1997, January 22). Quality Counts. Washington, DC: Author.

Education and Social Stratification Branch, U.S. Bureau of the Census. (1998, May). Educational Attainment in the United States: March 1997 (#PPL-87). Washington, DC: Author.

Ekstrom, R. B., Goertz, M. E., Pollack, J. M., & Rock, D. A. (Spring, 1986). Who drops out of high school and why? Findings from a national study. *Teacher's College Record*, 87, 356-373.

Ferguson, R. F. (1991). Paying for public education: New evidence on how and why money matters. *Harvard Journal on Legislation, 28,* 465-498.

Glass, G. V., & Smith, M.L. (1978). Meta-analysis of research on the relationship of class size and achievement. San Francisco, CA: Far West Laboratory for Educational Research & Development.

Greenwald, R., Hedges, L. V., & Laine, R. D. (1996). Have times changed? The effect on school resources on student achievement. *Review of Educational Research*, 66(3), 361-396.

Hanushek, E. (1989, Fall). The impact of differential expenditures on school performance. *Educational Researcher*, 18(4), 45-51, 62.



Hart, B., & Risley, T. R. (1995). *Meaningful differences in everyday experiences of young American children*. Baltimore, MD: P.H. Brookes.

Jones, L. V., Bekhuis, T., & Davenport, Jr., E. C. (1985). College-bound high-school graduates of 1972 and 1980 compared by college major. *Research in Higher Education*, 23, 319-331.

King, R., & Mathers, J. (1996). The promise and reality of rewards for school improvement. *Journal of Education Finance*, Fall 1997, 23(2), 147-176.

Mandala, C. M. (1998). Study: Less funding goes to charters. *Minnesota Journal*, 5(8), 1,6, & 7.

Mazzoni, T. (1993). The changing politics of state education policy making: A 20-year Minnesota perspective. *Educational Evaluation and Policy Analysis*, 15(4), 357-379.

Minnesota Department of Children, Families & Learning. (1997). School District Profiles 1996-97. St. Paul, MN: Division of Management Assistance.

Minnesota Department of Children, Families, & Learning (1998a). Advanced placement and international baccalaureate programs. St. Paul, MN: Author.

Minnesota Department of Children, Families, & Learning. (1998b, December). Completion study for the class of 1997. St. Paul, MN: Author.

Minnesota Department of Children, Families, & Learning. (1998c). Preliminary estimate: numbers of students enrolled in Minnesota charter schools for 1997-98. St. Paul, MN: Author.

Minnesota Department of Children Families, & Learning. (1998d, September). The 1997 Minnesota high school follow-up study. St. Paul, MN: Author.

Minn. Stat. § 120.064.

Minn. Stat. §120B.30, Subd. 1(b).

Mueller, D. (1997, November). Characteristics of successful urban elementary schools. St. Paul, MN: Wilder Research Center.

Murnane, R. J., & Levy, F. (1996). Evidence from fifteen schools in Austin, Texas. In G. Burtless (Ed.), *Does money matter? The effect of school resources on student achievement and adult success.* Washington, DC: Brookings Institution.

National Commission on Excellence in Education (1983). A nation at risk: The imperative for educational reform. (Stock No. 065-000-00177-2). Washington, DC: U.S. Government Printing Office.

News America Digital Publishing, Fox News Online. (1998, June 29, 1:13 a.m. E.T. [513 GMT]). *Black, white high school graduation rates equalizing.* (Online: http://www.foxnews.com/news/wires2/0629/n ap 0629 13.sml).

Program Evaluation Division, Office of the Legislative Auditor, State of Minnesota. (1998). *Remedial Education* (Rep. No. 9802). St. Paul, MN: Author.

Program Evaluation Division, Office of the Legislative Auditor, State of Minnesota. (1988, December). *High School Education*. St. Paul, MN: Author.

Robinson, G.E. (1990). Synthesis of research on effects of class size. *Educational Leadership*, 47(7), 80-90.



SciMath^{MN}. (1997, March). *Minnesota TIMSS Report: A preliminary summary of results*. Third International Mathematics and Science Study (TIMSS). St. Paul, MN: Author. (Online: <u>http://www.informns.k12.mn.us/scimathmn</u>).

SciMath^{MN}. (1998, June). *Minnesota 4th Grade Results*. Third International Mathematics and Science Study (TIMSS). St. Paul, MN: Author. (Online: <u>http://www.scimathmn.org</u>).

Shaughnessy, C. A., Nelson, J. E., & Norris, N. A. (1998). NAEP 1996 mathematics cross-state data compendium for the grade 4 and grade 8 assessment. (NCES 98-481). Washington, DC: National Center for Education Statistics, U.S. Department of Education.

Snyder, T. D. (1997). *Digest of educational statistics* (NCES 98-015). Washington, DC: U.S. Department of Education, National Center for Education Statistics.

Teddie, C., & Stringfield, S. (1993). Schools Make a Difference: Lessons Learned from a Ten Year Study of School Effects. New York: Teachers College Press.

U.S. Department of Education, National Center for Education Statistics. (1996). Reading Literacy in the United States: Findings From the IEA Reading Literacy Study. Washington, DC: Author.

Wang, M.C., Haertel, G.D., & Walberg, H.J. (1993). Toward a knowledge base for school learning. *Review of Educational Research*, 63(3), 249-294.

Wright, S. P., Horn, S. P., & Sanders, W. L. (1997). Teacher and classroom context effects on student achievement: Implications for teacher evaluation. *Journal of Personnel Evaluation in Education*, 11, 57-67.

Zigarelli, M.A. (1996). An empirical test of conclusions from effective schools research. Journal of Educational Research, 90(2), 103-110.



ERIC



U.S. Department of Education

Office of Educational Research and Improvement (OERI) National Library of Education (NLE) Educational Resources Information Center (ERIC)



REPRODUCTION RELEASE

(Specific Document)

I. DOCUMENT IDENTIFICATION:

Title: Minnesota Education Year book: The Status o Education in Minnesota	FPue-K-12
Author(s): Davidson, M.; Erickson, R.; Davenport, E; Kwak, N.; Bjelinski, U. Seo, V.; Smith, M.; Wick, S.	; Danielson, H.; Kim, S.;
Corporate Source:	Publication Date:

II. REPRODUCTION RELEASE:

In order to disseminate as widely as possible timely and significant materials of interest to the educational community, documents announced in the monthly abstract journal of the ERIC system, *Resources in Education* (RIE), are usually made available to users in microfiche, reproduced paper copy, and electronic media, and sold through the ERIC Document Reproduction Service (EDRS). Credit is given to the source of each document, and, if reproduction release is granted, one of the following notices is affixed to the document.

If permission is granted to reproduce and disseminate the identified document, please CHECK ONE of the following three options and sign at the bottom of the page.

The sample sticker shown below will be affixed to all Level 1 documents	The sample sticker shown below will be affixed to all Level 2A documents	The sample sticker shown below will be affixed to all Level 2B documents	
PERMISSION TO REPRODUCE AND DISSEMINATE THIS MATERIAL HAS BEEN GRANTED BY	PERMISSION TO REPRODUCE AND DISSEMINATE THIS MATERIAL IN MICROFICHE, AND IN ELECTRONIC MEDIA FOR ERIC COLLECTION SUBSCRIBERS ONLY, HAS BEEN GRANTED BY	PERMISSION TO REPRODUCE AND DISSEMINATE THIS MATERIAL IN MICROFICHE ONLY HAS BEEN GRANTED BY	
TO THE EDUCATIONAL RESOURCES INFORMATION CENTER (ERIC)	TO THE EDUCATIONAL RESOURCES INFORMATION CENTER (ERIC)	TO THE EDUCATIONAL RESOURCES INFORMATION CENTER (ERIC)	
1	2A	2B	
Level 1	Level 2A	Level 2B	
	1	t	
L			
Check here for Level 1 release, permitting reproduction Check here for Level 2A release, permitting reproduction Check here for Level 2B release, permitting reproduction Check here for Level 2B release, permitting reproduction and dissemination in microfiche and in electronic media reproduction and dissemination in microfiche or other ERIC archival for ERIC archival collection subscribers only			
Doc If permission to	uments will be processed as indicated provided reproduction quality pro preproduce is granted, but no box is checked, documents will be proce	ermits. sssed at Level 1.	
I hereby grant to the Educational Res as indicated above. Reproduction f contractors requires permission from to satisfy information needs of educ	ources Information Center (ERIC) nonexclusive permiss rom the ERIC microfiche or electronic media by pers the copyright holder. Exception is made for non-profit re ators in response to discrete inquiries.	sion to reproduce and disseminate this document ons other than ERIC employees and its system production by libraries and other service agencies	
Sign Signature:	Printed Name/Position/Title:		
here, > Mark L. Daves	Mark Davidson, Ph.D., Divector		
plage Grganization/Address: The Office of	FEducational Accounta- Telephone:	1 Accounta - Telephone: FAX: 6(2-627-1009 612-627-1009	
ERIC VAIVErsity of Minnebota, 1313 Fifth St. SE, Ste. B-1,	University Technology Center E-Mail Address: Minneupolis, MN 55414 MLOO.	tc. umn.edu Date: 3/29/99	

III. DOCUMENT AVAILABILITY INFORMATION (FROM NON-ERIC SOURCE):

If permission to reproduce is not granted to ERIC, or, if you wish ERIC to cite the availability of the document from another source, please provide the following information regarding the availability of the document. (ERIC will not announce a document unless it is publicly available, and a dependable source can be specified. Contributors should also be aware that ERIC selection criteria are significantly more stringent for documents that cannot be made available through EDRS.)

Publisher/Distributor:		
Address:		
Price:	 	

IV. REFERRAL OF ERIC TO COPYRIGHT/REPRODUCTION RIGHTS HOLDER:

If the right to grant this reproduction release is held by someone other than the addressee, please provide the appropriate name and address:

Name:

Address:

V. WHERE TO SEND THIS FORM:

Send this form to the following ERIC Clearinghouse:

ERIC Clearinghouse on Educational Management 1787 Agate Street 5207 University of Oregon Eugene, OR 97403-5207

However, if solicited by the ERIC Facility, or if making an unsolicited contribution to ERIC, return this form (and the document being contributed) to:

ERIC Processing and Reference Facility 1100 West Street, 2nd Floor Laurel, Maryland 20707-3598

> Telephone: 301-497-4080 Toll Free: 800-799-3742 FAX: 301-953-0263 e-mail: ericfac@inet.ed.gov WWW: http://ericfac.piccard.csc.com

ERIC -088 (Rev. 9/97) VIOUS VERSIONS OF THIS FORM ARE OBSOLETE.